A Reduced Data Dynamic Energy Model of the UK Houses

By

Ali Badiei

Doctoral Thesis

Submitted in Partial Fulfilment of the Requirements for the Award of

Doctor of Philosophy of Loughborough University

6 June 2018

© Ali Badiei 2018

To My Parents

"The answers we have found only serve to raise a whole set of new questions. In some ways we feel we are as confused as ever, but we believe we are confused on a higher level and about more important things..."

Earl C. Kelley

Abstract

This thesis describes the development of a Reduced Data Dynamic Energy Model (RdDEM) for simulating the energy performance of UK houses. The vast quantity of Energy Performance Certificate (EPC) data stored at the national scale provides an unprecedented data source for energy modelling.

The majority of domestic energy models developed for the UK houses in recent years, including the Standard Assessment Procedure (SAP) model used for generating EPCs, employ BREDEM (Building Research Establishment Domestic Energy Model) based steady-state calculation engines. These models fail to represent the transient behaviours that occur between building envelope and systems with external weather conditions and occupants. Consequently, there is an ongoing debate over the suitability of such models for policy making decisions; which has raised the interest in dynamic energy models to overcome these shortcomings.

The RdDEM eliminates the main drawback associated with dynamic energy modelling, namely the large amount of required input data compared to steady-state models, by enhancing a reduced set of data which was originally collected for EPCs. A number of new inferences and methodological enhancements were tested and implemented in the RdDEM using a sample of semi-detached houses. In this way, SAP equivalent input data could be converted automatically for use in dynamic energy modelling software, EnergyPlus.

Simulations of indoor air temperatures and space heating energy demand from the RdDEM were compared to those from SAP for 83 semi-detached houses. The comparison was also carried out with more detailed models, on a sub-set of the modelled dwellings. Finally, the predicted energy savings that resulted from energy efficiency improvements of the dwellings were compared and estimated potential for saving energy from the RdDEM was quantified. The results show that it is technically feasible to develop dynamic energy models of these houses using equivalent inputs. In the majority of cases, the RdDEM predicted lower indoor air temperatures than SAP, and consequently the energy demands were lower. The RdDEM predicted annual space heating demand to be lower than SAP in 72% of the houses, however the difference was less than 10% in 94% of the houses. The RdDEM predicted slightly higher (< 2%) energy saving potentials compared to SAP when the same set of energy saving measures were implemented in both models.

The development of these new methods for automatically creating SAP equivalent inputs from reduced data but for use in a dynamic energy model offers new opportunities for inter-model comparisons as well as a dynamic alternative to the SAP when variations in energy demand and indoor air temperatures are required.

Acknowledgment

I am very thankful to my principal supervisor Dr David Allinson who has always been available to provide guidance, support and encouragement throughout all stages of my PhD. My second supervisor, Prof Kevin Lomas deserves my deepest thanks for believing in me and giving me the opportunity to study this PhD. His insightful guidance and experience have been extremely valuable in the course of this PhD.

During the course of this study I have been grateful to be a part of Loughborough University's Building Energy Research Group (BERG) and I would especially like to thank Dr Steven Firth for his valuable comments on this work.

A special thank to my internal examiner Prof Dennis Loveday and external examiner Prof Li Shao from Reading University for their constructive feedback.

Finally, I would like to dedicate this thesis to my family for their endless love, support and encouragement.

Contents

Abst	tract	i
Ackı	nowle	edgmentiii
List	of Fi	gures viii
List	of Ta	ıbles xii
Abb	revia	tions xiv
1.	INTR	ODUCTION1
1.1	1. E	Background1
1.2	2.	Justification of the Research4
1.3	3. <i>I</i>	Aim and Objectives 5
1.4	4. (Contribution to Knowledge6
1.5	5. (Dutline of the Thesis
2.	LITE	RATURE REVIEW 8
2.1	1. I	ntroduction
2.2	2. E	Energy Performance Certificate (EPC)8
	2.2.1	. Context
	2.2.2	. National Calculation Methodologies: SAP and RdSAP 10
2.3	3. I	Energy Modelling in Domestic Buildings14
	2.3.1	. Top-down Modelling Approach16
	2.3.2	. Bottom-up Modelling Approach18
	2.3.3	. Top-down or Bottom-up Approaches? 21
2.4	4. F	Reduced Data Building Energy Models24
	2.4.1	. Steady-State Models 24
	2.4.2	. Critical Analysis of the UK based Models
	2.4.3	. Dynamic Energy Models

2	.5.	Мо	del Validation	43
2	.6.	Dat	a Translators for Dynamic Simulation	46
	2.5.	1.	The gbXML Translator	46
	2.5.	2.	The HPXML Translator	49
2	.7.	Sur	nmary	50
3.	ονι	ERV	IEW OF METHODS	53
3	.1.	Intro	oduction	53
3	.2.	Мо	delling Dataset	55
3	.3.	Dat	a Preparation Process	60
	3.3.	1.	Zoning	61
	3.3.	2.	Geometry	62
	3.3.	3.	Equivalent Construction Materials	64
	3.3.	4.	Thermal Mass	67
	3.3.	5.	Internal Boundary Conditions	67
	3.3.	6.	External Boundary Conditions: Weather Data and Orientation	68ء
3	.4.	Tra	nslation of Data: The RdDEM Translator	69
3	.5.	Sur	nmary	71
4.	ZO	NING	AND ENHANCED GEOMETRY	.72
4	.1.	Intro	oduction	72
4	.2.	Ref	erence Model	73
	4.2.	1.	Building Geometry	73
	4.2.	2.	Construction Materials	75
	4.2.	3.	Internal Gains	78
	4.2.	4.	Heating System	81
	4.2.	5.	Ventilation and Infiltration	82
	4.2.	6.	Orientation and Weather Data	. 82

4.2	2.7.	Results of the Reference Model	83
4.3.	Zor	ning	
4.3	3.1.	Zoning Strategies	
4.3	3.2.	Simulation Results	
4.4.	Enł	nanced Geometry	91
4.4	4.1.	Modelling Geometry	91
4.4	1.2.	Simulation Results	
4.4	1.3.	Modelling Extensions	
4.5.	Sur	nmary	102
5. EC	QUIVA	ALENT CONSTRUCTION MATERIALS AND BOUND	ARY
CONDI	TION	S	104
5.1.	Intr	oduction	104
5.2.	Equ	vivalent Construction Materials	104
5.2	2.1.	External Walls	105
5.2	2.2.	Roofs	107
5.2	2.3.	Ground Floors	108
5.2	2.4.	Doors and Windows	111
5.2	2.5.	Equivalent Thermal Mass	112
5.3.	Equ	uivalent External Boundary Conditions: Weather Data	113
5.4.	Sur	nmary	115
6. TR		LATION, SIMULATION AND RESULTS	117
6.1.	Intr	oduction	117
6.2.	Tra	nslation	117
6.2	2.1.	Template Input Data File (TIDF)	118
6.2	2.2.	Translator	119
6.3.	Pro	cess Verification	121

6.4.	Sim	nulation Results	124
6.4	4.1.	Comparison of Energy Demand Results to SAP Estimate	es 124
6.4	1.2.	Comparison of Temperature Predictions to SAP Results	128
6.4	1.3.	Comparison of the Predicted Energy Improvements	132
6.5.	Sur	mmary	134
7. DI	scus	SSION	136
7.1.	Intr	oduction	136
7.2.	Ove	erview of Thesis and Achievements	136
7.3.	Mo	del Predictions and Comparison with SAP	141
7.4.	Lim	nitations of the Research	143
7.5.	Imp	plications of the Research	144
7.6.	Sur	mmary	145
8. CC	ONCL	USIONS	147
8. CC 8.1.	DNCL Intr	USIONS	 147 147
8. CC8.1.8.2.	DNCL Intr Sur	USIONS oduction mmary of the Main Conclusions	 147 147 147
 8. CC 8.1. 8.2. 8.2 	Intr Sur 2.1.	USIONS oduction mmary of the Main Conclusions Methodological Conclusions	 147 147 147 148
 8. CC 8.1. 8.2. 8.2 8.2 	Intr Sur 2.1. 2.2.	USIONS oduction mmary of the Main Conclusions Methodological Conclusions Model Verification	 147 147 147 148 149
 8. CC 8.1. 8.2. 8.2 8.2 8.2 8.2 	Intr Sur 2.1. 2.2.	USIONS roduction mmary of the Main Conclusions Methodological Conclusions Model Verification RdDEM Results and Comparison with SAP	 147 147 147 148 149 150
 8. CC 8.1. 8.2. 8.2 8.2 8.2 8.2 8.3. 	Intr Sur 2.1. 2.2. 2.3. Red	USIONS oduction mmary of the Main Conclusions Methodological Conclusions Model Verification RdDEM Results and Comparison with SAP commendation for Future Research	147 147 147 148 149 150 151
 8. CC 8.1. 8.2. 8.2 8.2 8.2 8.3. REFER 	DNCL Intr Sur 2.1. 2.2. 2.3. Rec RENC	USIONS roduction mmary of the Main Conclusions Methodological Conclusions Model Verification RdDEM Results and Comparison with SAP commendation for Future Research	147 147 147 148 149 150 151 153
 8. CC 8.1. 8.2. 8.2 8.2 8.2 8.3 REFER APPEN 	DNCL Intr Sur 2.1. 2.2. 2.3. Rec RENC	USIONS oduction mmary of the Main Conclusions Methodological Conclusions Model Verification RdDEM Results and Comparison with SAP commendation for Future Research ES	147 147 147 148 149 150 151 153 167
 8. CC 8.1. 8.2. 8.2 8.2 8.2 8.2 8.3 REFER APPEN 	DNCL Intr Sur 2.1. 2.2. 2.3. Rec RENC IDIX	USIONS roduction mmary of the Main Conclusions Methodological Conclusions Model Verification RdDEM Results and Comparison with SAP commendation for Future Research ES A B	147 147 147 148 149 150 151 153 167 172
 8. CC 8.1. 8.2. 8.2 8.2 8.2 8.2 8.3 REFER APPEN APPEN APPEN 	DNCL Intr Sur 2.1. 2.2. 2.3. Rec RENC IDIX	USIONS oduction mmary of the Main Conclusions Methodological Conclusions Model Verification RdDEM Results and Comparison with SAP RdDEM Results and Comparison with SAP ES SES A B C	147 147 147 148 149 150 151 153 167 172 188

List of Figures

Figure 1.1 Contribution of different sectors to the UK's total carbon dioxide
emissions of 2011 (DECC, 2012)
Figure 1.2 Residential energy consumption shown as a percentage of national energy consumption and in relative international form (Saidur et al.,
2007)
Figure 2.1 Example of EPC energy and EI ratings (RdSAP Manual, 2012) 9
Figure 2.2 Top-down and bottom-up modelling approaches (IEA, 1998) 15
Figure 2.3 Bottom-up modelling approach and its sub-groups based on input data [re-created from (Swan & Ugursal, 2009)]
Figure 2.4 Structure and form of BREHOMES model (Shorrock and Dunster, 1997)
Figure 2.5 Structure and form of Johnston model (Johnston, 2003)
Figure 2.6 Structure and form of the UKCDM (Boardman, 2007)27
Figure 2.7 Structure and form of the DECarb model (Natarajan and Levermore, 2007)
Figure 2.8 Structure and form of the EEP model (Jones et al., 2007) 29
Figure 2.9 Structure and form of the CDEM (Firth et al., 2010)
Figure 2.10 Structure and form of the CHM (Hughes, 2011)
Figure 2.11 Structure and form of the DDM (Mhalas et al., 2013)
Figure 2.12 Structure and form of the CHREM (Swan et al. 2009) 38
Figure 2.13 Analytic, Comparative and Empirical techniques (Judkoff et al. 2008)

Figure 2.14 Structure and form of the BIM to BEM translator (Dimitriou et al., 2016)
Figure 2.15 The translation process to convert gbXML files into EnergyPlus IDFs (Dimitriou et al., 2016)
Figure 2.16 The translation process to convert HPXML files into various IDF formats (Polly et al., 2012)
Figure 3.1 Overall structure and the modelling framework
Figure 3.2 Dwellings distribution in the modelling dataset based on construction age band and the corresponding construction years
Figure 3.3 Dwellings distribution in the modelling dataset based on external wall construction
Figure 3.4 Dwellings distribution in the modelling dataset based on loft insulation thickness
Figure 3.5 Dwellings distribution in the modelling dataset based on total floor area
Figure 3.6 Party wall in semi-detached dwellings as modelled in DesignBuilder
Figure 4.1 Floor plans of the semi-detached A&P house (left: ground floor, right: first floor) and floor area (m ²) of each room (re-created from (Allen and Pinney, 1990))
Figure 4.2 Front (left) and back (right) view of the semi-detached A&P house model in DesignBuilder
Figure 4.3 Window frame definition in DesignBuilder (DesignBuilder, 2015).
Figure 4.4 Occupancy gain profiles for individual rooms of A&P semi- detached house model

Figure 4.5 Lighting gain profiles for individual rooms of A&P semi-detached
house model
Figure 4.6 Zoning strategies, their level of complexity and number of zones85
Figure 4.7 Internal temperatures of the three zoning strategies and the
reference model under summer conditions. a) Maximum mean daily
temperatures, b) minimum mean daily temperatures c) maximum mean
monthly temperatures, d) minimum mean monthly temperatures
Figure 4.8 Internal temperatures of the three zoning strategies and the
reference model under winter conditions. a) Maximum mean daily
temperatures, b) minimum mean daily temperatures c) maximum mean
monthly temperatures, d) minimum mean monthly temperatures
Figure 4.9 Monthly space heating energy demand of the three zoning
strategies
Figure 4.10 Possible geometry layouts and party wall location for the
dwellings in the dataset91
Figure 4.11 The methodology developed to model dwellings' geometry using
the reduced data available in the modelling dataset
Figure 4.12 Left to right: The reference model, the reference geometry model
and the design geometry model as created in DesigBuilder software package
Figure 4.13 Transforming the reference geometry model to design geometry
model
Figure 4.14 Monthly averaged space heating demand, external infiltration
and solar gain estimates of the design geometry model compared to the
reference geometry model
Figure 4.15 Internal temperatures of the reference geometry model and the
design geometry model. a) Maximum mean daily temperatures, b) minimum

mean daily temperatures c) maximum mean monthly temperatures, d) minimum mean monthly temperatures
Figure 4.16 Possible solutions to model extensions (red lines represent extension walls)
Figure 4.17 Modelling the extensions using five-vertex solution 100
Figure 4.18 The reference model with single storey and two storey extensions
Figure 4.19 Annual space heating demand comparison of the reference
model with extensions with five-vertex solution
Figure 6.1 Geometrical layout of the dwellings' model with red dots showing the vertices of the rectangular blocks, the origin vertex and the IDF notation
Figure 6.2 Translator framework to process varying data from XML files and write input data into Template IDF (TIDF)
Figure 6.3 Comparison of the space heating demands from different variants of the reference model
Figure 6.4 Comparison of the space heating demand from SAP, RdDEM and detailed DesignBuilder models of the three test houses
Figure 6.5 Comparison of space heating demand predictions for 83 modelled dwellings
Figure 6.6 Comparison of the distributions of the annual space heating demand predictions for the 83 modelled dwellings
Figure 6.7 Comparison of RdDEM mean monthly internal temperature predictions to SAP (The black line represents x=y)
Figure 6.8 Comparison of the distributions of mean monthly internal air temperature

List of Tables

Table 2.1 Advantages and disadvantages of the three major residential
modelling approaches [re-created from (Swan & Ugursal, 2009)]
Table 2.2 Comparative analysis of prominent UK based steady-state
domestic energy and carbon models
Table 2.3 Comparison of validation techniques (Judkoff 1988)
Table 3.1 Corresponding U-values (W/m ² K) of the wall types specified in the
dataset based on dwellings' age band (re-created from Table S6 SAP 2012
(SAP, 2012))
Table 4.1 Construction materials, each layer's thickness, physical properties
of building elements (density, thermal conductivity and specific heat capacity
Table 4.2 A&P rates and times of occurrence of internal heat gains from
occupants, cooking, lighting, refrigerator, television and hot water (re-created
from (Allen and Pinney, 1990))79
Table 4.3 Heating set-point temperature for individual rooms semi-detached
dwelling (re-created from (Allen and Pinney, 1990))
Table 4.4 Mean monthly temperature, wind speed and horizontal global
radiation of Kew67 Example Weather Year (EWY) weather file
Table 4.5 Heating set-points averaged over zone area in each zoning
strategy
Table 4.6 Space heating energy demand of the three zoning strategies
compared to the reference model90
Table 5.1 SAP equivalent external wall constructions, thickness and the
physical properties of each layer of materials 106

Table 5.2 Roof construction materials and corresponding density, thermalconductivity and specific heat capacity values108
Table 5.3 Ground floor construction materials and corresponding density,
thermal conductivity and specific heat capacity values
Table 5.4 Window area estimated based on age band and total floor area (re-created from Table S4 (SAP, 2012)111
Table 5.5 Mean monthly external temperature, wind speed and solar
irradiance from UK average weather data (SAP 2012) and IWEC Birmingham
(IWEC, 2001) weather data showing the conversion factors (CF) 114
Table 6.1 Comparison of space heating demand predictions between
RdDEM and SAP 125
Table 6.2 External wall improvements on the three test houses and
corresponding wall types, U-values, age bands, total areas, roof insulation,
and ground floor types
Table 6.3 Improved external wall constructions, thickness and the physical
properties of each layer of materials 133
Table 6.4 Annual space heating demands from SAP and RdDEM, before and
after external wall improvements in the test houses and the percentage
reductions

Abbreviations

ACH	Air Changes per Hour
ADEPT	Annual Delivered Energy and Temperature
ASHRAE	American Society of Heating Refrigeration and Air-Conditioning
A&P	Allen and Pinney
BERG	Building Energy Research Group
BGT	Building Generation Tool
BRE	Building Research Establishment
BREDEM	Building Research Establishment Domestic Energy Model
BREHOMES	Building Research Establishment's Hosing Model for Energy Studies
CDA	Conditional Demand Analysis
CDEM	Community Domestic Energy Model
CF	Conversion Factor
СНМ	Cambridge Housing Model
CHREM	Canadian Hybrid Residential End-use Energy and Emissions Model
CIBSE	Chartered Institution of Building Services Engineers
CREEM	Canadian Residential Energy End-use Model
CSDDRD	Canadian Single-Detached and Double/Row Housing Database
CSV	Comma Separated Value
DHW	Domestic Hot Water
DDM	Domestic Dwelling Model

DECC	Department of Energy and Climate	Change
------	----------------------------------	--------

- DEFACTO Digital Energy Feedback and Control Technology Optimisation
- DEN&T Departments of the Environment and Transport
- DOE Department of Energy
- DTI Department of Trade and Industry
- DETR Department of the Environment Transport and the Regions
- EDEM ESRU Domestic Energy Model
- EEP Energy and Environment Prediction Model
- EHCS English House Condition Survey
- EHS English Housing Survey
- EI Environmental Impact
- EPBD Directive on Energy Performance of Buildings
- EPC Energy Performance Certificate
- EPW EnergyPlus Weather
- ESRU Energy Systems Research Unit
- EU European Union
- EWY Example Weather Year
- gbXML Green Building XML
- GDP Gross Domestic Product
- HLP Heat Loss Parameter
- HPXML Home Performance XML
- HVAC Heating, Ventilation and Air-Conditioning
- IDF Input Data File (EnergyPlus)

IEA	International Energy Agency
IPCC	Inter-Governmental Panel on Climate Change
IWEC	International Weather for Energy Calculations
LARA	Local Area Resource Analysis
NCM	National Calculation Methodology
NN	Neural Network
ONS	Office for National Statistics
RdSAP	Reduced Standard Assessment Procedure
RdDEM	Reduced Data Dynamic Energy Model
SAP	Standard Assessment Procedure
STEP	Seasonal Temperature Energy Price
TFA	Total Floor Area
TIDF	Template IDF
TMP	Thermal Mass Parameter
TRV	Thermostatic Radiator Valve
UKDCM	UK Carbon Domestic Model
XML	Extensible Mark-up Language

1. INTRODUCTION

1.1. Background

The scientific community has widespread agreement that changes to the global climate are taking place, primarily due to an increase in anthropogenic greenhouse gases, and human societies will be required to adapt to these changes (Hulme & Jenkins, 1998; Royal Commission on Environmental Pollution, 2000; IPCC, 2001; and McCarthy et al., 2001). The main greenhouse gases are carbon dioxide (CO₂), methane (CH₄), nitrous oxide (N₂O), hydrofluorocarbons (HFCs), perfluorocarbons (PFCs) and sulphur hexafluoride (SF₆). Each of these gases adds to global warming to a varying extent. Carbon dioxide is the most important greenhouse gas, due to its vast concentration in the atmosphere and its long atmospheric lifetime (IPCC, 2001).

The scientific evidence in support of anthropogenic climate change is now overwhelming (IPCC, 2007). In the latest IPCC report (IPCC, 2017), the importance of preventing scenarios to be implemented by all governments is highlighted. The global community is warned that, failure to implement effective policies would result in significant changes. These changes including: a 2 to 3.5°C increase in average annual temperature by the 2080s; an increase in the frequency of high summer temperatures; an increase in winter rainfall; a rise in the relative sea level around most of the UK's shoreline; and an increase in the temperature of UK coastal waters (Hulme et al., 2002). All of these effects will have potentially significant socio-economic and political impact in the UK. If such disruption to the global climate system is to be minimised, significant reductions in CO₂ emissions will be required during this century (Johnston, 2003).

As a part of 2008 Climate Change Act, the UK government made a commitment to reduce the greenhouse gas emissions by at least 80% compared to 1990 levels by 2050 (Office of Public Sector Information, 2008). The Climate Change Act which was initially targeted to reduce emissions by

26% by 2020 was later tightened to 34% (Office of Public Sector Information, 2009). Figure 1.1 shows the contribution of each sector to the total UK carbon dioxide emissions.



Figure 1.1 Contribution of different sectors to the UK's total carbon dioxide emissions of 2011 (DECC, 2012)

Achieving the Climate Change Act targets will require substantial reductions in energy consumption in different sectors; though reductions in the domestic sector are considered to be "relatively low cost" and "realistically achievable" (Committee on Climate Change, 2008). Since 1990, emissions from fossil fuel use in the residential sector have fluctuated but in 2010 they were 8% above the 1990 level (DECC, 2011c). In 2010, the UK residential sector emissions of carbon dioxide increased by 13.4% compared to the previous year (the highest rise for any single sector) due to a considerable rise in residential gas use for space heating as 2010 was on average the coldest year since 1986 (DECC, 2011c). In 2013, the emissions from this sector were estimated to be 3% below the 1990 level (DECC, 2014).

The energy consumption of UK residential buildings accounts for 31% of national energy consumption (Figure 1.2), which is the largest proportion in Europe (Saidur et al., 2007). The UK's housing stock is one of the oldest and least efficient in Europe (Boardman et al., 2005) and the majority of energy

consumption in UK dwellings is due to space heating which in 2009 accounted for 61% of the total energy consumption in the domestic sector (DECC, 2011a).



Figure 1.2 Residential energy consumption shown as a percentage of national energy consumption and in relative international form (Saidur et al., 2007)

The recognition of the domestic sector's significance in reducing greenhouse gas emissions has led to an ongoing development of technologies to reduce energy demand from the housing stock. In order to develop the policy required to meet the criteria set by the Climate Change Act, it is important to try to predict the potential impact that these technologies can have on energy consumption and the resultant CO₂ emissions. The reliable prediction of the impact of the energy saving technologies requires a detailed model of energy and emissions from the housing stock.

The increasing power of personal computers, their reasonable price and increasing need for computer-based analysis have resulted in a considerable increase in the number of building energy-analysis tools in recent years (Neymark et al. 2001). An on-line directory (BETD, 2014) supported by US Department of Energy (DOE) lists more than 200 building energy software

tools developed worldwide that have thousands of users. The main reason for the increasing number of building energy software tools is regulations imposed by governments which require various government bodies and organisations to use computer-based analysis (Raslan & Davies 2012). A good example is the Directive on Energy Performance of Buildings (EPBD) which was introduced by the European Union (EU) to speed up the process of adopting performance based energy standards in buildings. The EPBD urges member states to adopt a National Calculation Methodology (NCM) in order to demonstrate that buildings comply with energy performance standards and to provide computer based tools to enable this (Raslan & Davies 2012).

1.2. Justification of the Research

The National Calculation Methodology (NCM) for dwellings in the UK is SAP 2012 (SAP, 2012), which is based on Building Research Establishment Domestic Energy Model (BREDEM) and uses monthly steady-state calculations to estimate energy consumption and carbon emissions. Many studies have investigated modelling tools and their capabilities (Crawley et al., 2008; Gale, 1990; Kenny and Lewis, 1995) and still there is debate over the suitability of steady-state models in policy making decisions (Schwartz and Raslan, 2013). The steady-state models are not capable of taking into consideration the complex, interdependencies, and dynamic nature of the energy consumption and carbon emission (Oladokun and Odesola, 2015). A recent study (Kane et al. 2015) criticised the use of simple steady-state models by stating that these kind of models were used traditionally to enable the equations to be solved manually. However, we no longer need to understand mathematics of model and it is time to adopt a more realistic approach to modelling housing stock.

Dynamic modelling of the UK dwellings, using well-established simulation software such as EnergyPlus (EnergyPlus, 2015) has the potential to overcome these shortcomings and provide insight into the transient energy-

use and thermal behaviours, such as peak heating loads and indoor air temperature extremes. The main issue associated with use of dynamic modelling of dwellings is the increased amount of input data required, compared to simple steady-state models. There are currently no datasets of the UK housing stock that contain sufficient data for dynamic simulation, and considerable additional expense would be required to collect all of the additional detailed data required to achieve this.

There has been little research to investigate the possibility of developing robust dynamic energy models of UK dwellings using only the available reduced datasets. Therefore, this research was conducted to answer the following questions:

- i. Is it technically feasible to develop robust dynamic energy models of UK dwellings using available (reduced) datasets?
- ii. How close will the estimates of such dynamic energy models be to the results of equivalent steady-state models and to more detailed dynamic energy models?

1.3. Aim and Objectives

The overall aim of this research was to develop a Reduced-data Dynamic Energy Model (RdDEM) that is capable of simulating the transient energy and thermal behaviours of UK dwellings.

The following objectives were undertaken to meet this aim:

 Identify and review literature on the modelling approaches that could be used to forecast energy use and CO₂ emissions in the UK dwellings, and perform a critical analysis of the work that has been undertaken to utilise available reduced data for energy modelling purposes.

- Identify a suitable dataset of UK dwellings to be used as the source of modelling data.
- 3) Develop and test a data preparation process that will enhance the reduced data in order to produce an equivalent set of detailed data that is suitable for dynamic energy simulation.
- 4) Develop and run a reduced data dynamic energy model of UK dwellings that translates the prepared data into a form suitable for dynamic simulation using established models.
- 5) Compare the results of the reduced data dynamic energy model with those from equivalent steady-state models.

1.4. Contribution to Knowledge

This is, to the author's knowledge, the first dynamic energy modelling exercise to be undertaken using the EPC data for a set of UK dwellings. Automating the data enhancement and translation processes eliminates the additional time and cost associated with collecting further data and modelling each of the homes individually. The reduced data dynamic energy model introduces a set of new methods for enhancing reduced data in order to create equivalent detailed geometry and zoning information. These methods enable a unique inter-model comparison to be carried out across 83 buildings, demonstrating that similar results can be obtained from different models when the inputs are equivalent. Ultimately, the techniques developed here can be used to provide new insights into the transient aspects of energy use and indoor air temperatures in the UK housing stock and therefore, the model has value as both a policy and a research tool.

1.5. Outline of the Thesis

- Chapter 2 presents the literature review which was conducted for this study. This covers the existing steady-state and dynamic energy models developed for the dwellings with focus on strength and weakness of each model, technical specifications, and the input data sources.
- **Chapter 3** provides an overview of the methodology to develop the reduced data dynamic energy model and describes the source of the input data.
- **Chapter 4** describes the zoning and geometry enhancements made to the reduced input data in order to develop detailed geometry and zoning information.
- **Chapter 5** describes the process of developing the equivalent construction materials and, inside and outside boundary conditions that were suitable for dynamic energy simulation.
- **Chapter 6** describes the translation process and how the modelling process was verified. The chapter also presents the results and compares them to equivalent steady-state models.
- Chapter 7 discusses the achievements against the aim and objectives alongside the contributions to knowledge. The limitations of the work are quantified, and the application of the modelling framework developed in this thesis is expanded for academia, policy makers and industry.
- **Chapter 8** summarises the main conclusions from the studies and investigations undertaken as part of this thesis.

2. LITERATURE REVIEW

2.1. Introduction

This chapter presents the context for development of the Reduced Data Dynamic Energy Model (RdDEM) of the UK dwellings, in pursuit of Objective 1 (*Identify and review literature on the modelling approaches that could be used to forecast energy use and* CO₂ *emissions in the UK dwellings, and perform a critical analysis of the work that has been undertaken to utilize available reduced data for energy modelling purposes*). The chapter starts with a description of the Energy Performance Certificates (EPC) which provide a rich source of reduced-data for the UK housing stock. The chapter continues with a critical analysis of existing modelling techniques for dwellings and also other models that have used reduced-data datasets. The detailed description and critical analysis of eight prominent steady-state models developed for the UK dwellings as well as four dynamic energy models of the dwellings are also provided in this chapter. The chapter concludes with a description of data translators that have been created for the purpose of energy modelling of the housing stock.

2.2. Energy Performance Certificate (EPC)

2.2.1. Context

Following the oil crisis of 1972-1979, the first World Climate Conference took place in 1979, which was consequently resulted in the setting up of the Inter-Governmental Panel on Climate Change (IPCC) in 1988. The IPCC's first report assisted the formation of the United Nations' Framework Convention in 1992 in Rio. The IPCC's second report, which was published in 1995, highlighted the 'human influence' on global warming. It was this report that eventually resulted in the creation of the Kyoto Protocol of 1997 (which came

into force in 2005). The initial form of Kyoto Protocol insisted on overall 5% reduction in greenhouse gas emissions (Watson, 2009).

The EU produced a European directive: '2002/91/EC Energy Performance of Buildings' to assist in achieving the Kyoto protocol goals (European Union, 2002). This was followed in the United Kingdom by development of the White paper: 'Our Energy Future – Creating a Low Carbon Economy' in 2003, which insisted on reducing CO₂ emissions by 60% by 2050, compared to 1990 levels. The EU directive led to a series of legislation across Europe, as the main requirement, which was to produce Energy Performance Certificates (EPCs) for dwellings and residential buildings.

In the UK, the Housing Act (2004) for the first time made it compulsory to present EPC and improvement recommendations when an existing home was let or sold. In 2006 it was also made compulsory for new built dwellings to have EPC. The EPCs include both energy rating and Environmental Impact (EI) rating (Figure 2.1). A full example of an EPC is presented in APPENDIX A.



Figure 2.1 Example of EPC energy and EI ratings (RdSAP Manual, 2012)

The energy and EI ratings are based on the energy costs and the annual CO₂ emissions associated with space heating, water heating, ventilation and lighting, less cost savings from energy generation technologies. Both ratings

are expressed on a scale of 1 to 100. The higher the energy rating the lower the running costs. In case of the EI rating, the higher the number the better the standard. The ratings are adjusted for floor area so that it is essentially independent of dwelling size for a given built form (SAP, 2012).

2.2.2. National Calculation Methodologies: SAP and RdSAP

In order to meet the requirements of the EU directive for providing EPCs on all buildings in England and Wales, a National Calculation Methodology was developed. This methodology contains two different approved calculation methods for dwellings: Standard Assessment Procedure (SAP) and Reduced Data SAP (RdSAP). SAP was introduced in the 1995 building regulations in order to represent the compliance with energy efficiency standards in UK domestic sector and in 2005 RdSAP was introduced as a lower cost method of assessing energy performance of existing UK dwellings.

i. Standard Assessment Procedure (SAP)

The Standard Assessment Procedure (SAP) underpins the Building Research Establishment Domestic Energy Model (BREDEM), and is the UK government's approved methodology for assessing the energy ratings of dwellings. Designs for new domestic buildings and those with major renovations are evaluated according to SAP for their estimated energy consumption and carbon emissions. SAP has been an important methodology in delivering a number of key energy and environmental policy initiatives including Energy Performance Certificates (EPCs).

As is the case for standard energy models in many countries, the calculations and assumptions contained in SAP are laid out in freely available documentation. However the implementation of SAP in approved software such as BREDEM is a 'black box', since it is impossible to inspect directly the implementation (Summerfield et al., 2011).

The SAP calculation is based on the energy balance taking into account a range of factors that contribute to energy efficiency (SAP, 2012) including:

- Materials used for construction of the dwelling.
- Thermal insulation of the building fabric.
- Air leakage ventilation characteristics of the dwelling, and ventilation equipment.
- Efficiency and control of the heating system(s).
- Solar gains through openings of the dwelling.
- The fuel used to provide space and water heating, ventilation and lighting.
- Energy for space cooling.
- Renewable energy technologies.

The SAP calculation is independent of factors related to the individual characteristics of the household occupying the dwelling like: household size and composition; ownership and efficiency of particular domestic electrical appliances, and individual heating patterns and temperatures (SAP, 2012).

ii. Reduced Data Standard Assessment Procedure (RdSAP)

SAP usually requires hundreds of input parameters and it is too complex and time consuming (and therefore expensive) to collect these data for assessment of existing dwellings. Hence, a Reduced Data SAP (RdSAP), which required considerably less input parameters, was developed for assessment of existing dwellings. RdSAP was developed in the form of a spreadsheet which can be accessed and used from <u>www.bre.co.uk/SAP2012</u>. The RdSAP spreadsheet is presented in APPENDIX B. The RdSAP employs extensive inference algorithms which automatically deduce the missing data (RdSAP manual, 2012).

The RdSAP is also a system of dwelling data collection, together with defaults and inference procedures that generates a complete set of input data for calculation. The calculation using the reduced data is done in two stages. First the reduced data set is expanded into a full data set, and then the SAP calculation is undertaken using the expanded data set. The actual SAP calculation is therefore identical, whether starting from a reduced data set or a full data set.

Since the EPCs were introduced and were made a mandatory requirement for all dwellings sold or rented in England and Wales, researchers have investigated different aspects of this initiative: from technical advantages and deficiencies to its effectiveness and impact on energy demand and carbon emissions reduction. Such research can be categorised into three main groups: research investigating the impact of EPCs on households, research investigating the impact of EPCs on energy demand and CO₂ emissions, and research investigating the technical aspects of EPCs.

The first group has mainly looked into social and psychological impacts of EPCs and the resultant changes made to occupants' behaviour and space heating habits. An example of such research is the work carried out by Watts et al. (2011) on 2000 households in Southampton on the South coast of England which presented the results of a questionnaire survey with response rate of 17%. The authors found out that EPCs had little impact on decision making or price negotiation. Where retrofitting measures have been undertaken, results were inconclusive as to whether retrofitting was done as a result of EPCs. Energy efficiency was not found to be a priority for home buyers. The authors conclude: "*Whilst there is an awareness of the scheme in general, there appears to be limited recognition of its potential*" (Watts et al., 2011).

The second group has mainly focused on the implications of EPCs in reducing future energy demand and carbon dioxide emissions from domestic and non-domestic buildings. The majority of such research highlighted the performance gap between real life data and EPC estimates. The magnitude of this gap is significant, with reports suggesting that the measured energy use can be as much as 2.5 times the predicted energy use (Menezes et al., 2012). Energy efficiency is only one of the various performance aspects of buildings; it is highly likely that similar performance gaps exist between predicted and measured indoor air quality, thermal comfort, acoustic performance, daylighting levels and others (De Wilde, 2014).

The last group of research provide insight to areas of EPCs needing improvement. Two of the main works in this group are the ones undertaken

by Khayatian et al. (2016) and Koo and Hong (2015). The former employs a neural network evaluation technique to investigate accuracy of residential EPCs. The latter, on the other hand, employs a dynamic rating system to overcome shortcomings of conventional EPCs.

The methodology of employing Neural Networks for EPC prediction, suggested by Khayatian et al. (2016) offers an autonomous approach for detecting anomalies in building energy certificates. The developed model provides the opportunity to compare the predicted energy performance index with dynamic energy simulation tools. This process can define a correlation between steady-state and dynamic simulations with the intention of obtaining a more realistic overview on the energy consumption trend in the region. Furthermore, the provided methodology can be adopted to predict the actual energy performance of buildings.

This study by Koo and Hong (2015) analysed the potential problems of the conventional operational rating system for existing buildings by using the statistical and geostatistical approaches and developed the dynamic operational rating (DOR) system by using the data-mining technique and the probability approach. The developed DOR system can be used as a tool for building energy performance diagnostics. The developed DOR system can be applied for various purposes such as encouraging all the public to voluntarily participate in energy-saving campaigns, evaluating the historical trend in the energy performance of existing buildings, estimating the operational ratings of new buildings in the early design phase.

2.3. Energy Modelling in Domestic Buildings

In recent years, a variety of models have emerged that are capable of analysing energy and environmental performance of domestic buildings, and examining different strategies that are designed to reduce energy consumption and improve internal thermal comfort. These models are capable of modelling the complex interactions which occur between building envelope and systems with external weather conditions, investigate the influence of various energy demand reduction measures and policies, and suggest the resultant impact that these strategies and policies may have on future energy use. As a result, these models are needed in order to understand which strategies and policies are required, when such strategies and policies should be implemented, and estimate the potential impact of their implementation. The available models range from global, international and national energy models, to more detailed sectoral models. Such diversity has resulted in these models varying considerably in terms of their level of detail, their complexity, the data input required by the user, the time periods covered, their geographical coverage, and the methodological approach taken (Johnston, 2003). Since describing and critically analysing such wide variety of models is beyond scope of this theses, the remainder of this chapter investigates the main energy models which are considered to be of particular relevance to this study, namely: those that have been developed for domestic buildings in the UK using a limited set of data.

Energy models of the domestic building stock have been developed for a number of reasons which include: predicting energy demand in dwellings to adjust energy supply, identifying the subdivisions of society that consume more energy and enable policy makers to target high energy users better when designing demand reduction measures and assessing the potential of energy efficiency policy (Kane, 2013).

Techniques to model domestic energy consumption can be broadly divided into two main groups: "top-down" and "bottom-up". The terminology is with reference to hierarchal position of input data as compared to the housing sector as whole. The top-down approach considers the residential sector as energy sink and is not concerned with the individual dwellings (Swan & Ugursal, 2009). It uses historical statistics of energy use and households on a national level and estimates the effect of changes in top level factors such as energy price, climate and macroeconomic indicators such as gross domestic product, unemployment and inflation on energy consumption of the whole housing stock (Swan & Ugursal, 2009). Bottom-up models are based on the principles of building physics and are able to quantify specific changes to the domestic building stock such as the impact of a national roof insulation programme.

The main difference between these two approaches is the adopted perspective (Figure 2.2).



Figure 2.2 Top-down and bottom-up modelling approaches (IEA, 1998)

Top-down methods start with aggregate information and then disaggregate down as far as they can. Since they begin with aggregate data, top-down methods provide a comprehensive approach to modelling. Bottom-up methods, on the other hand, start with detailed disaggregated information and then aggregate this information as far as they possibly can. Since bottom-up approach only tends to model part of the whole picture, they lack the comprehensiveness of top-down methods.

Although top-down and bottom-up methods are two different approaches, they have some similarities. First of all, they are capable of operating at the same level of disaggregation, and secondly, they both make use of the same facts, but describe and use them in different ways (Johnston, 2003).

These two modelling approaches are introduced and examples of where they inform this work given in following sub-sections.

2.3.1. Top-down Modelling Approach

Top-down models take a macroeconomic approach to modelling energy supply and demand. They concentrate on the interaction between the energy sector and the economy at large, and use econometric equations to model the relationships that exist between the energy sector and economic output. They rely on aggregate economic factors to predict future changes in energy use and CO₂ emissions (IEA, 1998). Top-down models have been developed to inform policy makers regarding the social and economic drivers for energy consumption. They seek to improve understanding of how energy use relates to geographical areas, economic factors, and demographics; how this has changed historically and what impact policy instruments might have on future energy use in different segments of the population (O'Neill & Chen, 2001). Consequently, top-down models purposefully exclude detailed technology descriptions, as their focus is not on the individual physical factors that can influence energy demand, but rather on the macroeconomic trends and relationships (MIT, 1997). Therefore, the data input required for top-down

models consist of econometrically based data, such as Gross Domestic Product (GDP), fuel prices, and income.

Top-down modelling techniques are mainly appropriate for modelling the societal cost-benefit impacts of different energy and emissions policies and scenarios (MIT, 1997). Therefore, they have been widely used in the UK by Government organisations to identify future trends in energy use and CO₂ emissions and to study the effects of macroeconomic policy decisions on the various factors that drive energy use. Such organisations include: the Department of Energy in Energy Papers 39 and 58 (DoE, 1979a, 1979b & 1990); the Departments of the Environment and Transport in Research Report 33 (DEn&T, 1981); the Department of Trade and Industry in Energy Papers 59, 65 and 68 (DTI, 1992, 1995 and 2000); and by the Department of the Environment Transport and the Regions (DETR, 1998). However, the models developed by these organisations generally do not identify energy demand at a technological or process specific level (DTI, 1995).

The most significant UK based top down model is MARKEL which is used as a core policy tool for the UK Government (Kannan et al., 2007) and has been used to establish pathways to the achieve CO₂ emissions reduction goal by 2050 (DECC, 2011b). Druckman and Jackson (2008) developed a socioeconomic model of the UK Local Area Resource Analysis (LARA) which calculates CO₂ emissions at the national and regional levels. Summerfield et al. (2010) developed two regression models to predict future UK energy demand. The first model, Annual Delivered Energy and Temperature (ADEPT), used linear regression on data available from 1970 and the second model, Seasonal Temperature Energy Price (STEP), used a polynomial regression and was based on quarterly energy data from 1998.

The strength of top-down modelling approach is that they only require historical aggregate data which is largely available. The dependence on aggregate data is also a disadvantage for top-down approach as it doesn't allow modelling discontinues advances in technology. Besides, the lack of information on individual end-use energy consumption eliminates the ability to identify main areas for energy demand reduction (Swan & Ugursal, 2009).
2.3.2. Bottom-up Modelling Approach

The bottom-up approach includes all models that use input data from a hierarchal level less than that of the sector as a whole. Bottom-up modelling approach takes a disaggregated approach to modelling energy supply and energy demand. Models using bottom-up approach can account for the energy consumption of individual end-uses, individual houses, or groups of houses and are then extrapolated to represent the region or nation (Swan & Ugursal, 2009).

Based on the data inputs used by each model, the bottom-up approach can be divided into a number of sub-groups (Figure 2.3). The common data input for bottom-up models are dwelling properties such as: geometry, construction materials, appliances and systems, weather data, internal temperatures and occupancy patterns. This high level of details is both strength and weakness of the bottom-up approach. High level of details allows modelling new technologies and identifying the areas of improvements. However, the input data requirement of such high level of details is considerably greater than top-down models and simulation techniques are more complex.



Figure 2.3 Bottom-up modelling approach and its sub-groups based on input data [re-created from (Swan & Ugursal, 2009)]

Statistical methods use historical data and regression analysis to assign dwellings energy consumption to individual end-uses. Once the relationships between end-uses and energy consumption have been established, the model can be used to predict the energy consumption of domestic building stock. Engineering method, on the other hand, explicitly takes the energy consumption of end-uses into account. In this method, consumption values are estimated based on use of equipment and systems, and heat transfer and thermodynamic relationships (Swan & Ugursal, 2009).

Since customer energy billing information is stored in vast quantity worldwide, researchers have applied different statistical methods to regress energy consumption as a function of dwellings characteristics. The main advantage of statistical method is the ability to separate the effect of occupant behaviour from model (Swan & Ugursal, 2009). This provides great benefits in domestic stock modelling as occupant behaviour has been found to vary widely and is poorly represented by simplified assumptions (Seryak and Kissock, 2003; Lutzenhiser, 1992; Emery and Kippenhan, 2006).

Three well-documented statistical method techniques as identified in Figure 2.3 are: Regression, Conditional Demand Analysis (CDA) and Neural Network (NN). Models employing the regression technique regress the aggregate dwelling energy consumption data onto parameters which are expected to affect energy consumption and the input data with negligible effect are removed from model for simplicity. The CDA technique regress total dwelling energy consumption onto the list of owned appliances. In order for the CDA technique to produce reliable results, data from hundreds or even thousands of dwellings are required based on number of variables used. The NN method uses a simplified mathematical model based on interconnected parallel structure of neural networks (Swan & Ugursal, 2009). Among the three statistical method techniques, regression is the least favoured due to widely varying input parameters in different models and limited comparison possibility. The CDA, however, is focused on simplified end-uses and its predictions a comparable among different studies.

The engineering method, in contrary to statistical method, can estimate the domestic energy consumption without any historical consumption data. The engineering method has high capability in modelling new technologies with no historical data. However, occupant behaviour must be included in this method. The main engineering method techniques (Figure 2.3) are: Distribution, Archetypes and Sample. The distribution technique calculates end-use energy consumption of appliances based on common appliances

ratings and doesn't account for interactions between appliances. The archetype technique broadly classifies the housing stock according to vintage, size, etc. and scales up the energy consumption estimates from each archetype to represent national housing stock. The sample technique uses the actual sample house data to model energy consumption (Swan & Ugursal, 2009). As the variety of houses vary in UK, this technique requires a large database of representative dwellings. However, if the aim is to study new technologies and their impact on energy consumption, the engineering methods are currently the only option. The main drawbacks of the engineering method are: 1) assumption of occupant behaviour; and, 2) the high level of expertise required in developing such models.

The data input required for all bottom-up models largely consists of quantitative data on physically measurable variables like the thermal performance of walls, the efficiency of a space heating systems, or the specific energy consumption of appliances. Economic variables, such as income and fuel prices, are not explicitly modelled within bottom-up methods. Instead, they are incorporated within the model in terms of their effect on physically measurable variables, such as mean internal temperatures, the ownership and usage of appliances and the different fuels that are used (Johnston, 2003).

The use of physically measurable data within bottom-up modelling techniques has resulted in these techniques being widely used to suggest the likely outcome of policies, or to identify a range of technological measures that are intended to improve end-use efficiencies (Shorrock, 1994). Consequently, over the last 30-40 years, bottom-up models have been extensively developed and used by a number of researchers in the UK: Leach et al. (1979); Barrett (1981); Olivier et al. (1983); Evans & Herring (1989); Shorrock & Henderson (1990); ETSU (1994); Shorrock (1994 & 1995); DECADE (1994, 1995 & 1997); Evans (1997); Shorrock & Dunster (1997); Hay et al. (1999); ECI (2000); Letherman & Samo (2001), Shorrock et al. (2001); and, Johnston (2003).

2.3.3. Top-down or Bottom-up Approaches?

The top-down and bottom-up approaches each have many commonalities, similarities and differences, as well as advantages and disadvantages. Although each approach views the domestic energy sector from a different perspective, they are in fact complementary to one another, and each method can give insights into a particular problem that the other may miss (Shorrock & Dunster, 1997). Consequently, no approach is clearly superior to the other. Two of the main characteristic differences between these two approaches are the required input data and range of energy saving scenarios that can be implemented in each approach. The main advantages and disadvantages of the three major residential energy modelling approaches (top-down, bottom-up statistical and bottom-up engineering) are summarised in Table 2.1.

Swan and Ugursal (2009) discussed all three modelling methods and concluded that the models are useful considering the current focus placed on efficient use of energy and technology implementation. When the supply side is the main point of consideration, top-down models provide great advantages compared to bottom-up methods. Bottom-up statistical method takes into account the occupants behaviour and individual appliances which helps to identify areas of improvement in energy consumption. Finally, bottom-up engineering models help identify the impact of new technologies and account for wide variety of housing stock.

Despite the comprehensiveness and macroeconomic coherence that this approach provides, there are important limitations that make top-down approach inappropriate for this study. Firstly, top-down models employ historical economically based input data which is contrary to aim of this PhD in using reduced data available on individual dwellings. Secondly, top-down models lack the level of technological detail that is contained within bottom-up methods. As a result, top-down methods tend to parameterise technological advance, rather than explaining it within the model (MIT, 1997); which makes them inappropriate for identifying energy saving potentials of various energy saving measures.

Table 2.1 Advantages and disadvantages of the three major residential modelling approaches [re-created from (Swan & Ugursal, 2009)]

Method	Advantage	Disadvantages		
Top-down	 Long term forecasting in the absence of any discontinuity Inclusion of macroeconomic and socioeconomic effects Simple input information Encompasses trends 	 Reliance on historical consumption information No explicit representation of enduses Coarse analysis 		
Bottom-up statistical	 Encompasses occupant behaviour Determination of typical end-use energy contribution Inclusion of macroeconomic and socioeconomic effects Uses billing data and simple survey information 	 Multicollinearity Reliance on historical consumption information Large survey sample to exploit variety 		
Bottom-up engineering	 Model new technologies "Ground-up" energy estimation Determination of each end-use energy consumption by type, rating, etc. Determination of end- use qualities based on simulation 	 Assumption of occupant behaviour and unspecified end-uses Detailed input information Computationally intensive No economic factors 		

Bottom-up methods, on the other hand, provide a number of advantages which makes them more appropriate for this research. Firstly, they allow use of physically measurable input data, which is the main type of data in EPC datasets. Secondly, bottom-up methods explicitly calculate energy consumption of different end-uses based on detailed description of houses (Swan & Ugursal, 2009). Finally, and most importantly, bottom-up methods can explicitly evaluate and calculate the impact of different energy saving measures on delivered energy use (Johnston, 2003).

As with all modelling approaches, there are limitations associated with bottom-up methods. The bottom-up methods tend to increase the amount of input data that is required by the modelling system, due to the disaggregation of information, and the greater complexity associated with this (Johnston, 2003). The increased amount of required input data is reported as one of the main barriers to modelling domestic buildings by number of previous studies (Judkoff and Neymark, 1995; Karlsson et al., 2007; Kalema et al., 2008; Judkoff et al., 2008).

The vast majority of UK based domestic energy models have dealt with the issue of lacking required data by employing steady-state calculation engines which require relatively less amount of input data compared to dynamic energy models. However, a few studies have been identified outside UK which have employed dynamic energy modelling and have dealt with inadequate input data in different ways. Following section presents critical analysis of both steady-state and dynamic models developed within and outside the UK.

2.4. Reduced Data Building Energy Models

2.4.1. Steady-State Models

In the UK a number of energy models have been developed in past decades targeting to estimate domestic sector energy consumption as well as to predict future residential energy demand. Eight main energy models developed for the UK housing stock were identified. Critical analysis of these models provides a great opportunity to identify the weakness and strength of different modelling methodologies applied to same building stock. The identified models are described and analysed as bellow:

 The Building Research Establishment's Hosing Model for Energy Studies (BREHOMES) developed by Shorrock and Dunster (1997a, 1997b, 2005)

The BREHOMES model disaggregate the UK housing stock into more than 400 categories which are separated by 4 age bands, 17 built forms and by whether or not central heating is present (Shorrock and Dunster, 1997). The categories into which the model disaggregates are mainly based on data source used. The large number of categories in BREHOMES is made possible through employing more than one data source which makes BREHOMES a database as well. The most important sources used in developing BREHOMES are: the Digest of UK Energy Statistics published annually by Department of Energy, the English House Condition Survey (EHCS) published every five years by Department of Environment, the Central Household Survey published annually by office of Population Census and Surveys, and the Family Expenditure Survey published annually by Department of Employment.

The main drawback associated with this model is using a single dwelling type to predict future trends in all the stock which results in simplified calculations at the cost of the thoroughness (Natarajan and Levermore, 2007). Figure 2.4 presents the overall structure and form of the BREHOMES model.



Figure 2.4 Structure and form of BREHOMES model (Shorrock and Dunster, 1997)

As shown in Figure 2.4, central to the BREHOMES model is the Building Research Establishment's Domestic Energy Model (BREDEM) (Dickson et al., 1996). BREDEM consists of a set of heat balance equations and empirical relationships to estimate annual (in case of BREDEM-12 (Anderson et al., 2002)) or monthly (in case of BREDEM-8 (Shorrock et al., 1991; Anderson and Chapman, 2002)) energy consumption of individual dwellings. An important modified version of BREDEM (BREDEM-9) forms the basis of the UK Government's Standard Assessment Procedure (SAP) (BRE, 2005) which is used for identifying energy rating of dwellings and creating Energy Performance Certificates (EPC).

BREHOMES examined two illustrative scenarios with 1996 as their base year and projected scenarios to 2050: 'Reference' scenario and 'Efficiency' scenario (the same, but the uptake of efficiency measures, such as loft insulation, is increased). These scenarios have been improved and used by Johnston (2003) to investigate demand- and supply-side of domestic energy sector with regards to the UK Government's plans to cut CO₂ emissions.

ii. The Johnston model developed by Johnston (2003)

The Johnston model explores the technological feasibility of achieving CO_2 emission reductions in excess of 60% within the UK housing stock by 2050. This model investigates three illustrative scenarios of energy use and CO_2 emission: a 'Business-as-Usual' scenario, which represents a continuation of current trends in fabric, end-use efficiency and carbon intensity trends for electricity generation; a 'Demand Side' scenario, which represents the improvements to current rate of uptake of fabric and end-use efficiency measures; and an 'Integrated' scenario which shares the same demand side assumptions as the 'Demand Side' scenario, but represents what may happen if the carbon intensity of electricity generation were to fall even further (Johnston, 2003). The Johnston model employ BREDEM-9 calculation engine to estimate energy and CO_2 emission of the UK housing stock (Figure 2.5).



Figure 2.5 Structure and form of Johnston model (Johnston, 2003)

In contrary to BREHOMES, the Johnston model has a very low level of disaggregation and is constructed around only two notional dwelling types; namely: pre- and post-1996. The drawback caused by low level of disaggregation is that the model only provides broad results when comparing impact of different energy saving measures. Johnston (2003) discuss that two notional dwelling types makes it difficult "*to explore*

what reductions in energy consumption and CO₂ emission could be achieved if different age classes of the UK housing stock were selectively upgraded or demolished". The main reason behind low level of disaggregation is usually the lack of sufficient data to support high level of disaggregation as acknowledged by Johnston (2003). The Johnston model has used a number of data sources including: population projections from the Office for National Statistics, mean household size data from the Department of Environment, Transport and the Regions (DETR) and, the English House Condition Survey (EHCS).

iii. The UK Carbon Domestic Model (UKDCM) developed by Boardman (2007)

The UKDCM is basically a numerical model of energy flows, which takes into account all the sources of heat gain and heat loss in a stock of dwellings whose characteristics change through time (Figure 2.6).



Figure 2.6 Structure and form of the UKCDM (Boardman, 2007)

Similar to the BREHOMES and the Johnston model, the UKCDM has 1996 as the base year and projected scenarios to 2050 and makes use of government statistics on energy and housing, as well as population. The UKCDM disaggregate the UK housing stock into 9 regions, 12 age bands, 10 dwelling types, 6 tenure types, 4 classes for number of floors, and 6 construction types. In addition to these classification parameters, double glazing, loft insulation and wall insulation have been included as defining characteristics of the energy performance of individual dwellings (Boardman, 2007).

iv. The DECarb Model developed by Natarajan and Levermore (2007)

The DECarb is another prominent model for the UK housing stock which is capable of implementing different energy saving and CO₂ reduction scenarios in order to predict future trends in consumption and emissions. The DECarb model is based on building physics approach. Figure 2.7 shows the structure and form of the model. The DECarb model is highly disaggregated and has unique 8064 combinations of six age bands of the UK housing stock. Similar to models discussed above, the DECarb model uses the BREDEM-8 procedure for calculating consumption and emissions of the dwellings, and has 1996 as the base year (Natarajan and Levermore, 2007).



Figure 2.7 Structure and form of the DECarb model (Natarajan and Levermore, 2007)

The DECarb model has examined the scenarios developed by BREHOMES, Johnston, and UKDCM instead of adding further scenarios. The findings suggest that neither of the two low carbon scenarios tested with the Johnston model would reach the target of 50% reduction in carbon emissions by 2050. The results from the DECarb model, however, approve the UKDCM's 40% scenario of achieving the targeted 60% CO₂ emission reduction by 2050 (Natarajan and Levermore, 2007).

v. The Energy and Environment Prediction Model (EEP) developed by Jones, Patterson and Lannon (2007)

The EEP model is different to the previously introduced models in that it is based on Geographic Information System (GIS) techniques (not BREDEM) and employs a number of sub-models (Figure 2.8) to estimate current energy consumption and CO₂ emissions from domestic and non-domestic buildings, traffic and industrial processes for a city or region (Jones et al., 2007). Each sub-model uses UK Government's accepted procedures to predict energy use and emissions with the exception of the traffic sub-model that has been developed using Spatial Analysis procedures.



Figure 2.8 Structure and form of the EEP model (Jones et al., 2007)

The domestic energy use sub-model of EEP consists of 1300 dwellings. This sub-model employs the SAP rating as an indicator of the energy performance of dwellings. The SAP rating is calculated based on the UK Government's Standard Assessment Procedure which assesses the energy performance of dwellings (BRE, 2005). There is an important weakness in the SAP procedure regarding the way electrical appliances and cooking is handled. SAP takes into account the internal heat gain from electrical appliances and cooking but excludes their energy use in the calculation of total energy consumption. This issue was not explicitly addressed in the domestic sub-model of EEP. Although this sub-model has been evaluated against the SAP ratings, the validation of the actual energy consumption and CO₂ emission estimates was not clearly described (Oladokun and Odesola, 2015).

vi. The Community Domestic Energy Model (CDEM) developed by Firth, Lomas and Wright (2010)

The CDEM is another prominent model of energy consumption and carbon emissions of the UK housing stock that was developed by the staff in the Building Energy Research Group (BERG) of Department of Architecture, Civil and Building Engineering, Loughborough University in 2009 (Firth et al., 2010). This model is highly disaggregated, with 47 house archetypes derived from unique combinations of built form and dwelling age bands. For energy and emission calculations, the model requires input from many sources including English House Condition Survey (EHCS), Standard dwelling types (Allen and Pinney, 1990), BREDEM-8 (Shorrock et al., 1991; Anderson and Chapman, 2002) calculation engine, SAP rating (BRE, 2005), and etc. (Figure 2.9).

The main data requirement for the CDEM was provided from the BREDEM-8 calculation engine, monthly average external temperatures and monthly average solar radiation which is available from the Met Office. The model estimates monthly energy consumption and carbon emissions of the whole UK housing stock.

The main drawback associated with the CDEM is that instead of examining future scenarios, the model only estimated energy consumption and carbon emissions of the 2001 English housing stock (Firth et al., 2010).



Figure 2.9 Structure and form of the CDEM (Firth et al., 2010)

Unlike, other UK models the CDEM investigated effect of the uncertainties on the results associated with the input variables. Frith et al. (2010) carried out an extensive local sensitivity analysis and assigned sensitivity coefficients to the primary input parameters of the model. They found that the various input parameters have widely varying effects on the prediction outputs. The characteristics and usage patterns of heating systems (such as the thermostat temperature and hours of heating use) and the heat losses of the dwellings were identified as highly determining factors of domestic space heating demand.

vii. The Cambridge Housing Model (CHM) developed by Hughes (2011); Hughes and Palmer (2012)

The CHM is another steady-state bottom-up model that uses the calculations formulated and established by SAP 2009 (BRE, 2011) and BREDEM engine (Shorrock and Dunster, 1997b) to run necessary calculations. The model has three basic data input components (Figure 2.10); namely: climate data, housing data, and building physics data. For climate data input, the model uses SAP's monthly solar declination and regional latitude data, BREDEM-8's monthly/regional solar radiation data, and monthly/regional year-specific wind speed and external temperature data from a number of different stations across the UK.



Figure 2.10 Structure and form of the CHM (Hughes, 2011)

The main source of housing data is based on 16,670 dwellings defined in English Housing Survey of 2010 (Palmer and Cooper, 2012) with an adjustment to reflect the UK housing stock. However, the building physics data inputs are the direct results of the calculations performed in SAP and BREDEM. The model then reads in data for individual representative dwelling in order to perform building physics calculations. The CHM is one of the most transparent models because the model is built and all its calculations performed in Microsoft Excel. The output of CHM is one of the studies that made up the UK housing fact file in the Department of Energy and Climate Change (DECC) (Palmer and Cooper, 2012).

viii. The Domestic Dwelling Model (DDM) developed by Mhalas, Kassem, Crosbie, and Dawood (2013)

The DDM is a new approach which has very different characteristics to the previously described UK based steady-state model. The DDM models energy consumption and carbon emissions of dwellings and visualisation. The neighbourhood based on model is highly disaggregated as it estimates each dwelling independently within the neighbourhood. The model uses the SAP and BREDEM energy calculation engine. The DDM utilises information from aerial and terrestrial imagery, digital maps, household surveys, census, and ONS (Figure 2.11) (Mhalas et al., 2013).



Figure 2.11 Structure and form of the DDM (Mhalas et al., 2013)

This model is implemented on a GIS platform which makes it possible to replicate data entry process for several dwellings that have similar characteristics. This is particularly useful in undertaking the energy assessment for terraced, semi-detached and detached houses built during similar time periods (Mhalas et al., 2013). The DDM currently include scenarios on fabric change, PV addition, condensing boiler improvement, etc. but not future scenarios or energy saving measures. Further work is planned to identify the impact of improvements and the

development of a more accurate calculation of the rate of return of investment based on inflation indices.

2.4.2. Critical Analysis of the UK based Models

The models introduced all share the same calculation engine, BREDEM, modified to varying degrees based on the aims and needs of each model. They are capable of estimating baseline energy consumption of existing housing stock, predicting energy saving and carbon emission reductions from a variety of scenarios and except for EEP, they are all capable of predicting future energy demand and savings from proposed scenarios. Natarajan et al. (2011) has judged BREDEM as a well-established method which estimates UK dwelling energy consumption and predicts dwellings' energy consumption and carbon emissions at a highly disaggregated level based on building physics. However, one of the major drawbacks associated with BREDEM is the generalisation of occupant behaviour patterns, as in simplifying appliances consumption based on floor area and number of occupants (Cheng and Steemers, 2011).

All eight models are used as policy decision-making tools. However, the models are varied in terms of their output levels, extent of stock disaggregation, and the scenarios analysed, as shown in Table 2.2. These models have been developed since the early 1990s and each model was criticised by its successor and by more recent researchers due to a number of limitations. Firstly, all the models have been criticised for their low level of transparency. Kavgic et al. (2010) and Mhalas et al. (2013) discussed that the models' transparency, in terms of the architecture and data sources, is one of the main issues that needs to be addressed in future models. Cheng and Steemers (2011) state "No model is perfect. Models become useful if their assumptions and limitations are known to the users so that the users can make informed decisions on the practical application of the results. This transparency is generally lacking in the existing UK domestic energy models and as a result significantly limits the viability of the existing models in assisting policy formation".

Table 2.2 Comparative analysis of prominent UK based steady-state domestic energy and carbon models

Model	Year	Calculation	Output	Disaggregation	Analysed scenarios
BREHOMES	Early 1990s	BREDEM-12	Annual energy consumption	1000 dwelling types: based on age, tenure type, etc	Reference and efficiency until 2020
Johnston	2003	BREDEM-9	Annual energy consumption and carbon emission	Two dwelling types: pre and post-1996	Business-as-usual, demand, integrated until 2050
UKDCM	2006	BREDEM-8	Monthly energy consumption and carbon emission	20,000 dwelling types by 2050	Business-as-usual, 44% reduction, 25% emission reduction below 1990 levels by 2050
DECarb	2007	BREDEM-8	Monthly energy consumption	8064 combinations of dwellings from 6 age bands	Scenarios from: UKIPO2, BREHOMES, Johnston, UKDCM, Back-cast (1970– 96)
EEP	2007	SAP	Annual energy consumption and carbon emission	1300 dwelling based on age band and built form	Fabric change, PV addition, condensing boiler improvement
CDEM	2009	BREDEM-8	Monthly energy consumption and carbon emission	47 archetypes based on age band and built form	A scenario to predict 2001 housing stock
СНМ	2010	BREDEM-8 and SAP	Annual energy consumption and carbon emission	16,670 dwelling types	Conducted sensitivity analysis for the 15 most sensitive parameters in the model
DDM	2013	BREDEM-8 and SAP	Annual energy consumption and carbon emission	756 dwelling types	Fabric change, PV, I-CHP, Condensing boiler, ASHP (Under-floor and radiator)

Secondly, and more importantly, these models are not capable of taking into consideration the complex, interdependencies, and dynamic nature of the energy consumption and carbon emission (Oladokun and Odesola, 2015). This is because the modelling approaches of these models are based on steady-state calculations. These models therefore work with particular sets of data inputs trying to generate particular sets of outputs that have little or no room to accommodate uncertainty in input datasets. Kavgic et al. (2010) state that "the new generation of bottom-up building stock models should include multidisciplinary and dynamic approaches, so that for instance they can improve the synergy in policy development on energy efficiency, comfort, and health".

The critical analysis of UK-based domestic energy models suggests that there is the need to look for more detailed modelling approaches capable of dealing with the limitations discussed (Oladokun and Odesola, 2015). It is evident that the uniformity of the assumptions made by BREDEM will result in systematic errors that could have negative consequences for energy policy making, and the targeting of energy efficiency measures. A more recent study by Kane et al. (2015) discussed that much has changed in the 25 years since the BREDEM modelling framework was created, in the way we use our homes, the amount data made available to the research community, and the availability of more detailed energy models. They state: "We no longer expect to be able to understand the mathematics behind models or be able to complete calculations by hand; arguments that have been used in support of simple BREDEM-like modelling". Consequently, recent researches have modelled stocks using more detailed dynamic simulation programmes. Detailed analysis of such models, which are found to be of relevance to the model developed in this thesis, is presented in the following sub-sections.

2.4.3. Dynamic Energy Models

Only a few dynamic energy models of building stocks have been developed and the advantages or disadvantages of using such models have not been investigated in depth. This review identified four prominent dynamic energy models which were developed for Canadian, US, Scottish and English housing stock. Critical analysis of these models provides a great opportunity to identify the weakness and strength of current dynamic energy building stock models.

i. The Canadian Residential Energy End-use Model (CREEM)

developed by Farahbakhsh, Ugursal and Fung (1998)

The Canadian residential stock includes five major types of dwellings: single-detached, single-attached, apartments, high-rise apartments, and mobile homes. Single- detached and single-attached houses account for 68% of the households in Canada and are responsible for the largest share of residential energy consumption. Hence, the CREEM only considers single-detached and single-attached dwellings (Farahbakhsh et al., 1998). The CREEM used the HOT2000 simulation programme to calculate the delivered energy use from Canadian residential stock. The CREEM included 8767 dwellings which were divided into 16 archetypes based on built year (pre-1941, 1941- 1960, 1961-1977, 1978 and later) and regional location (Western Canada, Prairies, Central Canada, Atlantic Canada).

Actual energy billing data from fuel suppliers and utility companies for a complete year were available for 2524 of the 8767 houses. Hence the data on these dwellings were compared to simulation results in order to verify the model. The refinements identified from the verification process were applied to the rest of the 8767 house files as necessary to improve the accuracy of the simulation results (Farahbakhsh et al., 1998). The CREEM was then used to assess the reductions in energy consumption and carbon dioxide emissions from the Canadian residential sector as a result of various energy efficiency measures.

The work carried out in developing the CREEM was evolved over time and with addition of new datasets on Canadian housing stock to develop a new hybrid model, namely: the Canadian Hybrid Residential End-use Energy and Emissions Model (CHREM). The CHREM relied on the 17,000 house details of a representative database called the Canadian Single-Detached and Double/Row Housing Database (CSDDRD) (Swan et al. 2009). This database accounted for the wide range of climate, construction types, and energy sources found throughout Canada's regions which could be used to develop dynamic energy models. The CSDDRD included detailed information on each dwelling's location, geometry and orientation, thermal zone presence, construction materials including windows and doors, air-tightness, and HVAC and DHW components.

The CHREM employed two energy modelling techniques: statistical and engineering (Figure 2.12).



Figure 2.12 Structure and form of the CHREM (Swan et al. 2009)

These techniques were used to estimate the energy consumption of the two main end-use groups: domestic appliances and lighting (AL), domestic hot water (DHW), and space heating and cooling. The CHREM employs neural network (NN) technique as the statistical half of the model for use in estimating the annual energy consumption for AL and DHW loads; and uses the ESP-r dynamic simulation engine to calculate heating and cooling loads (Figure 2.12).

The CHREM assumed only one thermal zone for the main part of the dwellings and justified this zoning strategy by highlighting lack of data on thermal zones. The individual storeys of the main zone were not identified, but instead combined into one thermal zone with a modification to building height to account for each storey's floor area (Swan et al. 2009).

The CHREM also made a few simplifications to the geometry of the modelled dwellings. All houses were modelled as a rectangular block using a width to depth ratio (Swan et al. 2009). The authors identified that this method only partially accounted for the perimeter to area relationship that affects energy consumption due to exposed surface area and no sensitivity analysis was performed to investigate impact of such simplification on thr model estimates.

Another assumption made in the CHREM was regarding the conditions on the outside face of each zone surface. In case of Double/Row houses, adiabatic conditions are specified on one or both side walls with regard to opposing dwelling location and the thicknesses are halved to account for thermal mass attribution (Swan et al. 2009). Adiabatic party walls specified in ESP-r dynamic simulation programme, will assume that the neighbouring dwellings have exactly same thermal conditions and will neglect the heat transfer between the two dwellings.

Finally, the CHREM used an XML reporting technique to store simulation results in annual form to evaluate energy consumption and contributions due to a variety of housing components.

ii. The Huang and Brodick Model developed by Huang and Brodick (2000)

This model was developed for the US building stock (both residential and commercial) and is based on the aggregated cooling and heating loads attributable to different building envelope components in the stock, such as windows, roofs, walls, and space heating and cooling systems. The modelled building stock comprised of 112 single-family, 66 multi-family housing and 481 commercial buildings. The dataset used included information on age (pre-1940s, 1950-1959, etc.), dwelling type (single- family, small multi-family) (Huang and Brodick, 2000). The overall energy use of the US housing stock was calculated using the DOE-2.1E simulation tool.

DOE-2 is a widely used dynamic energy simulation tool that can estimate the energy consumption and cost for all types of buildings. DOE-2 uses a description of the building layout, constructions, operating schedules, and air conditioning systems, along with weather data, to perform an hourly simulation of the building and to estimate utility bills. DOE-2.1E is the "legacy" version of DOE-2 and provides for more detailed modelling of the thermal and optical properties of windows (DOE-2, 2016).

The authors of this model acknowledge that the totals for the non-space conditioning end-use such as water heating and lighting were modelled very simply. Only gas was included as the primary fuel source for space and water heating, even though electricity and other fuels are also used as a primary energy source (Huang and Brodick, 2000). The model provides information on potential improvements in certain building components, such as improving windows from single to double- glazing, but doesn't specify in which parts of the stock these gains would occur or would benefit most from the change. The inadequate description of model and lack of evidence on decisions made in development of model are the main weakness of the Huang and Brodick Model.

The Energy Systems Research Unit (ESRU) Domestic Energy Model (EDEM) developed by Clarke et al. (2004); Clarke, Johnstone, Kim and Tuohy (2009)

The EDEM is a web based tool developed at the University of Strathclyde for Scotland housing stock. The EDEM is able to estimate energy consumption and carbon emissions both at individual and national scale. The EDEM was used to rate the energy and carbon performance of individual dwellings as required by the EU Directive on the Energy Performance of Buildings (European Union, 2002). The model used the 2002 Scottish House Condition Survey (Scottish Homes, 2003) as main source of data. The model categorise the dwellings in terms of Thermodynamic Classes (TC) so that different Architecture Classes (AC) may belong to the same TC (Clarke et al., 2009).

The EDEM employed the ESP-r dynamic energy simulation programme to determine dwelling performance by subjecting the dwelling models to long-term weather sequences. Clarke et al. (2004) justified use of dynamic energy simulation over BREDEM based steady-state models by stating "*Simplified methods cannot adequately represent the performance of the myriad upgrade options that may be applied individually or in combination. Also, as buildings have extended lifetimes, it is important to assess performance under likely future contexts*".

The EDEM results were verified using detailed models of 5 real houses. The house models were subjected to energy efficiency improvements and simulations were re-run. The predicted heating energy demands resulting from the detailed simulations were then compared to the value associated with the matched TC model. The results indicated discrepancies ranging from 3% to -13%, indicating that the TC approach is a reasonable proxy for the real situation (Clarke et al., 2009).

The EDEM was also used to investigate 6 case studies: national stock upgrade, regional housing upgrade, dwelling energy labelling, impact of grid electricity generation mix, financial appraisal of upgrade options, and financial appraisal of individual dwelling upgrade. iv. The He et al. Model developed by He, Lee, Taylor, Firth and Lomas (2014); He, Brownleeb, Lee, Wright and Taylor (2015)

The He et al. model (2014) is a dynamic energy model which uses English Housing Survey (EHS) database as main source of input data and employs EnergyPlus as simulation engine. EnergyPlus takes an input data file (IDF), in which a building model is specified, and a weather file to run a dynamic simulation of a building. The model simulates the housing stock in the North East region of England to examine the possible CO₂ reductions corresponding to different scenarios.

The model used 2008-9 EHS database which contains 935 sample dwellings in the North East England. These sample dwellings are representative of about 1.2 million homes in that region. The 935 sample dwellings were distributed among 6 dwelling types, 10 age bands, 8 types of wall construction, and 12 loft insulations. Only 759 houses, including 90 detached houses, 329 semi-detached housed, 221 mid-terrace houses and 119 end-terrace houses were considered in the model. All the dwellings were assumed to have East/West orientation (He et al., 2014).

The dwellings were modelled with two separate zones: the living area and the rest of the dwelling. The authors justify using such zoning configuration by referencing the study performed by Taylor et. al. (2013) where it was found that the two zone models separating living area from the rest of house can predict annual energy demand within about 10% of the best estimate using individual room zones. The results of the model were verified through inter-model comparison with CHM results (see section 2.4 [vii] for CHM description). Since both models take inputs from the EHS database and simulate each dwelling individually, the results from both models were comparable (He et al., 2014). The dynamic model developed by He et. al. (2015) predicted lower demand values compared to the steady-state CHM result which were consistent with findings of other studies (Shorrock et. al., 1996; Yilmaz et. al., 2014).

2.5. Model Validation

Although building simulation has been widely used during the past three decades to investigate the effect of retrofit measures on energy savings and comfort, without validation of the base case model, results produced are not reliable (Westphal & Lamberts, 2005). A large number of studies have shown discrepancies (which were often significant) between the model predictions and measured building energy use (Coakley, Raftery & Keane, 2014). The purpose for validation is to ensure that the model could reasonably represent the thermal and energy behaviour of the real building and thus achieve confidence in model predictions (Westphal & Lamberts, 2005).

Judkoff (1988) identifies a number of ways to validate the whole-building energy models: comparative testing (Inter-model comparison), analytic verification and Empirical validation. The three validation techniques are shown schematically in Figure 2.13 Analytic, Comparative and Empirical techniques (re-created from (Judkoff et al., 2008))Figure 2.13. Table 2.3 compares these techniques by highlighting their advantages and disadvantages.



Figure 2.13 Analytic, Comparative and Empirical techniques (re-created from (Judkoff et al., 2008))

Validation Method	Advantages	Disadvantages	
Comparative (Inter- model comparison)	 No input uncertainty Any level of complexity Many diagnostic comparisons possible Inexpensive and quick 	• No absolute truth standard (only statistically based acceptance ranges are possible)	
Analytical	 No input uncertainty Exact mathematical truth standard for the given model Inexpensive 	 No test of model validity Limited to highly constrained cases for which analytical solutions can be derived 	
Empirical	 Approximate truth standard within experimental accuracy Any level of complexity 	 Experimental uncertainties: Instrument calibration, spatial/ temporal discretization Imperfect knowledge/ specification of experimental object (building) being simulated High quality detailed measurements are expensive and time consuming Only a limited number of test conditions are practical 	

Table 2.3 Comparison of validation techniques (Judkoff 1988)

Comparative testing is an inexpensive and quick technique which involves no input uncertainty. A comparative test directly compares results of two or more building energy simulation tools which have used similar inputs. In this type of validation, a piece of code can be compared to itself by changing a specific parameter and determine sensitivity of simulations to that parameter. One main advantage of comparative testing is that it doesn't require any data from a real building. Comparative testing enables the investigator to control accuracy of input data and eliminate any external error. Furthermore, input parameters can be modified to test the sensitivity of simulations to change in input data. In Analytical verification simulation results are compared with the results of a solved analytical solution (ASHRAE, 2009). The same as comparative testing, analytical verification is inexpensive and has no input uncertainty. In analytical verification the investigator can test some specific heat transfer mechanisms in the building. In this technique output of modelling tools, which is based on numerical solution used in the code of programme, is compared to a unique analytical solution of heat transfer problem. An example of work using analytical verification (and comparative testing) is Building Energy Simulation Test, BESTEST (Judkoff & Neymark 1995). BESTEST is a method for systematically testing building energy simulation programmes, diagnosing sources of disagreement, and validating the capabilities of building energy simulation programs.

In empirical validation results from building simulation are compared with the data measured in real buildings (ASHRAE, 2009). In other words, empirical validation is the comparison of the estimates of the model with physical measurements (Bowman & Lomas 1985). There are various published literature on validation mainly for residential buildings rather than large commercial buildings; where conducting detailed measurements require considerable efforts and costs (ASHRAE, 2009). A number of empirical validation studies are summarized by Neymark and Judkoff (2002).

One of the main challenges researchers have been faced with to calibrate building energy models using empirical validation is the lack of detailed empirical data particularly for residential buildings which is necessary to understand the operational complexities and develop better models (Buswell et al., 2013). In majority of the cases, even when the measured data is available, it has not been measured by end use and for example the gas use measured include the use for space heating, hot water and cooking which makes the calibration difficult. In addition, the measured data has also an uncertainty and the differences observed between the models and measurements will be due to errors in either set of data (ASHRAE, 2009).

2.6. Data Translators for Dynamic Simulation

The four dynamic energy models described in previous section, all use some sort of translator to develop Input Data Files (IDFs) which are suitable for the employed calculation engines. The CREEM (Farahbakhsh et al., 1998), the Huang and Brodick (2000) Model, and the EDEM (Clarke et al., 2004) didn't provide any documented detail on technical aspects of the translators they developed. He et. al. (2014; 2015), on the other hand, identified that available tools to create IDFs were not suitable to simulate a relatively large number of real houses with different characteristics. Consequently, He et al. (2015) developed an in-house programme called the Building Generation Tool (BGT) to automatically create the IDFs of the modelled dwellings. The BGT implemented all the assumptions which were required to developed dynamic energy models based on EHS data.

2.5.1. The gbXML Translator

A study by Dimitriou et al. (2016) was identified where a translator was developed to convert data from gbXML format into EnergyPlus IDFs. This translator was developed as part of the Design4Energy retrofit scenario which uses Building Information Modelling (BIM) of existing domestic buildings to assess their energy performance using a Building Energy Modelling (BEM) technique (Dimitriou et al., 2016). The XML based gbXML format enables easy incorporation of additional information that might be required for energy analysis. The conversion process extracts as much information as possible the gbXML file and introduce additional parameters to create the IDF files that can be used by EnergyPlus to perform the analysis (Figure 2.14).

As seen in Figure 2.14, the selected BIM software was Autocad's REVIT. The process begins by developing REVIT models based on the data collected from building surveys. Then the gbXML files were used to fill in for lacking data and create EnergyPlus IDFs. The simulations were run and results were exported in csv file formats for further analysis and to proceed

with the model calibration. Multiple iterations of the calibration process was required to achieve good agreement between the modelled and the actual building energy performance (Dimitriou et al., 2016).



Figure 2.14 Structure and form of the BIM to BEM translator (Dimitriou et al., 2016)

The translator developed by Dimitriou et al. (2016) is the best documented one identified within UK building energy modelling context. The translator converts gbXML files into IDFs in two steps (Figure 2.15). The first step in the conversion process is the conversion of the gbXML file to a XML file that contains all the information required for EnergyPlus simulation. All the nodes required by EnergyPlus were included in a 'idfXML' file and default values were stored prior to converting gbXML files. Then the idfXML file was populated with the available information from the gbXML file. Having developed the representative idfXML files, the EnergyPlus IDFs were created based on these files in the second stage of the translation process (Dimitriou et al., 2016).



Figure 2.15 The translation process to convert gbXML files into EnergyPlus IDFs (Dimitriou et al., 2016)

The translation process described above can handle multiple files at the same time. The Translator was tested on a single house located in Loughborough, UK. The most significant assumption made in developing the BIM to BEM translator was that the heating system can be represented by an ideal loads system where calibration of the model to the measured data was not performed. Despite such assumption, the model presented relatively good agreement to the measured data. Dimitriou et al. (2016) concluded "*The deviations observed between the modelled and measured temperatures highlight the importance of transparent data exchange and default setting when forming the energy model*". Further work to validate the translator using other BIM tools was found necessary before using the translator for retrofit decision-making purposes.

2.5.2. The HPXML Translator

An ongoing work was also identified which is being carried out by National Renewable Energy Laboratory (NREL) and Building Performance Institute (BPI) in the USA. The first part of this initiative works on developing a new data storage national standard, called Home Performance XML (HPXML), to support the needs of home performance programs, quality assurance agents, and financial institutions. HPXML can also aggregate data across programs and remove redundancies in incentive compliance to support these additional consumers (Andrulis and Thomas, 2012). The second part of this work develops translators to convert HPXML data into different input data files which could be accessed and executed with various building energy modelling tools (Neymark and Roberts, 2013). Figure 2.16Figure 2.16 shows the overall structure of the work being carried out to use HPXML data for developing dynamic energy models of the US housing stock.



Figure 2.16 The translation process to convert HPXML files into various IDF formats (Polly et al., 2012)

The capacity of HPXML to store utility billing and building description details also provided an opportunity for developing standardised accuracy test for residential energy analysis tools (Neymark and Roberts, 2013). The authors identified the development of HPXML can enable software developers to create translators suitable for their input data scheme for efficient access to data. Such translators can facilitate the possibility of modelling whole building stock using dynamic energy simulation software and implement the standardised software accuracy test to translated data.

The work on developing translators for different software was ongoing at the time of writing this thesis. The work is being carried out within a working group which includes modelling expert from different organisations and software developers. The work presented in this thesis is the only contributor to this working group which is based outside US and provides an international insight into possibility of developing a worldwide building performance data repository which could be used for simulation purposes.

2.7. Summary

This chapter presented a review of the literature and completes objective 1 (*Identify and review literature on the modelling approaches that could be used to forecast energy use and* CO_2 *emissions in the UK dwellings, and perform a critical analysis of the work that has been undertaken to utilize available reduced data for energy modelling purposes*). The chapter presented a critical analysis of UK based domestic energy and emission models. The main application of all models is common as they are all used as policy decision making tools. However, the models are varied in terms of their output levels, extent of stock disaggregation, and scenarios analysed. These models have been developed since early 1990s and each model was criticized by its successor and by more recent researchers due to a number of limitations:

 All the models have been criticised for their low level of transparency. The model transparency, in terms of the architecture and data sources, is one of the main issues that need to be addressed in future models. The RdDEM model described in this thesis is an effort to remove all the ambiguities observed in the previous modelling practices by clearly describing the employed dataset, modelling methodologies including zoning and geometrical details, and model outputs.

- These models are not capable of taking into consideration the complex, interdependencies, and dynamic nature of the energy consumption and carbon emission. The RdDEM model described in this thesis employs a dynamic energy simulation engine, instead of steady-state BREDEM engine used by previous modelling exercises, enabling the model to capture transient and dynamic behaviours in the dwellings.
- The uniformity of the assumptions made by the models result in systematic errors that could have negative consequences for energy policy making, and the targeting of energy efficiency measures. The RdDEM model described in this thesis introduces evidence-based decision-making procedures to handle zoning and geometry aspects of the dwellings. These procedures reduced the uniformity of assumptions considerably compared to previous modelling exercises.

Dynamic energy models developed for Canadian, US, Scottish and English housing stock were also described providing details on positive and negative attributes of each model:

- The Canadian model only considered single-detached and singleattached dwellings, and employed two energy modelling techniques: statistical and engineering. The model assumed only one thermal zone for main building parts, and the storeys of the main zone were combined into one thermal zone. The party wall was modelled as adiabatic wall neglecting the heat transfer between the two dwellings. The RdDEM model described in this thesis improves the Canadian model by introducing more detailed zoning configuration and also through modelling a non-adiabatic party wall which is capable of representing heat transfer effects through the wall.
- The US model was based on the aggregated cooling and heating loads with very simple water heating and lighting details. Only gas was included as the primary fuel source for space and water heating, even though electricity and other fuels are also used as a primary energy source.

- The Scottish model categorised the dwellings in terms of Thermodynamic Classes (TC) and Architecture Classes (AC). The model results were verified using detailed models of 5 real houses. A similar verification process was applied to the RdDEM model developed in this study.
- The English model used EnergyPlus and custom weather files from North East region of the UK. All the dwellings were assumed to have East/West orientation. The dwellings were modelled with SAP zoning. The RdDEM model described in this thesis also uses EnergyPlus as calculation engine but applies a floor zoning configuration to the dwellings model instead of SAP zoning.

The chapter concluded with identifying and describing the available data translators developed for the purpose of domestic dynamic energy modelling. The focus was on the translators which convert XML based data into various input data files for dynamic energy simulation software: gbXML and HPXML translators. Based on this analysis of the literature, and to address the shortcomings of other models, a new model (the RdDEM) is proposed as outlined in the next chapter.

3. OVERVIEW OF METHODS

3.1. Introduction

This chapter presents an overview of the methods used to develop the Reduced Data Dynamic Energy Model (RdDEM) of the UK dwellings, based on the findings from the literature review (Chapter 2) and in pursuit of Objectives 2 (*Identify a suitable dataset of UK dwellings to be used as the source of modelling data*), 3 (*Develop and test a data preparation process that will enhance the reduced data in order to produce an equivalent set of detailed data that is suitable for dynamic energy simulation*) and 4 (*Develop and run a reduced data dynamic energy model of UK dwellings that translates the prepared data into a form suitable for dynamic simulation using established models*). The new modelling framework that was developed is shown in Figure 3.1. It takes existing data from Energy Performance Certificate datasets and converts them into files suitable for use in EnergyPlus dynamic energy simulation software. EnergyPlus was chosen because of its technical capability, ease of use (text input file), cost (free), and validated code (US Department of Energy, 2013).

Data preparation process and translation are described below, with extended details in Chapter 4 (Zoning and enhanced geometry), Chapter 5 (Equivalent construction materials and boundary conditions), and Chapter 6 (Translation, simulation and results).

The reduced data used to test this approach was from the DEFACTO (Digital Energy Feedback and Control Technology Optimisation) project (Mallaband et al., 2014) which is being carried out by Loughborough University from 2012 to 2018. The dataset included 165 semi-detached dwellings located in the Midlands region of the UK. The format of the reduced data was individual XML files for each dwelling, as is described in detail in section 3.2. This modelling dataset was run through a data preparation process in which modifications and enhancements were made to increase the level of detail to meet the requirements of dynamic energy simulation.


Figure 3.1 Overall structure and the modelling framework

The prepared data was called the 'Equivalent Data' as the nature of reduced data was kept intact, and only controlled modifications were made to facilitate dynamic energy simulation. The data preparation process was carried out in three steps:

- i. Defining zoning and enhanced geometry,
- ii. Defining equivalent construction and thermal mass, and
- iii. Defining equivalent boundary conditions.

These three steps of developing equivalent data are described in section 3.3. The equivalent data was then run through a translation process which converted the equivalent data into the format required for the dynamic energy simulation. The translation process created individual IDFs for each dwelling in the dataset. The IDFs were then fed into the EnergyPlus engine and simulations were run. Description of the translation process and IDFs is presented in section 3.4. The translation process together with three steps of the data preparation process (as shown in Figure 3.1) formed the Reduced Data Dynamic Energy Model (RdDEM).

3.2. Modelling Dataset

The DEFACTO project investigates how the use of digital control and feedback technologies to enable reduction and management of energy use (Mallaband et al., 2014). The DEFACTO dataset describes semi-detached dwellings which are owner occupied and are located in the Midlands region of the UK. The reduced data were collected by professional home energy assessors for the purpose of creating Energy Performance Certificates (EPCs). EPCs are calculated using steady-state BREDEM (Dickson et al., 1996) based calculations as used in the Government's Reduced Data Standard Assessment Procedure (RdSAP). These reduced data are stored in individual XML files for each dwelling, as in the national EPC database (Watson, 2009).

The reduced data was stored under 'SAP Data' heading of the XML files (see APPENDIX C for example EPC XML File). The data described each dwelling with just enough details to enable RdSAP calculations (for more details see Section 2.2.2). The data had three main categories: geometrical dimensions of the dwellings, construction details and, heating and hot water systems.

The geometrical details available in the modelling dataset included: floor area, floor height, heat loss perimeter and party wall length for each floor of the dwellings and, roof area. The construction details included: construction age band, wall, roof and floor construction types, wall thickness, multiple and single glazing proportion and, number of extensions and conservatories. Where extensions existed, all construction and geometrical details were provided separately from the main dwelling. Details of heating and hot water systems, boiler type, heat emitter type, control type and fuel type were also included in the dataset.

In this research, 83 of the 165 dwellings described in the DEFACTO dataset were chosen to test the RdDEM approach. It was assumed that if the method was found to be appropriate for these dwellings, further work could be carried out to expand it to other dwellings. All the DEFACTO dwellings were semidetached and located in the same geographical region but had very diverse characteristics, being constructed from before 1900 to 2012. This resulted in dwellings having very different layouts, construction types, materials, insulation levels and systems. Since such diversity of characteristics added further complexity to the modelling process, a subset of dwellings in the dataset was selected for this study. This subset will be referred to as 'Modelling Dataset' for the rest of this thesis. The modelling dataset was selected such that while reducing complications and obstacles in the dynamic simulation process, it would still be representative of common UK semidetached dwellings.

The modelling dataset included most common age band and construction types in the dataset. In the DEFACTO dataset, 155 of the dwellings had one of 'B', 'C', 'D' or 'E' age bands and only 10 dwellings had other age bands ('A', 'F', 'J' or 'L'). These 10 dwellings were excluded from the modelling dataset.

The modelling dataset also excluded a number of dwellings based on their external wall construction type. 148 of the dwellings had either cavity or solid brick external walls. Only 7 dwellings had other external wall construction types (Timber frame, Sandstone or system built). These 7 dwellings were excluded from the modelling dataset and only dwellings with cavity and solid brick external walls were included. This round of shortlisting left the modelling dataset with 148 dwellings.

The majority of the dwellings in DEFACTO dataset had a pitched roof and the remaining had a 'room in roof'. The dwellings with room in roofs were excluded from the modelling dataset which left 134 dwellings.

In the DEFACTO dataset, 92 dwellings had solid floor, 4 of which were insulated. Of remaining 42 dwellings, 41 had non-insulated suspended floor and only one had insulated suspended ground floor. The 5 dwellings with insulated ground floors were excluded leaving 129 dwellings in the modelling dataset.

Most of the remaining dwellings (122) had 100% double glazing, 4 had more than 90% double glazing, 2 had more than 70% double glazing and only one dwelling had 10% double glazing proportion. The dwellings with less than

100% double glazing proportion were excluded from the modelling dataset to allow a single fenestration type in all models.

The majority of houses in the dataset were two-storey and there were only 3, three-storey dwellings. The three-storey dwellings were excluded from the modelling dataset. The DEFACTO dataset also specified the number of extensions and conservatories for each dwelling. From the remaining 119 two storey dwellings in the dataset, 51 had no extensions, 54 had only one extension, 10 had two extensions, 3 had three extensions and one had 4 extensions. The dwellings with more than one extension were excluded from the modelling dataset. Dwellings with conservatories were also excluded from the modelling dataset to leave 85 dwellings.

All of these dwellings had a gas central heating boiler and radiators as the main heating and hot water system. The main heating control for dwellings in the modelling dataset was a boiler programmer with room thermostat. Only two dwellings had 'TRVs and bypass' as the main heating control and these were removed to leave the modelling dataset with 83 dwellings.

Having completed the shortlisting process, the modelling dataset included a simpler set of houses with which to test the RdDEM process. The modelling dataset was representative of many UK semi-detached dwellings with common age bands, construction types and systems. In the modelling dataset (Figure 3.2), 22 (27%) of dwellings had age band 'B', 20 (24%) had age band 'C', 17 (20%) had age band 'D' and 24 (29%) had age band 'E'.

As Identified in the shortlisting process, only the dwellings with cavity and solid external wall constructions were included in the modelling dataset: 9 (11%) dwellings had cavity walls with no insulation, 50 (60%) had cavity walls with filled cavities and 24 (29) dwellings had solid walls with no insulation. Figure 3.3 shows the distribution of each external wall construction type in the modelling dataset.



Figure 3.2 Dwellings distribution in the modelling dataset based on construction age band and the corresponding construction years



Figure 3.3 Dwellings distribution in the modelling dataset based on external wall construction

In the case of roof construction, only the dwellings with pitched roof were included in the modelling dataset: 62 dwellings had pitched roofs with known insulation, 16 had pitched roof with no insulation, 2 had pitched roofs with insulation at rafters and 3 dwellings had pitched roofs with unknown insulation level i.e. the assessor couldn't measure the loft insulation thickness. The dwellings with known level of roof insulation, had loft insulation thickness ranging from 50 mm to 300 mm (Figure 3.4): 2 (3%) had 50 mm loft insulation, 4 (6%) had 75 mm, 15 (25%) had 100 mm, 8 (13%) had 150 mm, 16 (26%) had 200 mm, 10 (16%) had 250 mm, 4 (6%) had 270 mm and 3 (5%) dwellings had 300 mm.



Figure 3.4 Dwellings distribution in the modelling dataset based on loft insulation thickness

Dwellings in the modelling dataset had two main ground floor constructions: 57 (69%) had solid ground floor and 26 (31%) had suspended ground floor. The dwellings in the modelling dataset had total floor area ranging from 62 m² to 191 m² (Figure 3.5): 54 (65%) dwellings had total floor area of 50-100 m², 21 (25%) had total floor area of 100-150 m² and 8 (10%) had total floor area of 150-200 m².





3.3. Data Preparation Process

This section describes how the modelling dataset was modified and improved in order to prepare it for dynamic simulation as part of the RdDEM. The main areas of improvements in the modelling dataset were identified and each improvement was identified and tested:

- Zoning,
- Geometry,
- Construction materials and thermal mass,
- Internal boundary conditions,
- External boundary conditions (weather data and orientation).

These areas had a lack of required data to run dynamic simulations because they were not required for the SAP model, or were assumed to have a fixed value in SAP. The enhancements to zoning and geometry required a coherent and detailed investigation of options which led to evolutions of the model. Whereas, the enhancements for internal boundary conditions, internal thermal mass, construction materials and weather data were simply based on SAP guidelines in order to achieve a set of equivalent input data for the dynamic energy simulation.

3.3.1. Zoning

In designing for heating ventilation and air conditioning systems (HVAC), a 'zone' is an area of a building in which temperature is controlled by one thermostat. The number of zones is based on the thermal demand of various building spaces. In dwellings, rooms with different heating set-point temperatures could be individual thermal zones. When modelling dwellings, the thermal requirements of different spaces should be considered, and thermal zones should be assigned accordingly.

SAP allows two zones to be implemented in the model: living area and the rest of the house. EnergyPlus has the capacity to implement as many thermal zones as required. Previous studies developing dynamic energy simulation of dwellings have justified their zoning strategy by highlighting the limitations of the dataset that was used.

In the Canadian Residential Energy End-use Model (CREEM) developed by Farahbakhsh, Ugursal and Fung (1998), which used dynamic energy simulation to investigate the impact of various carbon reduction strategies (see Section 2.4.3); all the habitable rooms in dwellings were assigned to a single thermal zone. The He et al. Model developed by He, Lee, Taylor, Firth and Lomas (2014), on the other hand, followed the SAP approach and considered two zones, separating the living area from the rest of the house (see Section 2.4.3). These authors justified using their preferred zoning strategy by highlighting the lack of geometry information in the original dataset but without presenting any detail on the impact of such a zoning strategy on the predictions of the model. To determine the optimum number of zones to be modelled, three possible zoning strategies, with varying number and configuration of zones, were identified and a zoning study was carried out to investigate each strategy.

In the zoning study, a single semi-detached dwelling was modelled three times with exactly the same physical and thermal characteristics but varying the zoning strategy in each model:

- i. Single zone strategy: where a single zone was assigned to the whole dwelling,
- ii. Floor zoning: where two zones were considered ground floor and first floor, and
- iii. SAP zoning: where two zones were considered living area and the rest of the dwelling.

The predictions of each model were compared to a reference model, where every habitable room in the dwelling was modelled as an individual thermal zone, giving a total number of nine zones to the model. The 'Reference Model' is described in detail in Section 4.2. The energy demand and internal temperature predictions of each model were compared to the predictions of the reference model and the most suitable zoning strategy was identified and then used in the RdDEM. The detailed description of zoning strategy models, their results and outcomes of the study is provided in Section 4.3.

3.3.2. Geometry

Building geometry is an important aspect of energy models: heat loss through fabric and ventilation is highly dependent on building geometry information. The energy balance between heat loss from the building and the heat gains into the building identifies the amount of required space heating energy for maintaining desired thermal conditions. As a result, a very important step in modelling energy performance of buildings is to understand geometry details and to implement those details in the model. The modelled geometry should be capable of representing the actual heat loss through building fabric.

The (reduced data) modelling dataset provided a limited amount of information on building geometry:

- floor area (m²),
- floor height (m),
- exposed (heat loss) perimeter (m), and
- length of the party wall (m)

Floor area was measured from inner surface of external walls, floor height was measured from the floor surface to the ceiling surface, exposed perimeter was the total length of the external wall dividing the dwelling floor from the external environment or from an unheated adjoining space and, party wall length identified length of the wall separating two adjacent semidetached dwellings.

These details were provided for each floor of the buildings in the modelling dataset and were intended for use by SAP to calculate the heat loss from the building fabric. However, a dynamic energy model of the dwellings requires full three-dimensional geometry. Hence, a methodology was needed to develop this geometry while retaining the heat loss areas using the limited details in the modelling dataset.

For the RdDEM, the available data was used to create a 3-dimensional rectangular geometry which maintained the limited details given in the modelling dataset. The methodology that was developed used the floor areas, heights, exposed perimeters and party wall lengths of each dwelling to create three-dimensional geometry for a rectangular building. The methodology was further developed to include extensions. The resultant geometry was still rectangular one, with correct party wall length, extension and main building exposed perimeter and correct floor area.

The RdDEM methodology was tested on a reduced data version of the 'Reference Model' and predictions were compared to the model which used all of the information in the 'Reference Model'. Ultimately, it was possible to successfully recreate representative three-dimensional geometry from the reduced data. All details and results are presented in Section 4.4.

3.3.3. Equivalent Construction Materials

The modelling dataset only included the construction type of the external wall, roof and ground floor. There was no further information on construction materials or corresponding U-values. Hence, an equivalent set of construction materials were developed for the RdDEM.

The equivalent construction materials for external walls were created such that the overall U-value of the walls matched the given values in Table S6 of SAP (SAP, 2012). Table S6 provides the external wall U-values based on wall type and dwellings' age band for houses located in England and Wales. The dwellings in the modelling dataset had three external wall types and belong to four age bands. Table 3.1 shows the relevant part of Table S6 for the dwellings in modelling dataset.

Wall type	Age band					
wan type	В	С	D	E		
Solid brick as built	2.1	2.1	2.1	1.7		
Cavity as built	1.6	1.6	1.6	1.6		
Filled cavity	0.5	0.5	0.5	0.5		

Table 3.1 Corresponding U-values (W/m²K) of the wall types specified in the dataset based on dwellings' age band (re-created from Table S6 SAP 2012 (SAP, 2012))

As seen in Table 3.1 there are four different U-values for the external walls of the dwellings in the modelling dataset. Hence, four sets of construction materials were required:

- Solid brick wall with U-value of 2.1 W/m²K for dwellings with age bands 'B', 'C' and 'D',
- Solid brick wall with U-value of 1.7 W/m²K for dwellings with age band 'E',
- iii. Cavity wall with U-value of 1.6 W/m²K for dwellings with age bands 'B',
 'C', 'D' and 'E'; and
- Filled cavity wall with U-value of 0.5 W/m²K for dwellings with age bands 'B', 'C', 'D' and 'E'.

All of the dwellings in the modelling dataset had pitched roof with no or some insulation. A single construction type was used for all dwellings but with varying insulation levels. For the dwellings with known insulation thickness, the roofs were modelled such that they achieved the suggested U-values in Table S9 SAP 2012 (SAP, 2012): 2.3 W/m²K for roofs with no roof insulation and U-values between 0.68 W/m²K and 0.14 W/m²K for roof insulation thickness of 50 mm to 300 mm. For the dwellings with unknown roof insulation, Table S10 SAP 2012 (SAP, 2012) was used to provide the roof insulation thickness based on the dwelling's age band.

All of the ground floors were modelled as solid ground floor such that the correct U-values were retained. To do so, the guidelines in Section S5.5 SAP 2012 (SAP, 2012) were followed to calculate U-values of suspended ground floors based on ground floor area and exposed perimeter. The corresponding U-value was then used to calculate the thickness of necessary insulation for an equivalent solid ground floor. Knowing the insulation thickness, an individual solid ground floor reflecting the U-value of each suspended ground floor in the dataset was modelled.

All dwellings in the modelling dataset had 100% double glazed windows. Hence, based on Table 6E SAP 2012 (SAP, 2012) a double glazed, air filled window with 6 mm gap and U-value of 3.1 W/m²K was modelled for all the dwellings. RdDEM modelled all of the semi-detached houses as stand-alone buildings, without the adjoining building. Hence, the models would look like a rectangular block with three external walls and one party wall (Figure 3.6). Therefore, a party wall construction was developed in order to capture the heat loss from this element.



Figure 3.6 Party wall in semi-detached dwellings as modelled in DesignBuilder

In case of cavity wall construction, the party wall can provide a mechanism for heat loss via air movement within the cavity between lower floors and the loft space and between the cavity and outside. Hence, SAP 2012 suggests a U-value of 0.5 W/m²K to account for this party wall bypass. In the case of solid wall construction, it is assumed that no heat transfer occurs across the party wall.

Where dwellings in the modelling dataset were of cavity construction, a single solid party wall with a U-value of 0.5 W/m²K was modelled by the RdDEM. In the case of dwellings with solid wall construction, the party wall was set to be

adiabatic and a component block (shown in grey in Figure 3.6) was added to shade it from incident solar radiation.

In creating the equivalent construction materials for the RdDEM, thermal bridging was included following the guidelines in SAP 2012 by increasing the U-value of individual building elements by 0.15W/m²K.

Detailed descriptions of the equivalent construction materials used in the RdDEM, and the related calculations, are provided in Section 5.2.

3.3.4. Thermal Mass

Thermal mass is the ability of building element to store and release heat. SAP defines a Thermal Mass Parameter as the sum of (area times heat capacity) over all construction elements divided by the total floor area (SAP, 2012). So, to calculate the thermal mass of dwellings it is necessary to know the area of each of the building element (walls, roofs, floors, etc) and their corresponding thermal heat capacity. In RdSAP, the overall thermal mass parameter of all existing dwellings is assumed to be 250 kJ/m²K. This same convention was used in the RdDEM. A process to derive the thermal mass of the external elements of the building, from their equivalent constructions, was developed. This thermal mass value was then deducted from the overall value of 250 kJ/m²K to find the additional thermal mass required to represent the internal walls. This is described fully in Section 5.2.5.

3.3.5. Internal Boundary Conditions

The internal boundary conditions in the RdDEM were designed to exactly match those in SAP to enable direct inter-model comparison. Internal heat gains from occupants, appliances, lighting and cooking were the same as defined in SAP 2012 Table 5 (SAP, 2012) for the typical gains. Where required, these gains were calculated based on the number of occupants

using the SAP guidelines to calculate number of occupants from the dwellings' total floor area (see APPENDIX D for details).

For the heating system, the heating periods and set-point temperatures, for living room and rest of the dwellings, were as described in SAP 2012 Table 9 (SAP, 2012). Detailed descriptions of the calculations used to achieve internal gains, heating periods and heating temperatures are presented in APPENDIX D.

3.3.6. External Boundary Conditions: Weather Data and Orientation

To make the predictions from the RdDEM comparable to those from SAP, equivalent weather files were created. SAP uses monthly average values of the local regional weather for EPC calculations (SAP, 2012): monthly external temperatures are given in SAP 2012 Table U1, monthly wind speeds are given in SAP 2012 Table U2 and, monthly solar radiations on horizontal surfaces are given in SAP 2012 Table U3 (SAP, 2012).

An equivalent hourly weather file was created for the RdDEM that was based on the average values given in SAP 2012. To achieve this, "typical weather year" data from the International Weather for Energy Calculations (IWEC) (IWEC, 2001) were used. The IWEC data was available for different regions of UK. Since all dwellings in the dataset were located in Midlands region of the UK, the IWEC weather data for Birmingham was used as the basis for creating the equivalent SAP Midlands weather data.

The hourly data for external temperature, wind speed and solar radiation in IWEC Birmingham weather data was averaged for each month. Using the SAP average values for Midlands, a conversion factor for converting monthly averaged Birmingham weather data to SAP data was calculated for each of external temperature, wind speed and solar radiation monthly values. Then, all the hourly values in IWEC Birmingham weather file were multiplied by the corresponding conversion factor. In this way, an equivalent weather file which had the matching monthly average values to SAP Midlands weather data was created.

Due to unknown orientation of dwellings in the dataset, SAP orientation of East-West was used in modelling dataset dwellings. As a result, the main windows are located on East and West facing walls of the dwellings.

Detailed description of the calculations used to achieve monthly averages of external temperature, wind speed and solar radiation, and the resultant values in the equivalent weather file are presented in Section 5.3.

3.4. Translation of Data: The RdDEM Translator

A translator is a code script that converts a programme written in one programming language into functionally equivalent programme in a different programming language without losing the logical structure of the original programme. In this study, the translator was a piece of code script that converted the reduced data into input data file format required for EnergyPlus, without changing the characteristic nature of data. The RdDEM translator script was written in MATLAB R2015a software package to convert XML files in the dataset into EnergyPlus version 8.3.0 Input Data File (IDF). Both XML and EnergyPlus IDF are text based formats used to store data and could be accessed, read and modified by most of available text editors. MATLAB is an object-oriented programming software which has advantage over simple text editors in handling text based formats. MATLAB allowed storing XML files in form of MATLAB Structures while converting the data in XML format into EnergyPlus IDF. Use of MATLAB Structures provides flexibility in handling large datasets and speeds up the translation process.

In the translation process, there were two types of data which needed to be handled differently. The first set of data were exactly the same for all of the dwellings in the modelling dataset and therefore could be translated into the IDFs only once. This fixed set of data included: zoning details, a scalable rectangular geometrical layout, a full set of construction materials, heating systems and heating periods, simulation details, and weather data. The second set of data varied from dwelling to dwelling and needed to be translated individually for each dwelling. This varying set of data included: internal mass, geometry and internal boundary condition details.

The fixed data were identified and written to a template IDF (TIDF) that the translator then filled with the varying data for each dwelling. A detailed description of the TIDF, the RdDEM translator and the checks performed to ensure effective and robust function of translation process is provided in Section 6.2 and 6.3. The translation process was tested using the reference model with different levels of detail and different modelling strategies:

- i. A detailed dynamic model created by hand in DesignBuilder.
- ii. A manually reduced dynamic model created by hand in EnergyPlus.
- iii. An automated reduced dynamic model in EnergyPlus, created using the RdDEM translator.
- iv. A steady-state model, created by hand in SAP.

Results from four variants of the reference model were compared and presented in Section 6.3.

Having tested and verified the performance of RdDEM translator, a further study was carried out to verify the code script for the data preparation process. To do so, the RdDEM predictions of three of the dwellings from the modelling dataset were compared to more detailed DesignBuilder models' predictions of the same three dwellings. All of the results of the RdDEM translator and data preparation verification together with the RdDEM predictions are presented in Chapter 6.

3.5. Summary

This chapter provided an overview of the methods for developing the RdDEM. A suitable modelling dataset was identified from DEFACTO project, which completes objective 2 (Identify a suitable dataset of UK dwellings to be used as the source of modelling data). The modelling dataset is formed of EPC XML files which contain reduced data on 83 semi-detached dwellings located in the Midlands region of the UK. A new data preparation process was developed for defining zoning and enhanced geometry, defining equivalent construction and thermal mass and, defining equivalent boundary conditions. This data preparation process will be tested in Chapter 4 (Zoning and enhanced geometry) and Chapter 5 (Equivalent construction materials and boundary conditions) to complete objective 3 (Develop and test a data preparation process that will enhance the reduced data in order to produce an equivalent set of detailed data that is suitable for dynamic energy simulation). Finally, the RdDEM was completed by defining a translation process in order to create EnergyPlus IDFs as detailed in Chapter 6, which completes objective 4 (Develop and run a reduced data dynamic energy model of UK dwellings that translates the prepared data into a form suitable for dynamic simulation using established models).

4. ZONING AND ENHANCED GEOMETRY

4.1. Introduction

The first step in the data preparation process of the RdDEM is to define the zoning and enhanced geometry of the building from its reduced data (see Section 3.1). This chapter describes the processes that were used to identify the most suitable zoning strategy and the best way to enhance the geometry in pursuit of Objectives 3 (Develop and test a data preparation process that will enhance the reduced data in order to produce an equivalent set of detailed data that is suitable for dynamic energy simulation) and 4 (Develop and run a reduced data dynamic energy model of UK dwellings that translates the prepared data into a form suitable for dynamic simulation using established models). The chapter starts with describing the 'Reference Model' (Section 4.2). The reference model uses a fully detailed input dataset and provides a comparator for investigating the impact of different zoning strategies and geometry enhancement techniques on the model predictions. The various strategies were developed based on the reduced data available in the modelling dataset and the suitability of each strategy was investigated through comparison with the detailed reference model.

The process for choosing the most suitable zoning strategy is described in Section 4.3 and the process of developing the most suitable technique to enhance the geometry is explained in Section 4.4. An enhanced geometry modelling technique is developed based on the reduced geometry details available in the modelling dataset. The DesignBuilder software package version 4.6.0.015 (DesignBuilder, 2015) was used to create the reference models and all subsequent models required to test the strategies and techniques. This version of Design Builder uses EnergyPlus version 8.3.0. EnergyPlus (US Department of Energy, 2012) is a well-known and powerful multizone building simulation tool which has more international recognition to its competitors (including IES VE which is used mainly in the UK for developing EPCs for commercial buildings). Besides, EnergyPlus is an open source code which provides more flexibility in automating the data translation.

4.2. Reference Model

The reference model was a two storey, semi-detached house, as described by Allen and Pinney (1990) - hereafter referred to as A&P. This house was chosen as the reference model because it represented a built form similar to the houses in the dataset (i.e. semi-detached); and possessed sufficient details so that no major assumptions were required.

4.2.1. Building Geometry

Building geometry details are provided as floor plans and building elevations by A&P, and re-created in Figure 4.1. Ground floor has floor-to-ceiling height of 2.40 m while floor-to-ceiling height of the first floor is 2.30 m. Internal and external walls have thickness of 0.14 m and 0.29 m, respectively. House geometry was modelled in using the exact dimensions from A&P. As suggested by A&P, double height was applied to the stair cases by creating a hole on the ceiling of ground floor hall (floor of first floor landing). Size of the hole is not specified explicitly by A&P but based on floor plans it was derived to be 2.3 m long and 0.95 m wide.





Both of the houses in the semi-detached pair were modelled, with the attached house being a mirror image about the party wall. Figure 4.2 presents the front and back view of the semi-detached A&P house model in DesignBuilder. External walls and internal floors were created by extruding wall and floor thickness towards the inside of the building (DesignBuilder, 2015). Hence, in creating the house model, internal floor thickness (0.255m) was added to the first floor height and external walls' thickness (0.29 m) was added to each side of floor plans. The roof and ground floor were treated as separate building elements and their thickness was not included in the floor heights.



Figure 4.2 Front (left) and back (right) view of the semi-detached A&P house model in DesignBuilder

The width and height of each window was entered separately, including the frame according to the window corner definition in DesignBuilder (Figure 4.3). No information was given about window dividers in A&P; hence, no window dividers were modelled in the reference model.

Internal and external doors were modelled explicitly. The dwelling has two external doors, one in each of ground floor hall and kitchen and the front door has a window occupying one third of the door area. This window increases the door U-value from 2.5 W/m²K to 3 W/m²K. To simplify for DesignBuilder, the door was modelled as solid wood having a U-value of 3 W/m²K to represent presence the same rate of heat transfer.





Note that dividers were not present in the reference model.

A&P do not provide any information on the dimensions of the roof: e.g. height, pitch and overhang. The roof of the reference model was modelled with 35° pitch, 0.3 m overhang and 2.5 m height to match the roof of the semi-detached house model in a similar study by Yilmaz et al. (2014). The roof was set to be a semi-exterior, unconditioned zone which was included in thermal calculations in the DesignBuilder.

4.2.2. Construction Materials

Construction details for the reference model, describing the materials used in each layer of the building elements and providing physical properties for these materials, were all taken from the A&P house description (Table 4.1).

The external wall type was a cavity wall with bricks on both sides (outer leaf brick has higher conductivity compared to inner leaf) and plaster on the inner side. Internal and party walls were constructed of only the inner leaf brick and plastered faces. Internal ceilings/floors were air cavity enclosed with timber and plasterboard layers. Roof was composed of roof tiles on outside with glass fibre quilt insulation and plasterboard.

A new glazing type with 6 mm clear float glass was created in DesignBuilder, using the simple glazing definition. All windows were single glazed with timber frame covering 30% of overall window area. The glazing had an overall U-value of 4.3 W/m²K. Window frames and doors were composed of softwood with 0.03 m and 0.07 m thickness, respectively. The glazing has total solar transmission of 0.78 and light transmission of 0.88.

The building ground floor was modelled as a 100 mm concrete slab which matched the U-value of 0.74 W/m²K given by A&P.

In EnergyPlus dynamic energy simulations, the average monthly ground surface temperatures under the building is used as the outside surface temperature for all surfaces adjacent to the ground. According to EnergyPlus documentation, the undisturbed ground temperatures calculated by EnergyPlus's weather converter program are often not appropriate for building heat loss calculations as these values are too extreme for the soil under a conditioned building (US Department of Energy, 2013). EnergyPlus documentation suggests using ground temperatures of 2°C below mean internal temperatures for large commercial buildings in the US. However, it does not suggest any method for calculating or estimating ground surface temperature for small residential buildings. Lstiburek (2008) suggests that a reasonable rule of thumb to estimate the ground surface temperature is to use the average annual ambient air temperature of that location. In absence of any other reference, the average annual ambient air temperature of 10°C was calculated from the weather file (Table 4.4) and used for all months of the year. Detailed description of the weather file is presented in section 4.2.6.

	Construction Materials as Described by A&P						
Building Element	Material	Thickness (m)	Density (kg/m³)	Thermal Conductivity (W/mK)	Specific Heat Capacity (J/kgK)		
	Plaster (medium density)	0.016	800	0.26	1000		
External Walls	Brick (inner leaf)	0.105	1700	0.62	800		
wans	cavity	0.065	N/A	N/A	N/A		
	Brick (outer leaf)	0.105	1700	0.84	800		
Internal	Plaster (medium density)	0.016	800	0.26	1000		
and party	Brick (inner leaf)	0.105	1700	0.62	800		
Walls	Plaster (medium density)	0.016	800	0.26	1000		
	Carpet	0.005	160	0.06	1000		
Internal	Timber	0.020	650	0.14	1200		
floors	Cavity	0.200	N/A	N/A	N/A		
	Plasterboard	0.010	950	0.16	840		
	Timber	0.020	650	0.14	1200		
Coiling	Cavity	0.200	N/A	N/A	N/A		
Cennig	Plasterboard	0.010	950	0.16	840		
Roof	Plasterboard	0.010	950	0.16	840		
	Glass fibre quilt	0.105	250	0.04	840		
	Roofing tiles	0.010	1900	0.84	800		
Window	Glazing	0.006	2500	1.05	750		
	Softwood (Frame)	0.03	230	0.12	2760		
Doors	Softwood	0.07	230	0.12	2760		
Ground Floor	Concrete slab	0.100	2400	0.16	880		

Table 4.1 Construction materials, each layer's thickness, physical properties of building elements (density, thermal conductivity and specific heat capacity

4.2.3. Internal Gains

A&P gives the rates and timings of internal heat gains from occupants, cooking, lighting, refrigerator, television and hot water for each room (Table 4.2). Separate profiles were created in DesignBuilder for each of occupancy, appliances and lighting gains in each room in order to model the identical heat gains. Figure 4.4 shows occupancy heat gain profiles for each room. It can be seen that the living room is occupied for 6 hours in the evening, while the dining room and the kitchen are occupied for one hour in the morning for breakfast and two hours in the evening for dinner. Bedrooms 1 and 3 (children's bedroom) are occupied for 12 hours and the main (parent's) bedroom is occupied for 9 hours during night. The latent (40%) and sensible (60%) split of heat gains, given by A&P, were used for the metabolic and hot water gains.

Lighting gain profiles were created in DesignBuilder for individual rooms (Figure 4.5). No lighting gains are given for bedrooms as A&P consider that these rooms are occupied only for sleeping purpose. As seen in Figure 4.4 and Figure 4.5, lighting profiles coincide with occupancy gain profiles which show that rooms are lit only when they are occupied.

Table 4.2 A&P rates and times of occurrence of internal heat gains from occupants, cooking, lighting, refrigerator, television and hot water (re-created from (Allen and Pinney, 1990))

Room	Internal Gains (W)								
Туре	Occupants	Light	τv	Cooker	Fridge	Hot water			
Living room	17.0-23.0 (144)	17.0-23.0 (212)	17.0-18.0 (135) 20.0-22.0 (158)						
Dining room	08.0-09.0 (140) 18.0-20.0 (115)	08.0-09.0 (126) 18.0-20.0 (171)							
Kitchen	07.0-08.0 (84) 18.0-21.0 (84)	07.0-08.0 (56) 18.0-21.0 (56)		07.0-08.0 (1190) 18.0-21.0 (1700)	00.0-24.0 (60)	00.0-24.0 (77)			
Bedroom 1	00.0-09.0 (38) 21.0-24.0 (38)								
Bedroom 2	00.0-08.0 (148) 23.0-24.0 (148)								
Bedroom 3	00.0-09.0 (38) 21.0-24.0 (38)								
Bathroom	07.0-08.0 (100) 17.0-18.0 (40) 21.0-23.0 (35)	07.0-08.0 (100) 17.0-18.0 (40) 21.0-23.0 (35)				00.0-24.0 (77)			



Figure 4.4 Occupancy gain profiles for individual rooms of A&P semi-detached house model



Figure 4.5 Lighting gain profiles for individual rooms of A&P semi-detached house model

4.2.4. Heating System

The heating system for the reference model comprised a central heating and radiators. A&P give no details on boiler type and so a regular natural gas boiler with 85% efficiency was used. These were modelled using DesignBuilder's detailed HVAC option (DesignBuilder, 2015).

The daily heating periods given by A&P are 07:00 to 09:00 and 16:00 to 23:00 for the heating season: 1st October to 31st May. This was modelled using the boiler operation availability schedule of DesignBuilder's circulating hot water loop data (DesignBuilder, 2015). The radiators availability schedules were set to be always 'ON'. There was no information available regarding the pipe run in A&P; and, all the pipes in the system were therefore assumed to be adiabatic in the reference model.

A&P provides heating set point temperatures for each room (Table 4.3) and so each room of the dwelling was modelled as a separate thermal zone which resulted in 9 different thermal zones. The zone type of the roof was set to a semi-exterior unconditioned, unoccupied and there was no heating or cooling assigned.

Room type	Heating set-point temperature	Room Type	Heating set-point temperature
Living room	21°C	Bedroom 1	18°C
Dining room	21°C	Bedroom 2	18°C
Kitchen	18°C	Bedroom 3	18°C
Hall and Landings	16°C	Bathroom	22°C

Table 4.3 Heating set-point temperature for individual rooms semi-detached dwelling (re-created from (Allen and Pinney, 1990))

4.2.5. Ventilation and Infiltration

A&P provide mean infiltration rates for each room and also a constant infiltration rate for the whole house as derived from whole house and single room measurements. These measurements were conducted in an unoccupied house with closed windows and A&P suggest that occupants will open and close windows in response to prevailing conditions and higher overall infiltration rates would be closer to reality; however, for the purpose of inter-model comparison and model verification they suggest a single infiltration rate of 0.7 ACH (Allen and Pinney, 1990). Hence, ventilation window opening was excluded from the reference model and fabric infiltration of 0.7 ACH was assigned to each room.

There was no information on the ventilation rate of the loft space in the A&P house description. The ventilation rates of loft spaces have been measured in a number of other studies. Dietz et al. (1986) conducted detail multi-zone PFT gas measurements in a number of homes in the US and reported 3 ACH as "typical" for ventilation rate of loft spaces. I'anson et al. (1982) measured loft space ventilation rate of 4.3 ACH in a middle terraced three-bedroom house using three tracer gases. Allinson (2007) modelled ventilated pitched roofs during low wind speed conditions in the UK and chose a ventilation rate of 2 ACH according to assumptions by Burch (1980). For the reference model, the same Burch assumption was employed and an infiltration rate of 2 ACH was assigned to the roof zone.

4.2.6. Orientation and Weather Data

A&P doesn't specify any building orientation or external weather. The reduced form of SAP (RdSAP) assumes an East-West orientation for all dwellings (RdSAP, 2012) but Yilmaz et al. (2014) modelled the same A&P building and used South-North orientation. In order to compare the results of the reference model with the work of Yilmaz et al (2014) the same South-North orientation was used here with the front of building facing South.

The weather file was also the same as used by Yilmaz et al. (2014). It was CIBSE 'Kew67' and has been widely used in other modelling studies (Shorrock et al., 1996). This weather file is in (.csv) format and an EnergyPlus weather file format (.epw) version was created using the EnergyPlus custom weather data translation tool (EnergyPlus Weather Convertor, 2015). The Kew67 weather file is summarised in Table 4.4.

Month	EWY monthly mean Temperature (°C)	EWY monthly mean Wind speed (m/s)	EWY monthly mean horizontal global radiation (W/m ²)
Jan	4.4	4.9	27.5
Feb	4.1	3.9	39.2
Mar	6.4	4.2	99.6
Apr	9.1	3.8	142.5
Мау	12.9	3.9	194.2
Jun	15.5	3.5	205.4
Jul	15.5	3.7	159.2
Aug	15.5	3.1	162.5
Sep	13.1	3.1	112.5
Oct	9.4	2.7	75
Nov	8.7	3.5	29.2
Dec	4.9	4.8	19.6

Table 4.4 Mean monthly temperature, wind speed and horizontal global radiation of Kew67 Example Weather Year (EWY) weather file

4.2.7. Results of the Reference Model

The reference model was verified by comparing the space heating energy consumption to the result of 8,491 kWh/year reported by Yilmaz et al. (2014). The reference model predictions showed space heating energy consumption

of 8,451kWh/year, which is only 1.5% lower. The small difference between the two predictions may be due to slightly different assumptions:

- Yilmaz et al. (2014) treated internal doors as internal walls while in this study internal doors were modelled explicitly. This affects the modelled thermal mass in as internal doors have a lower heat capacity.
- Yilmaz et al. (2014) modelled the party wall as an adiabatic wall, while in this study the neighbouring dwellings and party wall were modelled explicitly (see Section 4 and Figure 4.2).
- Yilmaz et al. (2014) did not provide details on infiltration rate of the roof space so it is not known if the same assumption of 2 ACH was used.

4.3. Zoning

This section describes the process of determining the most suitable zoning strategy for creating a model from reduced data in which room layout and dimensions are not given. Three simplifications of the thermal zoning were trialled to determine which simplification best reproduced the predictions of the reference model (where each room was modelled as an individual thermal zone).

4.3.1. Zoning Strategies

The first zoning strategy, 'SAP' zoning, employed the two zones defined by SAP (2012): the living area and the rest of the house. According to SAP (2012), living area is the room marked on a plan as the lounge or living room, or the largest public room, irrespective of usage by particular occupants. Hence, a thermal zone was assigned to the living room of the reference model (Figure 4.1) and all the remaining thermal zones were combined to create the second thermal zone in the model.

The second zoning strategy, 'Floor' zoning, was developed such that each floor of the house was a separate thermal zone. The third zoning strategy, 'Single' zoning, was developed by combining all the thermal zones in the reference model to create a single zone model

The three zoning strategies are illustrated by complexity in Figure 4.6. Although the 'Floor' and 'SAP' strategies have the same number of zones, 'SAP' zoning is geometrically more complex than 'Floor' zoning as the ground floor must be divided and the location of living room is not known in the modelling dataset. This also adds extra complexity in modelling internal gains and set-point temperatures.



Model complexity

Figure 4.6 Zoning strategies, their level of complexity and number of zones

In applying each of the zoning strategies to the reference model, the internal gains assigned to each room (section 4.2.3) were combined and averaged for each zone by floor area. Heating set-point temperatures in each of the

simplified zones were also averaged by floor area as summarised in Table 4.5. All other properties of the models, such as building geometry, construction materials, heating system, ventilation, infiltration, and orientation were independent of zoning strategy and were kept the same as the reference model.

Room type	Reference	SAP zoning		Floor zoning		Single zone
Bedrooms	18°C	Living		Ground		
Living room	21°C	room	21°C	floor	19.5°C	
Dining room	21°C					18.9°C
Kitchen	18°C	Rest of		First		
Bathroom	22°C	the	18.4°C	floor	18.3°C	
Hall/Landing	16°C	house				

Table 4.5 Heating set-points averaged over zone area in each zoning strategy

4.3.2. Simulation Results

Two different simulations were run for the reference model and the models with each zoning strategy. This was so that the performance of the models could be compared in different conditions: summertime (May to September) with no internal gains, and wintertime (October to April), with heating and internal gains. This tested different aspects of the assumptions used for the zoning strategies.

Internal temperature predictions from the three zoning strategies were compared to those from the reference model for summertime as seen in Figure 4.7. All three zoning strategies under-predicted maximum mean daily internal temperature by about 1°C in comparison with the reference model (Figure 4.7 (a)). Minimum mean daily internal temperature (Figure 4.7 (b)), is

over-predicted by all three zoning strategies with 'Floor' zoning showing results slightly closer to the reference model.

Monthly mean internal temperature graphs (Figure 4.7 (c) and (d)) show a similar trend to the daily graphs and all zoning strategies predicting higher maximum monthly temperatures and lower minimum temperatures compared to the reference model. Figure 4.7 (c) shows 'Floor' zoning predicts maximum mean monthly internal temperatures better than the other zoning strategies. In all graphs presented in Figure 4.7 the 'Single' zoning strategy gives the poorest predictions. These results suggest that in hot summer weather conditions, all zoning strategies have the potential to underestimate the number of overheating hours compared to the reference model. However, 'Floor' zoning is marginally better for overheating risk assessment.

The simulation results for winter conditions (Figure 4.8) show a larger difference between predictions. All three zoning strategies predict lower maximum mean daily temperatures in the winter (Figure 4.8 (a)) with 'SAP' zoning giving closer predictions to the reference model. During the entire winter period, 'SAP' zoning predicts maximum mean daily temperatures within about 0.5°C of the reference model. All three zoning strategies predict warmer minimum mean daily temperatures compared to the reference model (Figure 4.8 (b)). The maximum and minimum mean monthly temperatures have similar trend to mean daily maximum and minimum values (Figure 4.8 (c) and (d)). 'SAP' zoning predicts maximum mean monthly temperatures that are closer to the reference model during heating season and, similarly to the summer results, 'Single' zoning gave the worst predictions compared to the reference model.

Overall, the 'Single' zone strategy was not suitable, 'Floor' zoning gave better predictions of internal temperatures in summer condition and 'SAP' zoning was better under winter conditions. In order to decide on which of these zoning strategies is more suitable, space heating demand predictions from each zoning strategy were also analysed.



Figure 4.7 Internal temperatures of the three zoning strategies and the reference model under summer conditions. a) Maximum mean daily temperatures, b) minimum mean daily temperatures c) maximum mean monthly temperatures, d) minimum mean monthly temperatures



Figure 4.8 Internal temperatures of the three zoning strategies and the reference model under winter conditions. a) Maximum mean daily temperatures, b) minimum mean daily temperatures c) maximum mean monthly temperatures, d) minimum mean monthly temperatures
As seen in Figure 4.9, throughout the heating season 'SAP' zoning predicted the highest space heating demand each month while the results for 'Floor' zoning were closer to the reference model. A similar trend is observed in the annual space heating demand predictions shown in Table 4.6. Therefore, 'Floor' zoning was chosen as the most suitable strategy for implementation in the RdDEM.



Figure 4.9 Monthly space heating energy demand of the three zoning strategies

Table 4.6 Space heating energy demand of the three zoning strategies compared to the reference model

Zoning strategy	Space heating demand (kWh/year)	Difference to the reference model
Single zone	8612	1.9%
SAP zoning	8802	4.0%
Floor zoning	8463	0.1%
Reference	8451	-

4.4. Enhanced Geometry

This section describes the process of choosing the best method to enhance the reduced geometry data available in the modelling dataset. The predictions were compared to those from the reference model and a method for including extensions was added.

4.4.1. Modelling Geometry

The modelling dataset includes numerical values for floor area, floor height, party wall length, and heat loss perimeter but there are no details of the three-dimensional geometry of the dwellings. The aim in enhancing the geometry was to preserve the values given, while creating the full three-dimensional geometry. Three possible potential layouts were considered for a hypothetical dwelling in the dataset (Figure 4.10). It can be seen that ratio of floor area to party wall length to heat loss perimeter differs in each of the layouts. There is not one simple geometry that will work in all cases.



Figure 4.10 Possible geometry layouts and party wall location for the dwellings in the dataset

Figure 4.11 shows the methodology that was developed to enhance the geometry, which maintains the heat loss areas and preserves the reduced geometry details available in the dataset: party wall length, heat loss perimeter and floor height. The length of the created building is equal to the party wall length and its width is derived from the heat loss perimeter and party wall length (Equation 4.1).

$$W = \frac{P_{HL} - L_{PW}}{2} \tag{4.1}$$

Where 'W' is the width, ' P_{HL} ' is the heat loss perimeter and ' L_{PW} ' is the party wall length. Hence, the footprint area of the modelled building (A_{Model}) becomes (Equation 4.2):

$$A_{Model} = L_{PW} \times W \tag{4.2}$$

In this way heat loss perimeter, party wall length and floor height given in the dataset are all preserved. The resulting floor area was then checked against dwellings' actual floor area. If the actual building layout was rectangular (layout '1' in Figure 4.10), then the modelled building's floor area would be the same as the dwelling and no further processing would be necessary. However, if the actual building had non-rectangular layout (layouts '2' and '3' in Figure 4.10); two cases would be possible.

The first case is where the floor area of the actual dwelling is larger than that of the model (left branch of the graph in Figure 4.11). In this case the width (*W*) in the model was replaced with a dummy width (W_{dummy}) which was derived from the actual dwelling's area (*A*) and party wall length (L_{PW}) in Equation 4.3.

$$W_{dummy} = \frac{A}{L_{PW}}$$
 4.3



Figure 4.11 The methodology developed to model dwellings' geometry using the reduced data available in the modelling dataset

This preserves the floor area and party wall length but increase the heat loss perimeter. To preserve the heat loss perimeter, an adiabatic wall was added to the model with length (Equation 4.4):

$$W_{Adiabatic} = W_{dummy} - W \tag{4.4}$$

The second case is where the floor area of the model is larger than that of the actual dwelling (right branch of the graph in Figure 4.11). In this case a block with zero heat capacity was added to the middle of the modelled building to remove the excess floor area (A_{Excess}) and the additional volume of room air.

In applying this geometry enhancement methodology to the houses in the modelling dataset, it was found that the model always had a floor area larger or equal to the actual dwellings' floor area. Hence, the first case, where modelled area was smaller than the actual floor area, never happened in any of the modelling dataset houses.

A modified version of the reference model described in Section 4.2 was used to test the impact of modelling an L-shaped layout as a rectangular block. The reference model was extended to create an L-shaped layout. The width of the building was unchanged and an additional 25% was added to the floor area as an extension using DesignBuilder. As a result, individual window areas and internal wall areas were also increased by 25%. This 'geometry reference model' was created as a two zone model, separating ground and first floor following the findings of the zoning study in Section 4.3. The same total heat gains were used which resulted lower heat gains per square meter (due to the increased floor area). The lumped heat gain values were decreased to 1.6 W/m² and 0.9 W/m² for ground and first floors, respectively. The ground floor had heating set-point of 19.5°C and first floor had a heating set-point of 18.3°C. All other details were kept the same as the reference model.

To test the geometry enhancement method, the L-shaped layout of the geometry reference model was converted into a rectangular layout, following

Figure 4.11, and modelled in DesignBuilder as the 'design geometry model'. Figure 4.12 shows the reference model, the geometry reference model and, the design geometry model. The geometry reference model had floor area of 110.5 m² and party wall length of 15.4 m. The design geometry model's length was kept as 15.4 m and its width was derived from Equation 4.1 as 8.6 m. The rectangular design geometry model had floor area of 132.4 m².



Figure 4.12 Left to right: The reference model, the reference geometry model and the design geometry model as created in DesigBuilder software package

Figure 4.13 shows the process of transforming the reference geometry model to design geometry model created in DesignBuilder. A block with zero heat capacity was added to the middle of rectangular model to remove the excess floor area of 21.9 m². The space inside the extra block was excluded from thermal and radiance daylighting simulations and therefore it had no impact on model predictions.



Figure 4.13 Transforming the reference geometry model to design geometry model

4.4.2. Simulation Results

Simulations were run for one complete year on the reference geometry and design geometry models. Predictions from the two models were compared to test the geometry enhancement method. Figure 4.14 shows the comparison of monthly space heating demand, infiltration and solar gain; and Figure 4.15 shows the comparison of daily and monthly internal temperatures. It can be seen that there is very close alignment between monthly infiltration and solar gains. The difference in space heating demand was less than 1% in all months and the annual space heating demand was within 3 kWh/year. The monthly difference between internal air temperature predictions (Figure 4.15) did not exceed 0.5°C. This close alignment of the predictions demonstrated that this method for enhancing the geometry was suitable for use in the RdDEM.



Figure 4.14 Monthly averaged space heating demand, external infiltration and solar gain estimates of the design geometry model compared to the reference geometry model



Figure 4.15 Internal temperatures of the reference geometry model and the design geometry model. a) Maximum mean daily temperatures, b) minimum mean daily temperatures c) maximum mean monthly temperatures, d) minimum mean monthly temperatures

4.4.3. Modelling Extensions

More than half of the houses in the modelling dataset have one or more extensions. Extensions are additions to a house that were built after the main building and therefore have a different age band and construction materials compared to the main building. Hence, the methodology was extended to include extensions in the model. This was necessary as the extensions formed a considerable proportion of the dwellings' floor area and would impact model predictions.

The reduced dataset includes age band and wall/floor type for extensions as well as floor area, floor height and heat loss perimeter, but no details on the location of extensions. Figure 4.16 shows five possible ways that extensions could be included. The red lines show external walls of the extensions. The first three solutions consider the extension in different locations. The last two solutions combine the floor area and heat loss perimeter of the extension with that of the main building. A method was chosen to reduce the complexity and preserve the geometry.





Solutions four and five introduced the lowest number of vertices compared to the other solutions and were therefore preferred. Also, modelling extensions as separate blocks was problematic as the location was not known and this would impact on the location of fenestration and amount of solar heat gain into the building model. Solution four had the lowest number of vertices and therefore the least complexity but made it difficult to maintain the heat loss perimeter of the main building and extensions. Hence, the five-vertex model was chosen for implementation in the RdDEM.

To better understand the five-vertex solution, imagine a simple rectangular geometry with a square extension (Figure 4.17). The extension has floor area of 4 m² and exposed perimeter of 6 m while the main building has floor area of 40 m² and exposed perimeter of 16 m. The extension is combined with the main building while keeping the party wall length and increasing the width of rectangular geometry.



Figure 4.17 Modelling the extensions using five-vertex solution

The resultant five-vertex geometry has an extra area of 12 m² which is removed by introducing a block with zero heat capacity (as described in Section 4.4.1). As seen in the five-vertex geometry in Figure 4.17, the exposed perimeter of the extension (shown in red) and exposed perimeter of the main building (shown in black) are both conserved. In this way, different construction materials can be assigned for extension and main building walls, and party wall.

The five-vertex solution to model extensions was tested on the reference model to examine its suitability for modelling the houses in the dataset. Since the original reference model didn't have any extensions, a single-storey and a two-storey extension were added to the reference model (Figure 4.18). The single storey extension's floor area was 25% of the ground floors, and the two storey extensions floor area was 25% of the total floor area. The roof of the extensions was modelled in a similar way to the main building and window areas were increased in proportion with floor area. The total heat gains remained the same and therefore decreased per unit floor area, to 1.6 W/m^2 and 0.9 W/m^2 for ground and first floors.



Figure 4.18 The reference model with single storey and two storey extensions

The dwelling models shown in Figure 4.18 were re-created using the fivevertex solution such that the total floor area, party wall length and exposed perimeter of both the main building and extensions were conserved. Simulations were run, and annual space heating demands were compared (Figure 4.19).

As seen in Figure 4.19, the prediction of space heating demand for the fivevertex solution were 2.3% lower than the reference model. The close alignment between predictions was also observed in mean monthly internal temperatures and the five-vertex solution for modelling the houses with extensions was deemed to be suitable for use in the RdDEM.



Figure 4.19 Annual space heating demand comparison of the reference model with extensions with five-vertex solution

4.5. Summary

This chapter described the processes that were used to identify the most suitable zoning strategy and the best way to enhance the geometry. These were required because the modelling dataset (reduced data) did not include internal layout of the dwellings or the three-dimensional geometry required for dynamic energy simulation.

A reference model, based on a semi-detached dwelling that was similar to dwellings in the modelling dataset, was defined and modelled in detail. The model results were verified by comparison with results in the literature. The reference dwelling was then used to test zoning strategies and ways to enhance geometry.

Zoning the dwellings floor by floor was found to be better than using a single zone or a separate living room zone. The predictions of annual energy demand and internal temperatures were similar to the detailed model when this zoning strategy was used.

A method was developed to preserve all of the values given in the modelling dataset (floor area, party wall length and heat loss perimeter) while creating full three-dimensional geometry. The method was tested on an extended version of the reference model and a very close alignment was found (within 1%). The methodology was also tested when extensions were present, and a similar close alignment was observed.

The zoning strategy and method for enhancing the geometry were included in the RdDEM. This completed Objectives 3 (*Develop and test a data preparation process that will enhance the reduced data in order to produce an equivalent set of detailed data that is suitable for dynamic energy simulation*) and 4 (*Develop and run a reduced data dynamic energy model of UK dwellings that translates the prepared data into a form suitable for dynamic simulation using established models*) in part. The generation of further enhanced and equivalent data for the RdDEM is continued in the next chapter.

5. EQUIVALENT CONSTRUCTION MATERIALS AND BOUNDARY CONDITIONS

5.1. Introduction

This chapter describes the processes used in the RdDEM to derive SAP equivalent input parameters for construction materials, thermal mass, and the internal and external boundary conditions, towards Objective 4 (Develop and run a reduced data dynamic energy model of UK dwellings that translates the prepared data into a form suitable for dynamic simulation using established models) and completing Objective 3 (Develop and test a data preparation process that will enhance the reduced data in order to produce an equivalent set of detailed data that is suitable for dynamic energy simulation). The modelling dataset only provides information on construction types and does not give any details about materials used in different building elements. Section 5.2 explains the process of re-creating construction materials for various building elements based on reduced data such that the U-value of each building element match the values given by SAP. The process of matching the thermal mass to SAP assumptions is also described. The chapter continues with describing the process of deriving SAP equivalent internal boundary conditions: internal heat gains and heating systems, losses, infiltration and ventilation (APPENDIX D). The chapter concludes with describing the methodology used to develop SAP equivalent external boundary conditions, i.e. weather data (Section 5.3).

5.2. Equivalent Construction Materials

The RdDEM contains a library of constructions that are equivalent to those in SAP for the external walls, roofs, ground floors, doors and windows. In this way they have the same U-values as those identified in SAP 2012 (see Section 3.3.3). The constructions were created in DesignBuilder and added to the IDF template (see Section 6.2.1) for use with the RdDEM.

5.2.1. External Walls

In case of the external walls, four set of constructions were required to represent the houses in the modelling dataset, using Table S6 for U-values and Table S3 for wall thickness (SAP, 2012):

- i. Solid brick walls in age bands 'B', 'C, and 'D' with U-value of 2.1 $$\rm W/m^2K.$$
- ii. Solid brick walls in age band 'E' with U-value of 1.7 W/m²K.
- iii. Cavity walls in age bands 'B', 'C, 'D' and 'E' with U-value of 1.6 W/m²K.
- Filled cavity walls in age bands 'B', 'C, 'D' and 'E' with U-value of 0.5 W/m²K.

Each wall type was re-created using DesignBuilder's construction materials library in order to achieve the same overall U-value and thickness of each wall type (Table 5.1). The construction materials created in the DesignBuilder had an increased overall external walls U-value (by 0.15) to account for thermal bridging (see Section 3.3.3). Hence, the four modelled wall types mentioned above had overall U-values of 2.4 W/m²K, 2 W/m²K, 1.8 W/m²K, and 0.6 W/m²K, respectively.

As seen in Table 5.1, a single type of brick, plaster and insulating foam was used to create all four of the external walls. The only difference between solid walls of age bands 'B' to 'D' and solid walls in age band 'E' was addition of a thin air gap. This air gap is added to the plaster board for the purpose of dry-lining and is different from the air gap in cavity walls. Dry-lining is a plaster boarding system where an air gap is created between the wall and the plaster board, improving the U-value of a solid wall by about 25%. Dry-lining adds 20 mm to 40mm to the solid wall thickness (RdSAP mannual, 2012). This addition in the thickness also explains the difference in the thickness of solid brick walls from age bands 'B' to 'D' compared to the solid walls from age band 'E'.

Wall type	Construction Details						
and U- value as created in DB	Materials (outermost to innermost layer)	Thickness (m)	Density (kg/m³)	Thermal Conductivity (W/mK)	Specific Heat Capacity (J/kgK)		
Solid Brick	Brick	0.205	1700	0.77	1000		
(U=2.4 W/m ² K)	Dense plaster	0.015	1300	0.57	1000		
Solid	Brick	0.205	1700	0.77	1000		
Brick (U=2.0 W/m²K)	Air gap	0.020	N/A	N/A	N/A		
	Dense plaster	0.015	1300	0.57	1000		
	Brick	0.105	1700	0.77	1000		
Cavity	Air gap	0.035	N/A	N/A	N/A		
(0=1.0 W/m²K)	Brick	0.105	1700	0.77	1000		
	Dense plaster	0.015	1300	0.57	1000		
Filled	Brick	0.105	1700	0.77	1000		
Filled Cavity (U=0.6 W/m ² K)	Foam (phenol-rigid)	0.035	110	0.035	1470		
	Brick	0.105	1700	0.77	1000		
,	Dense plaster	0.015	1300	0.57	1000		

Table 5.1 SAP equivalent external wall constructions, thickness and the physicalproperties of each layer of materials

Physical properties of all the materials specified in Table 5.1 were checked against CIBSE Guide A, Appendix 3.A7: Properties of materials (CIBSE Guide A, 2017) to ensure the correctness of values reported by DesignBuilder.

5.2.2. Roofs

All of the dwellings in the modelling dataset had pitched roofs. The majority had a known thickness of insulation, ranging from 0mm to 300 mm. The level of roof insulation was unknown for 5 of the dwellings and therefore the thickness was assumed based on Table S10 SAP 2012. This table assumes no insulation for pitched roofs in age bands 'B' to 'D' and 12 mm of insulation for the pitched roofs in age band 'E'.

Each of these roofs was re-created in DesignBuilder such that each roof had the same U-value as specified in SAP 2012. The roof U-values given in SAP 2012 accounts for the insulation, the roofing materials and the thermal resistance of the air space in loft. In DesignBuilder, however, the pitched roof construction only includes the external sloped surfaces and loft insulation, consequently the reported overall U-vale doesn't take into account the resistance of the air space in loft. Hence, in the RdDEM, the roofing materials and loft insulation were modelled explicitly, and the U-values were compared to SAP values using the Equation 5.1.

$$U_{SAP} = \frac{1}{R_{Space} + \frac{1}{U_{DB}}}$$
5.1

Where U_{DB} is the overall U-value reported by DesignBuilder and U_{SAP} is the U-values in Table S9 SAP 2012 including thermal resistance of loft space (R_{Space}) for pitched roofs. The thermal resistance of the roof space (R_{Space}) for tiled roofs was taken from Table 3.5 in CIBSE Guide A, as 0.06 m²K/W.

The materials used in roof construction were clay tiles, glass fibre quilt and roofing felt. The thickness of insulation layer was adjusted accordingly such that the roofs U-values match the values given in SAP. Physical properties of the materials used in roof construction are presented in Table 5.2.

Roof materials (outermost to innermost layer)	Density	Thermal conductivity (W/mK)	Specific heat capacity (J/kgK)
Clay tile	1900	0.85	840
Glass fibre quilt	12	0.04	840
Roofing felt	960	0.19	840

Table 5.2 Roof construction materials and corresponding density, thermal conductivity and specific heat capacity values

5.2.3. Ground Floors

The dwellings in the modelling dataset had two different types of ground floor construction: solid and suspended timber. In the RdDEM, ground floors were modelled such that they have the same U-values as in SAP. SAP calculates the ground floor U-value according to BS EN ISO 13370 using dwelling's area (A) and exposed perimeter (P) and following parameters:

- wall thickness (w)
- soil type clay (thermal conductivity λg = 1.5 W/mK)
- $R_{si} = 0.17 \text{ m}^2 \text{ K/W}$ (Internal surface resistance)
- R_{se} = 0.04 m²K/W (External surface resistance)
- thickness and conductivity of floor insulation (0.035 W/mK)
- R_f = 0.001 x *d_{ins}*/0.035 where *d_{ins}* is insulation thickness in mm (R_f is the thermal resistance of floor deck)

For solid floors, U-value of the ground floor is calculated by SAP as (Equations 5.2 and 5.3):

If
$$d_t < B$$
 $U = 2 \times \lambda_g \times \ln(\pi \times \frac{B}{d_t} + 1)/(\pi \times B + d_t)$ 5.2

If
$$d_t > B$$
 $U = \lambda_g / (0.457 \times B + d_t)$ 5.3

Where d_t and B are calculated from Equations 5.4 and 5.5:

$$d_t = w + \lambda_g \times (R_{si} + R_f + R_{se})$$
 5.4

$$B = 2 \times A/P$$
 5.5

Since the solid ground floors in the modelling dataset had no insulation, the value of R_f was inserted as zero in Equation 5.4. The U-values for solid ground floors were calculated individually for each dwelling using dwellings' floor area and exposed perimeter.

In the RdDEM, the solid ground floors were modelled as three layers: underfloor clay, cast concrete, and flooring screed. The thickness of each layer was adjusted in order for the individual ground floors to match the SAP U-values. The physical properties of the materials used in ground floor construction are presented in Table 5.3.

Ground floor materials (outermost to innermost layer)	Density	Thermal conductivity (W/mK)	Specific heat capacity (J/kgK)
Clay (earth)	1.28	1460	880
Cast concrete	1900	1.4	840
Flooring screed	1200	0.41	1000

Table 5.3 Ground floor construction materials and corresponding density, thermal conductivity and specific heat capacity values

SAP uses additional parameters to calculate the U-value of suspended timber ground floors. These parameters include:

- thermal resistance of floor deck $R_f = 0.2 \text{ m}^2 \text{ K/W}$ if uninsulated or $R_f = [(\text{thermal resistance of insulation}) + 0.2]$ if insulated
- height above external ground level h = 0.3 m
- average wind speed at 10 m height v = 5 m/s
- wind shielding factor $f_w = 0.05$

- ventilation openings per m exposed perimeter $\varepsilon = 0.003 \text{ m}^2/\text{m}$
- U-value of walls to underfloor space $U_w = 1.5 \text{ W/m}^2\text{K}$

The suspended ground floor U-value is calculated from Equation 5.6 in SAP:

$$U = 1/(2 \times R_{si} + R_f + 1/(U_g + U_x))$$
5.6

Where U_g and U_x are calculated from Equations 5.7 and 5.8:

$$U_g = 2 \times \lambda_g \times \ln(\pi \times \frac{B}{d_g} + 1) / (\pi \times B + d_g)$$
 5.7

$$U_{x} = \left(2 \times h \times \frac{U_{w}}{B}\right) + (1450 \times \varepsilon \times v \times \frac{f_{w}}{B})$$
 5.8

And d_g and *B* are calculated from Equations 5.9 and 5.10:

$$d_g = w + \lambda_g \times (R_{si} + R_{se})$$
5.9

$$B = 2 \times A/P \tag{5.10}$$

Since the suspended ground floors had no insulation, the value of R_f was inserted as 0.2 in Equation 5.6. The U-values for suspended ground floors were calculated individually for each dwelling using dwellings' floor area and exposed perimeter.

In the RdDEM, suspended ground floors were modelled as a solid ground floor with an insulation layer. In this way, the thickness of insulation layer was modified accordingly for each dwelling to reflect the SAP equivalent U-value of the suspended ground floor as calculated in above equations. Hence, although all dwellings model had solid ground floor, the U-value of the ground floor was capture correctly for both the dwellings with solid and suspended ground floor. This was simpler than trying to model a ventilated cavity, with uncertain ventilation rates, in EnergyPlus.

5.2.4. Doors and Windows

For the RdDEM, the area of external doors was taken as 1.85 m² following the SAP 2012 guidelines. The external doors were modelled on the front and rear walls of each dwelling. Doors were modelled as softwood with, 0.07 m thickness, 0.12 W/mK thermal conductivity, and 230 kg/m³ density to match the U-value of 2.5 W/m²K as given by Table S15A SAP 2012 for dwellings with age bands A to J.

Window areas in the RdDEM were the same as in SAP, based on the age band and total floor area (*TFA*) from Table S4 SAP 2012. Table 5.4 shows the related part of table S4 which was used in estimating window area of the modelling dataset houses. Window areas were divided equally onto the front and rear external walls of the dwellings.

Table 5.4 Window area estimated based on age band and total floor area (re-
created from Table S4 (SAP, 2012)	

Age band	Window area (m ²)
A, B, C	0.1220 <i>TFA</i> + 6.875
D	0.1294 <i>TFA</i> + 5.515
E	0.1239 <i>TFA</i> + 7.332

The only known detail in the dataset about windows was that all dwellings had 100% double glazed windows. Hence, based on Table 6E SAP 2012 (SAP, 2012) a double glazed, air filled window with 6 mm gap and U-value of 3.1 W/m²K was considered for all the dwellings. The windows were modelled with an effective U-value which took account of the assumed use of curtains ($U_{w,effective}$), as show in Equation 5.11 from SAP 2012:

$$U_{w,effective} = \frac{1}{\frac{1}{U_w} + 0.04}$$
5.11

Where U_w is the window U-value without curtains.

5.2.5. Equivalent Thermal Mass

The thermal mass of the ground floor, ceiling, external walls, party walls, windows and doors is captured in the equivalent constructions. However, the modelling dataset does not include any details of internal partitions or furniture that add to a building's thermal mass. These elements may contribute considerably to the total thermal mass and so to make the models equivalent, the SAP 2012 guidelines were used.

In SAP a Thermal Mass Parameter (TMP) is calculated from Equation 5.12 (SAP, 2012).

$$TMP = \frac{\sum(\kappa \times A)}{TFA}$$
 5.12

Where ' κ ' is the heat capacity of construction materials (kJ/m²K), and the summation is over the area (A) of all elements (windows, door, external walls, party walls, ground floors and roofs) bounding the dwelling as well as both sides of all internal walls and floors/ceilings. '*TFA*' is the total floor area. The ' κ ' values for some typical constructions are given in Table 1e (SAP, 2012).

However, for RdSAP, an indicative value of the TMP is used instead of a detailed calculation. Table 1f (SAP, 2012) gives a fixed value for the TMP of 250 kJ/m²K. The RdDEM used this fixed TMP value to create a dynamic model with equivalent thermal mass by adding internal walls to create a TMP of 250kJ/m2K.

EnergyPlus does not calculate thermal mass in the same way as SAP; instead the materials within each building elements is defined with heat capacity, density and thickness values used to run heat transfer equations. Since the construction details of internal walls was not identified in the modelling dataset, the 'dense block and dense plaster' construction type given in Table 1f SAP 2012 with heat capacity (κ) value of 100 kJ/m²K was assumed. The internal wall area, on the other hand, was derived from Equation 5.12 with the TMP value set to 250 kJ/m²K. To do so, area and heat

capacity (κ) values from all other building elements were inserted into Equation 5.12 and then the equation was solved for internal walls area. The area of building elements were derived from geometry details provided in the dataset while the heat capacity (κ) was calculated from Equation 5.13 after the method described in (SAP, 2012).

$$\kappa = 10^{-6} \times \sum (d_j \rho_j c_j)$$
 5.13

Where ' d_j ' is the thickness (mm), ' ρ_j ' is density (kg/m³) and ' c_j ' is specific heat capacity (J/kgK) of the layers forming each building element. The summation is over all layers in the element from inside to outside until one of the following conditions is met: half way through the element; an insulation layer; total thickness of 100 mm.

5.3. Equivalent External Boundary Conditions: Weather Data

A new weather file, with hourly data suitable for use in dynamic energy simulation, was created for use in the RdDEM. The weather data were created to match the monthly external temperature, wind speed and global solar irradiance values given in SAP 2012 Tables U1 to U3 (SAP, 2012). The International Weather for Energy Calculations (IWEC) weather file for Birmingham was used as the basis of the weather file, since all the dwellings in the modelling dataset were located in UK Midlands.

The IWEC Birmingham weather file is derived from an average of 18 years (1982-1999) of hourly observations archived at the US National Climatic Data Centre. The weather data is supplemented by solar radiation estimated on an hourly basis from earth-sun geometry and hourly weather elements, particularly cloud amount information (IWEC, 2015).

The mean monthly external temperature, wind speed and solar irradiance values from IWEC Birmingham weather file were used to calculate conversion factors for each month. Table 5.5 compares the monthly

temperature, wind speed and solar irradiance values from SAP Midlands to IWEC Birmingham values. It can be seen that the conversion factors were relatively close to 1, with the biggest difference being 0.35 for solar irradiance in December.

Month	External Temp (°C)			Wind speed (m/s)			solar irradiance (W/m²)		
	SAP	IWEC	CF	SAP	IWEC	CF	SAP	IWEC	CF
Jan	4.3	4.6	0.94	4.5	5.2	0.87	28	67	0.42
Feb	4.8	3.7	1.30	4.5	3.1	1.45	55	96	0.57
Mar	6.6	6.4	1.03	4.4	3.9	1.13	97	150	0.65
Apr	9.0	7.5	1.2	3.9	4.7	0.83	153	169	0.91
Мау	11.8	11.0	1.07	3.8	4.6	0.83	191	164	1.16
Jun	14.8	14.2	1.04	3.4	3.6	0.94	208	179	1.16
Jul	16.6	17.2	0.97	3.3	3.4	0.97	194	166	1.17
Aug	16.5	16.3	1.01	3.3	3.3	1	163	150	1.09
Sep	14.0	13.2	1.06	3.5	3.3	1.06	121	116	1.04
Oct	10.5	9.9	1.06	3.8	3.6	1.06	69	93	0.74
Nov	7.1	6.9	1.03	3.9	3.9	1	35	76	0.46
Dec	4.2	5.0	0.84	4.1	3.5	1.17	23	65	0.35

Table 5.5 Mean monthly external temperature, wind speed and solar irradiance from UK average weather data (SAP 2012) and IWEC Birmingham (IWEC, 2001) weather data showing the conversion factors (CF)

The conversion factors were applied to the hourly values in the IWEC Birmingham weather file. For wind direction, relative humidity and atmospheric pressure, the same values in IWEC Birmingham weather file were used. In this way the weather was equivalised to the SAP Midlands data while maintaining the hourly values required for dynamic simulation. Additionally, the Latitude, Longitude and elevation of location from see level for all dwellings were set to 52.6 °N, -1.33 and 116 m, respectively as specified in Table U4 SAP 2012. The SAP Midlands equivalent hourly data was inserted into a CSV file and the EnergyPlus weather convertor programme (EnergyPlus Weather Convertor, 2015) used to generate the EPW file.

5.4. Summary

This chapter described the processes used in the RdDEM to derive SAP equivalent input parameters for construction materials, thermal mass, and the internal and external boundary conditions. Construction materials were defined for all of the different building elements (i.e. external walls, roofs, ground floors, doors and windows) such that the U-values matched those given in SAP.

The RdDEM maintained the same Thermal Mass Parameter (TMP) of 250 kJ/m²K as defined in SAP. This was achieved by calculating the thermal mass in the external walls, party wall, roof and ground floor and then adding sufficient area of internal partition wall to make up the remainder. Internal boundary conditions were defined for use in the RdDEM which matched those given in SAP: internal heat gains, losses, infiltration and heating systems. A weather file from IWEC Birmingham, suitable for dynamic thermal simulation, was modified to match the monthly external temperature, wind speed and solar irradiance values from SAP.

Overall, this completed Objectives 3 (*Develop and test a data preparation process that will enhance the reduced data in order to produce an equivalent set of detailed data that is suitable for dynamic energy simulation*). Objective

4 (Develop and test a reduced data dynamic energy model of UK dwellings that translates the prepared data into a form suitable for dynamic simulation using established models) is completed in the next chapter.

6. TRANSLATION, SIMULATION AND RESULTS

6.1. Introduction

This chapter describes the translations process used in the RdDEM to convert the reduced data from the XML format used for EPCs into the Input Data File (IDF) format used by EnergyPlus to complete objective 4 (*Develop and test a reduced data dynamic energy model of UK dwellings that translates the prepared data into a form suitable for dynamic simulation using established models*). The translation process was tested and verified. The results of using the RdDEM to simulate 83 dwelling from the DEFACTO dataset were compared with the SAP results in pursuit of the final objective: Objective 5 (*Compare the results of the reduced data dynamic energy model with those from equivalent steady-state models*).

6.2. Translation

Data translation is the part of the RdDEM that converts all of the prepared data into IDF format in readiness for running the EnergyPlus simulations. The translation process was formed of two main elements: the Template Input Data File, which was used for storing fixed input data, and the Translator, to handle varying input data.

The fixed data was written into a standard IDF template which formed the basis of the IDF for all the dwellings. This Template IDF (TIDF) contained all the fixed data and the lines allocated to varying data were left blank to be filled by the translator. The translator wrote the varying input data into the allocated spaces in the TIDF. The following sub-sections provide details of the TIDF and the translator; and the procedures followed to ensure the effective and robust function of translation process.

6.2.1. Template Input Data File (TIDF)

The TIDF was created from a two zone version of the reference model (see Section 4.3) using DesignBuilder, exported as an IDF and then edited to leave only the relevant information required by the RdDEM. The main components of the TIDF are: zoning details, geometrical layout, construction details, heating system details, and simulation details.

The TIDF was developed from the two zone strategy, with each floor of the dwellings modelled as individual heating zones (Section 4.3). In this way, all the details related to internal thermal mass, geometrical details, internal boundary conditions and construction materials were assigned to each floor of the dwellings separately.

Geometry in the TIDF followed the rectangular method, with an excess area block inside (Section 4.4). The geometry of the dwellings were defined in the TIDF through (x,y,z) coordinates (Figure 6.1) of the 16 vertices forming the blocks. The translator for the RdDEM (see Section 6.2.2) would then edit the (x,y,z) coordinates to suit an individual dwelling in the modelling dataset. As seen in Figure 6.1, the origin was set to the bottom left vertex of the main building block so that all other vertices have positive (x,y,z) coordinates.



Figure 6.1 Geometrical layout of the dwellings' model with red dots showing the vertices of the rectangular blocks, the origin vertex and the IDF notation

The construction details developed for all of the different external walls, roofs and ground floors (see Section 5.2) were included in the TIDF so that the translator could call the appropriate one for each dwelling. The heating system details and heating periods (see APPENDIX D) were also stored in the TIDF while heating set-point temperatures were left as blank to be completed by the translator.

The last set of fixed data in the TIDF was related to simulation details. All simulations were run with 10-minute time steps (6 time steps per hour) for a full year under the SAP equivalent weather data file (see Section 5.3). The simulation details, the address of the weather (CVS, EPW and 'definition') files, and the commands needed to store all the results in hourly, monthly and annual formats were stored in the TIDF.

In the TIDF unique text strings were added so that the translator could identify the right place for input parameters. These indicators were added as comments which were recognised by the translator but not the EnergyPlus compiler.

6.2.2. Translator

The Translator wrote all of the varying data into a TIDF for each dwelling in the modelling dataset (Figure 6.2). The translator reads each XML files in the modelling dataset and converts them into MATLAB structures. This speeds up the translation process by treating the whole dataset as a MATLAB directory and also removes the complications of handling text files. The translator calculates the internal thermal mass for each dwelling in terms of additional internal wall area (see Section 5.2.5). Then the 16 sets of (x,y,z) coordinates for each floor were calculated based on the rectangular geometry describe in Section 4.4. Finally, the internal boundary conditions were calculated (see APPENDIX D). After completing the data preparation process for each house, the translator wrote the data into the TIDF. The translation process was repeated for all the dwellings in the modelling dataset to create a unique IDF for each.



Figure 6.2 Translator framework to process varying data from XML files and write input data into Template IDF (TIDF)

6.3. Process Verification

Two lots of simulations were carried out to verify the translation process: 1) Diagnostic runs, and 2) Final runs. The primary goal of the diagnostic runs was to debug the translator script, troubleshoot any errors that occurred and to ensure that the data preparation process was implemented correctly. The final runs were aimed at verifying both the data preparation and translation processes, including checking the equivalent rectangular geometry, equivalent thermal mass and zoning strategy.

The diagnostic runs were performed on four variants of the reference model:

- i. A detailed dynamic model in DesignBuilder.
- ii. A manual reduced dynamic model in EnergyPlus.
- iii. An automated reduced dynamic model in EnergyPlus, using the RdDEM translator.
- iv. A reduced steady-state model in SAP 2012.

The detailed reference model developed in DesignBuilder is described in Section 4.2. All the reduced models were developed based on input data from a reduced XML file including reference model details. In order to have exactly same amount of details as the modelling dataset, the data on the semi-detached Allen and Pinney (1990) house was stored in an EPC XML file format. The SAP model was developed manually using SAP 2012 spreadsheet in APPENDIX B. All the dynamic energy models were simulated with 10-minute time step, for a full year under the SAP equivalent weather data file (see Section 5.3).

The annual space heating demands from the four variants of the reference model were compared (Figure 6.3). The two reduced models developed in EnergyPlus predicted annual space heating demand very close to each other (less than 1% difference). The very close alignment of the two reduced EnergyPlus models verified performance of the translation process.



Figure 6.3 Comparison of the space heating demands from different variants of the reference model

The predictions from the two reduced EnergyPlus models were 10% different to the detailed DesignBuilder model and 5% different to the SAP model. The close alignment between reduced EnergyPlus and SAP models supports the decisions made in the data preparation process to develop SAP equivalent data. However, larger difference between reduced EnergyPlus models with the detailed DesignBuilder model highlights the impact that simplifying model has on the model predictions. The main parameters simplified in developing reduced models were: geometry, thermal mass, and zoning. All internal boundary conditions were also simplified after SAP 2012 guidelines, but the close alignment of reduced models with SAP verified these simplifications. Hence, further investigation was necessary to verify geometry, thermal mass, and zoning simplifications.

In order to verify zoning strategy (Section 4.3), enhanced geometry (Section 4.4) and equivalent thermal mass (Section 5.2.5) techniques used to develop the RdDEM, the model predictions of the three of houses in the modelling

dataset were compared to predictions of more detailed models of the same houses. These test houses were selected based on their annual space heating demand estimated by SAP, such that they represent bottom, median and top demand values in the batch. The approximate building plans generated by the EPC assessors were available on the three test houses. Hence, the detailed building geometry, thermal mass and zoning configuration of these houses were modelled in DesignBuilder and simulations were run for a full year under the SAP equivalent weather data file. All other aspects of the detailed models were kept similar to the RdDEM.

The annual space heating demand from the RdDEM, detailed DesignBuilder and SAP models were compared (Figure 6.4). A general trend was observed in all of the test houses where the RdDEM and detailed DesignBuilder models underestimated the annual space heating demand compared to the steady-state SAP results and detailed DesignBuilder model underestimated the annual heating demands compared to the RdDEM.



Figure 6.4 Comparison of the space heating demand from SAP, RdDEM and detailed DesignBuilder models of the three test houses

While the difference between RdDEM and SAP results increases with increasing space heating demand, the difference between RdDEM and detailed DesignBuilder model is similar in all the test houses. The increasing difference of SAP predictions with RdDEM and detailed DesignBuilder model predictions highlighted the characteristic differences of the steady-state and dynamic energy models. However, the steady difference between RdDEM and detailed DesignBuilder model predictions, which was less than 5% different in all the studied houses, verified the data preparation process developed to model geometry, thermal mass and zoning using reduced data.

The difference observed between RdDEM and detailed version of the reference model was 10%, while the difference observed between RdDEM and detailed versions of the three test houses was less than 5%. The smaller difference observed in case of the test houses, which were chosen from the modelling dataset, showed that the reduced geometry, thermal mass and zoning strategies worked better on the dataset dwellings.

6.4. Simulation Results

The RdDEM results were compared with SAP predictions for 83 houses in the modelling dataset. This included annual space heating demand (Section 06.4.1), mean monthly internal temperatures (Section 6.4.2) and the estimated potential for improving energy efficiency of the houses (Section 6.4.3). All of the simulations were run in EnergyPlus version 8.3.0 using IDFs created in the RdDEM. Simulation of each house required approximately 8 minutes of single CPU time for a full year simulation at 10-minute time steps on a CORE i5 HP laptop running Microsoft Windows.

6.4.1. Comparison of Energy Demand Results to SAP Estimates

Annual spaces heating demand results from the RdDEM were compared to SAP predictions for the 83 houses in the modelling dataset (Figure 6.5 and Table 6.1). The (x=y) line, shown in blue, is where RdDEM predicts the same

space heating demand as SAP. The line of best fit through the data has the equation shown in Equation 6.1 and a coefficient of determination (R^2) of 0.96.

$$y = 0.82x + 1834.6$$
 6.1

The minimum difference observed between RdDEM and SAP was 74 kWh/year (1%) while the largest difference was 5898 kWh/year (17%). Of the 83 modelled houses, 46 were within 5% difference in annual space heating demand prediction and only 5 had more than 10% difference with only 2 more than 15%. The closest results are for the houses with space heating demand below approximately 15000 kWh/year.

Over/Under	Ву (%)	Number of houses (%)
RdDEM predicts	> 10%	2 (2%)
higher space heating	5-10%	9 (11%)
demand than SAP	< 5%	12 (14%)
RdDEM predicts	< 5%	34 (41%)
lower space heating	5-10%	23 (28%)
demand than SAP	> 10%	3 (4%)

Table 6.1 Comparison of space heating demand predictions between RdDEM and SAP

The box-whisker plots (Figure 6.6) shows the differences in the distributions of results from the two models. The RdDEM predictions have lower mean, median, maximum and minimum values of the annual space heating demand for 83 modelled houses. However, the mean and median values are remarkably close.


Figure 6.5 Comparison of space heating demand predictions for 83 modelled dwellings

The RdDEM predicts lower mean, median, maximum and minimum values of the annual space heating demand for 83 modelled houses. The overall distribution of annual space heating data was tighter compared to SAP. Bottom half of data (first quartile) showed closer alignment between SAP and RdDEM predictions compared to top half (third quartile) which once again highlights better alignment of RdDEM and SAP estimates in houses with less than 15000 kWh/year annual space heating demand. Despite the larger difference observed in the two models predictions in higher annual space heating demands, the close mean and median values show a close alignment of annual space heating demand in majority of the modelled houses.



Figure 6.6 Comparison of the distributions of the annual space heating demand predictions for the 83 modelled dwellings

6.4.2. Comparison of Temperature Predictions to SAP Results

The mean monthly internal air temperature predictions from the RdDEM were compare to SAP for the heating season (Figure 6.7). For SAP, the methodology described in the SAP spreadsheet section 7 (see APPENDIX B) was used to calculate mean internal temperatures for each month. The data points shown in Figure 6.7 are categorized into 5 groups based on the comparison of annual space heating demand predictions:

- i. The houses with less than 5% difference between RdDEM and SAP predictions of annual space heating demand,
- ii. The houses where RdDEM predicts higher annual space heating demand by 5-10%,
- iii. The houses where RdDEM predicts lower annual space heating demand by 5-10%,
- iv. The houses where RdDEM predicts higher annual space heating demand by more than 10% and,
- v. The houses where RdDEM predicts lower annual space heating demand by more than 10%.

As seen in Figure 6.7, RdDEM generally predicts lower mean monthly internal air temperatures throughout the heating season. In most of the months there is a clear difference between data points based on the difference in annual space heating demand predictions. In general, the RdDEM temperatures are higher when the energy demand predictions are also higher. This trend shows that the difference in space heating demand can be explained by the difference in internal air temperatures. The RdDEM tends to predict lower internal air temperatures and consequently lower energy demands.



Figure 6.7 Comparison of RdDEM mean monthly internal temperature predictions to SAP (The black line represents x=y)

The house models which predicted lower annual space heating demand by more than 10%, also predicted lower mean monthly internal temperatures to the same amount in all the months except for February. The distribution of the results for February is slightly different to other months and this could be related to relatively higher solar gains compared to December and January. Overall, the RdDEM gives a wider range of temperatures than SAP. The house models which predicted higher annual space heating demand by more than 10%, however, showed a more stable trend throughout the heating season. More work is required to understand these differences and their implications on the accuracy of predictions. This work, as an inter-model comparison, can only highlight the difference but cannot say which is right or wrong.

The average difference between minimum and maximum indoor air temperatures of the houses predicted by RdDEM is 4.6°C which is considerably larger the 1.3°C difference predicted by SAP. As seen in Figure 6.8, RdDEM gives a wider prediction of internal air temperature than SAP in all the heating season months. The RdDEM tends to predict higher maximum mean temperatures and lower minimums. These distributions suggest that SAP constrains the internal temperature estimates more than RdDEM. This trend also suggests RdDEM is more sensitive to external temperatures than SAP.

The mean, median and first quartile predicted by RdDEM (Figure 6.8) is constantly lower than SAP. However, the third quartile is lower in warmer months and higher in colder months. However, some of this variation cancels out when the annual energy demand is considered (Figure 6.6). Future work could look at a comparison of the monthly energy demand predictions of the two models, but the EPC data used in this study only included annual demand.



Figure 6.8 Comparison of the distributions of mean monthly internal air temperature

6.4.3. Comparison of the Predicted Energy Improvements

This section compares the RdDEM and SAP predictions of the energy savings that result from improving the energy efficiency of the dwellings in the modelling dataset. The introduced energy saving measure was improving U-values of the external walls by 20% by adding wall insulation. The comparison was done on three of the modelling dataset houses. The same test houses which were employed in verifying the data preparation and translation process (Section 6.3) were used in this part of study. The test houses had different external wall types, U-values, age band, total areas, roof insulation thickness, and ground floor types as shown in Table 6.2. All the dwellings had 100% double glazing, and pitched roof. Test house 1 has no extension while test houses 2 and 3 each have one extension.

Improved House	External wall construction	Wall U- value (W/m ² K)	Age band	Total area (m²)	Roof Insulation (mm)	Ground floor type
Test House 1	Filled cavity	0.6	D	79	0	Solid
Test House 2	Solid brick, no insulation	2.4	В	146	100	Suspended
Test House 3	Cavity, no insulation	1.8	D	151	0	Solid

Table 6.2 External wall improvements on the three test houses and corresponding wall types, U-values, age bands, total areas, roof insulation, and ground floor types

The U-value improvement of external walls was done by improving the filling material of the external walls in test house 1 and by adding insulating layers to test houses 2 and 3 (Table 6.3). The insulating layer used to improve the U-values was the same material used in the filled cavity wall type: phenol-rigid foam. The improved U-values shown in the Table 6.3 take into account the thermal bridging in the external walls (see Section 3.3.3).

Improved wall types	Construction Details						
and U- values as modelled in the RdDEM	Materials (outermost to innermost layer)	Thickness (m)	Density (kg/m³)	Thermal Conductivity (W/mK)	Specific Heat Capacity (J/kgK)		
Solid	Brick	0.205	1700	0.77	1000		
Brick (U=1.9 W/m²K)	Foam (phenol-rigid)	0.025	110	0.035	1470		
	Dense plaster	0.015	1300	0.57	1000		
Cavity (U=1.4 W/m ² K)	Brick	0.105	1700	0.77	1000		
	Air gap	0.035	N/A	N/A	N/A		
	Brick	0.105	1700	0.77	1000		
	Foam (phenol-rigid)	0.035	110	0.035	1470		
	Dense plaster	0.015	1300	0.57	1000		
Filled Cavity (U=0.5 W/m ² K)	Brick	0.105	1700	0.77	1000		
	Foam (phenol-rigid)	0.040	110	0.035	1470		
	Brick	0.105	1700	0.77	1000		
	Dense plaster	0.010	1300	0.57	1000		

Table 6.3 Improved external wall constructions, thickness and the physical properties of each layer of materials

The improvements were implemented in both RdDEM and SAP model and resultant energy savings were compared (Table 6.4). Table 6.4 summarises the annual space heating demand predictions before and after implementing the external wall improvements.

Table 6.4 Annual space heating demands from SAP and RdDEM, before and after external wall improvements in the test houses and the percentage reductions

Improved House	Annual space heating demand (kWh/year)				SAP	RdDEM
	SAP	RdDEM	Improved SAP	Improved RdDEM	demand reduction	demand reduction
Test house 1	11067	10624	10713	10221	3.2%	3.8%
Test house 2	19581	18798	18250	17256	6.8%	8.2%
Test house 3	25533	22980	24180	21624	5.3%	5.9%

As seen in Table 6.4, both models estimated the largest improvements for the test house 2 and the lowest for test house 1.

Similar to the annual space heating demand estimates before improvements (Figure 6.5), RdDEM predicted lower improved annual space heating demand compared to SAP. In general, RdDEM predictions for space heating demand were lower than SAP but the savings predictions were slightly higher.

The very small differences (less than 2%) observed in the space heating reduction predictions of the RdDEM and SAP once again verified the data preparation and translation process used for developing a SAP equivalent dynamic energy model. The tendency of RdDEM to predict lower demand improvements, on the other hand, highlighted the characteristic differences of the dynamic and steady-state models.

6.5. Summary

This chapter described the translations process used in the RdDEM to convert the reduced data from the XML format used for EPCs into the Input Data File (IDF) format used by EnergyPlus and completes objective 4 (*Develop and test a reduced data dynamic energy model of UK dwellings that*

translates the prepared data into a form suitable for dynamic simulation using established models). The results of using the RdDEM to simulate 83 dwelling from the DEFACTO dataset were compared with the SAP completing the final objective: Objective 5 (*Compare the results of the reduced data dynamic energy model with those from equivalent steady-state models*).

The translation process used a template IDF (TIDF) which included all of the fixed data required for creating individual IDFs for the dwellings in the modelling dataset. The translator handled the varying data and implemented all of the data preparation process in readiness for being written into the TIDF. The translation and data preparation processes were successfully verified: the process of enhancing the reduced data resulted in predictions that were around 5% higher than a more detailed model.

The RdDEM was successfully used to run EnergyPlus simulations of the 83 dwellings in the modelling dataset, using just the EPC XML data. The RdDEM predicted lower space heating demand in 60 dwellings. The minimum difference between predictions was 74 kWh/year (1%) while the largest difference was 5898 kWh/year (17%). The majority of the results were within 5% and only 5 were estimated with more than 10% difference. It was found that the difference in energy results could largely be explained by differences in the predicted internal air temperatures. The RdDEM predicted lower internal air temperatures in the majority of the modelled dwellings and gave a wider range of mean monthly temperatures in the heating season compared to SAP. Predictions of the energy savings from wall insulation in three of the houses showed reasonable agreement, though the RdDEM predicted lower energy demand and slightly higher percentage savings.

7. DISCUSSION

7.1. Introduction

This chapter provides an overview of the body of work presented in this thesis. The achievements against the aim and objectives are highlighted alongside the contributions to knowledge. The limitations of the work are quantified and finally, the application of the modelling framework presented here is expanded for academia, policy makers and industry.

7.2. Overview of Thesis and Achievements

Saving energy in the residential sector and in particular heating energy is essential to achieve the UK's 2050 carbon emissions reduction target. In recent years, a variety of models have been developed to analyse energy and environmental performance of domestic buildings, and investigate impact of different strategies that are designed to reduce energy consumption, reduce carbon emissions, and improve occupant's thermal comfort. In the UK, as part of government's plans to achieve carbon emission targets, the Energy Performance Certificates (EPCs) were introduced and it was made compulsory to present EPC and improvement recommendations when an existing or new home was let or sold. This initiative resulted in collection of vast majority of EPC data on the UK dwellings. EPCs are developed using government's Standards Assessment Procedure (SAP) which is based on steady-state BREDEM calculations to predict energy consumption and carbon emissions in the dwellings. A critical analysis of UK based domestic energy and emission models, and dynamic energy models of different housing stocks were carried out to complete the Objective 1 (Identify and review literature on the modelling approaches that could be used to forecast energy use and CO₂ emissions in the UK dwellings, and perform a critical analysis of the work that has been undertaken to utilize available reduced data for energy modelling purposes)

The dataset used for modelling was formed of XML files which were created for producing EPCs for 83 semi-detached dwellings located in Midlands region of the UK. In choosing this dataset a few points were considered:

- The dataset was conveniently accessible as it was collected by DEFACTO project which was running in the School of Architecture, Building and Civil Engineering, Loughborough University at the time of this research.
- ii. The XML files provided great advantages in handling and modifying data compared to traditional text based datasets in hard copy format.
- iii. The amount of data provided in the dataset was suitable for modelling the dwellings using the freely available SAP spreadsheet without any need for further modification of data or assumptions, which provided a great opportunity for inter-model comparison of results.
- iv. Besides the XML files, the dataset included approximate floor plans which were used in developing more detailed models of the dwellings to verify the results from the reduced data dynamic energy model.

Selecting the EPC XML files as modelling dataset completed Objective 2 (*Identify a suitable dataset of UK dwellings to be used as the source of modelling data*).

Faced with the problems regarding availability of required data for running dynamic simulations of the UK dwellings, a great deal of time and effort was spent on developing suitable algorithms to enhance the reduced data available in EPC datasets in order to be used in dynamic simulation of the dwellings. The outcome of this part of the work was development of a set of SAP equivalent data which, while remaining equivalent to the original EPC datasets, contained enough details to run dynamic simulations.

In order to develop the equivalent dataset, the areas with insufficient details in the modelling dataset had to be enhanced. Two main issues were identified: Zoning and Geometry. SAP and majority of UK steady-state domestic energy models (described in Section 2.4) are two zone models separating the living area from rest of the house while a few models use only one zone to model whole dwelling. As for dynamic energy models, the Canadian Residential Energy End-use Model (CREEM) developed by Farahbakhsh, Ugursal and Fung (1998) assumed only one thermal zone for main building parts of the modelled dwellings. The He et al. Model developed by He, Lee, Taylor, Firth and Lomas (2014), on the other hand, followed SAP approach in zoning and considered two zones separating the living area from rest of the house. These authors justified their choice of zoning strategy by highlighting the lack of necessary zoning details and identifying literature with suitable outputs to support their choice of zoning strategy.

Prior to this research there was a research gap in identifying the suitable zoning strategy to model the UK dwellings using dynamic simulation. This research investigated different zoning strategies and provided evidence on suitability of the two zone models in dynamic simulation of the UK dwellings, where each floor was assigned with an individual thermal zone. The zoning strategies investigated included: a single zoning strategy, floor zoning and SAP zoning (see Section 4.3). These strategies were compared to a detailed reference model where each habitable room was assigned with an individual thermal zone (see Section 4.2).

The considerable difference in annual and monthly demand estimates of the models with different zoning strategies highlighted the importance of zoning configuration in modelling exercises. The sensitivity of whole building energy models like RdDEM to zoning configuration was observed in the model results. Such sensitivity analysis was missing in the previous modelling studies which added to the uncertainty of model outputs. The sensitivity of zoning configuration was investigated in this thesis using a reference semi-detached dwelling. Expanding the zoning sensitivity analysis findings to other dwelling types would require further investigation which wasn't in the scope of this research.

The reference model used in this study to investigate zoning strategies was created based on Allen and Pinney (1990) Standard Dwellings Types. This document describes the common UK house types with enough detail that could be modelled without further need for assumptions. Allen and Pinney Standard Dwellings Types document is a well-known source of reference and has been used previously in many other modelling studies. Firth, Lomas and Wright (2010) used it to identify the archetypes in Community Domestic Energy Model (CDEM); Taylor, Allinson, Firth and Lomas (2013) modelled the period terraced house from Allen and Pinney (1990) Standard Dwellings Types at nine different levels of detail to study the impacts on energy consumption; Yilmaz, Allinson, Taylor and Lomas (2014) modelled the semi-detached house from Allen and Pinney (1990) Standard Dwellings Types and compared the space heating energy predictions from SAP, EnergyPlus, ESP-r, SERI-RES, BREDEM-8, and BREDEM-12 models.

The reference model was extended to investigate the approach used for modelling detailed geometry using reduced data (see Section 4.4). Lacking geometry details in the datasets is one of the main issues raised by previous dynamic energy modelling studies (Farahbakhsh et al., 1998; Swan et al., 2009; He et al., 2014) and each study dealt with this issue in different way. Swan et al. (2009) assumed a rectangular geometry layout and developed an average width to length ratio which was applied to all the modelled dwellings. He et al. (2014) considered two geometrical layouts: a rectangular and an Lshaped layout. In this PhD the rectangular approach proposed by Swan et al. (2009) was adopted but the width to length ratio which was found to add a considerable uncertainty to model outputs was improved. Instead of applying a fixed ratio to all dwellings, the 'Excess Area Block' approach was developed (see Section 04.4.1). This approached made it possible to model the exact floor area, exposed perimeter, and party wall length as identified in the dataset. The approach was tested on an extended form of the reference model by comparing main outputs from a detailed L-shaped reference model to outputs from a reduced geometry model incorporating the 'Excess Area Block' approach. The monthly mean internal temperatures and space heating demand predictions from the two models were closely matching.

Prior to this study no peer reviewed or documented research had looked into creating detailed geometries in dynamic simulation while staying completely loyal to the reduced dataset. The models developed by Farahbakhsh et al., 1998; Swan et al., 2009, and He et al., 2014 all made assumptions to handle lacking geometry details in the model without presenting a sensitivity analysis of the assumptions made. Consequently, the uncertainty added to model outputs in these studies wasn't quantified. This research, in contrary to previous modelling exercises, avoided introducing new assumptions to model geometry and dealt with the missing geometry details in a novel and efficient way.

Development of the equivalent dataset and enhancing zoning and geometry details made the results from RdDEM comparable to SAP and completed Objective 3 (*Develop and test a data preparation process that will enhance the reduced data in order to produce an equivalent set of detailed data that is suitable for dynamic energy simulation*).

The prepared data was written into EnergyPlus Input Data Files (IDFs) through the Template IDF (TIDF) and the translator developed in MATLAB (see Section 6.2). The data preparation and translation process were tested twice: once using the reference model, and once using three actual dwellings from the modelling dataset. The first test compared model outputs from 4 different models of the reference model with different levels of details. These models were: a detailed DesignBuilder model, a reduced EnergyPlus model with manual data input, a reduced EnergyPlus model with translator data input, and an SAP model. The second test was run on three dataset dwellings which represented the three cuts of the dataset based on space heating demand as reported in EPCs. The RdDEM outputs of these dwellings were compared to more detailed dynamic energy models.

The inter-model comparison with more detailed models has been used by other modelling exercises as well to verify model results. Clarke, Johnstone, Kim and Tuohy (2009) verified results of The Energy Systems Research Unit (ESRU) Domestic Energy Model (EDEM) using detailed models of 5 real houses. The results indicated discrepancies ranging from 3% to -13%, indicating that their modelling approach is a reasonable proxy for the real situation (Clarke et al., 2009).

The development of this model eliminated the additional time and cost associated with collecting further data and modelling each of the homes individually; and completed Objective 4 (*Develop and run a reduced data dynamic energy model of UK dwellings that translates the prepared data into a form suitable for dynamic simulation using established models*). The methods used in developing equivalent data enabled a unique inter-model comparison to be carried out across 83 buildings. Similar results were obtained from dynamic and steady-state models when the equivalent inputs were fed to each model.

This was the first dynamic energy modelling exercise to be undertaken using the EPC data for a set of UK dwellings. The reduced data dynamic energy model introduces a set of new methods for enhancing reduced data in order to create equivalent detailed geometry and zoning information. Ultimately, the techniques developed here can be used to provide new insights into the transient aspects of energy use and indoor air temperatures in the UK housing stock and therefore has value as both a policy and a research tool.

7.3. Model Predictions and Comparison with SAP

Simulations were run using the completed RdDEM and results were compared to equivalent steady-state SAP model to complete the Objective 5 of this study (*Compare the results of the reduced data dynamic energy model with those from equivalent steady-state models*). The model predictions showed RdDEM predicts lower annual space heating demand compared to SAP in majority of the houses. The RdDEM predicted higher annual space heating demand only in 28% of the houses and predicted lower annual demand in 72% of the houses compared to SAP results. The RdDEM's tendency to predict lower space heating demand was previously observed in other studies comparing a dynamic energy model to a steady-state one. Shorrock et al. (1996) modelled the semi-detached dwelling described by

Allen and Pinney (1990) in dynamic energy software ESP-r and SERI-RES, and compared the annual space heating demands to steady-state BREDEM-8 and BREDEM-12 models of the same dwelling. Both dynamic energy models underestimated the annual space heating demand compared to both BREDEM-8 and BREDEM-12 steady-state models. Yilmaz et al. (2014) modelled the same semi-detached dwelling in SAP 2009 and EnergyPlus where the dynamic EnergyPlus model underestimated the annual space heating demand compared to steady-state SAP model.

The RdDEM predicted annual space hating demand in 94% of the houses in the dataset within 10% margin of SAP estimates. The close alignment between RdDEM and SAP predictions verified the data preparation process to develop SAP equivalent input data. When compared to steady-state model estimates, the RdDEM with SAP equivalent input data performed better that the dynamic energy models developed by Shorrock et al. (1996) and Yilmaz et al. (2014). The ESP-r model developed by Shorrock et al. (1996) and the EnergyPlus model developed by Yilmaz et al. (2014) underestimated annual space heating demand with more than 18% difference to BREDEM and SAP steady-state models.

The rate of reduction in space heating demand predicted by RdDEM when subjected to same set of improvements were less than 1.4% different to steady-state SAP predictions. Shorrock et al. (1996) and Yilmaz et al. (2014) also investigated the potential savings from improving building fabric. The difference they found between dynamic and steady-state models ranged from 1.7% to 11.2%. The closer alignment between RdDEM and SAP in predicting space heating demand reduction compared to previous modelling exercises once again approves the successful data preparation process which has resulted in two equivalent models with close results. In the previous studies by Shorrock et al. (1996) and Yilmaz et al. (2014) the dynamic models predicted smaller improvements in space heating demand compared to steady-state models. In contrary to these studies, the RdDEM predicted larger improvement in space heating demand compared to SAP.

7.4. Limitations of the Research

This research addressed limitations of the previous energy models developed for UK houses (summarised in Section 2.7) by introducing a transparent dynamic alternative to traditional steady-state BREDEM calculations. The RdDEM was capable of taking into consideration the complex and dynamic nature of the energy consumption in the dwellings while using the same input data source as the steady-state SAP. However, this research had limitations which should be addressed by the future work. The limitations of this research can be summarised as follows:

- This research used the semi-detached dwelling description from Allen and Pinney (1990) Standard Dwellings Types to develop the reference model which was used to investigate the zoning and geometry enhancements. The Standard Dwellings Types by Allen and Pinney (1990) is a relatively old document and the house descriptions specified in this document had some major differences to the dwellings modelled in this study. These differences were observed in level of external wall insulation, loft insulation, and glazing types. Consequently, the outcomes of zoning and geometry studies were based on an old semi-detached dwellings type with poorer insulation and air tightness which increased the level of uncertainty when the same outcomes were applied to modelled dwellings.
- Despite including the most common semi-detached dwelling types, the model developed in this research doesn't take into account the dwellings with conservatories, room in roofs, and dwellings with more than one extension.
- The SAP equivalent weather data was re-created based on the details provided in SAP 2012 and a typical weather data file for Midlands region of UK. Although this weather file was the closest that could have been achieved to SAP weather data, it includes uncertainty due to insufficient amount of weather details provided by SAP.

- Despite the close alignment achieved between RdDEM and SAP results, this study was not able to conclude which model predicted the space heating demand more accurately and closer to actual consumption of the dwellings.
- The RdDEM was constructed on MATLAB platform. Although this platform is suitable for prototyping, it has not been possible to develop a user interface for the model.

7.5. Implications of the Research

The model developed based on reduced data will have a number of important implications for academia, industry and policy makers. The findings presented in this thesis provided insight into the complex relationship between amount of input data available and the level of complexity of models. Prior to this research many studies had identified the lack of real life data required for energy modelling as main issue in developing robust and detailed models. This study for the first time investigated the possibility of developing detailed dynamic energy models based on reduced data suitable for simple steady-state models. The data preparation process described in this thesis was able to develop a set of SAP equivalent data which was suitable for dynamic simulation. The most important implication of this research for academia and policy makers is that regardless of level of complexity the model has, if the required input data is not fed to the model, the model will fail to capture the reality of energy consumption in buildings. Although dynamic models are capable of solving more complex and detailed heat transfer equations, feeding these models with reduced data will transform them into simple reduced level models which are not capable of using their full potentials to predict realistic energy consumption in dwellings. Hence, the similar performance gap observed between steady-state model results and real life data will be observed in dynamic energy model results as well.

This research also established the grounds to use dynamic energy simulation in evaluating energy and environmental performance of dwellings for policy making decisions and developing more detailed EPCs. Current regulations only make use of dynamic simulation compulsory for commercial buildings due to higher cost associated with gathering required data and also higher level of expertise and time required for developing dynamic energy models. The possibility of using reduced data, which is already available on the UK dwellings, for dynamic simulation purposes provides a great opportunity to use capacities of dynamic models in introducing robust policies. The algorithms developed for producing SAP equivalent input data for dynamic energy simulation proved to be capable of predicting energy consumption and temperature distribution of dwellings with close alignment to SAP predictions. The use of dynamic energy simulation will enable policy makers to use wider range of hourly and sub-hourly model predictions to put stronger and more practical policies into action.

The capacities of a reduced data dynamic energy model will also benefit energy providers by providing hourly consumption predictions which will enable the energy providers to identify peaks of consumption in district or national level. A good example of the implications this research has in energy management industry is the capability of the RdDEM to predict space heating and hot water energy demand of groups of existing dwellings, which are heated by community energy systems. The model can predict likely consumption of community dwellings for the next day or two which will enhance the energy management process by providing energy forecast data for a specific group of dwellings. In this way, the RdDEM will pave the way for developing better community energy management systems, especially in urban areas.

7.6. Summary

This chapter discussed the process of developing SAP equivalent input data for dynamic simulation and development of the reduced data dynamic energy

model (RdDEM). The chapter started with revisiting the methodology employed to investigate the possible design solutions and decisions made in developing the model. The choice of modelling dataset and the reference model used to examine design solutions was discussed and related literature were cited to support the decisions made in the process of preparing data. The two main methodological enhancements made to model zoning and geometrical aspects of the dwellings were also discussed.

The chapter continued with discussion of model results (Section 7.3). Annual space heating demand and mean monthly internal temperature predictions from the model was compared to other modelling studies. The difficulties of data preparation process and developing SAP equivalent input data for dynamic simulation and the limitation of the research were explained in Section 7.4. Finally, the possible implications that the model described in this thesis has for the academia, the policy makers, and the industry were presented in Section 7.4.

8. CONCLUSIONS

8.1. Introduction

This thesis has described the development of a Reduced Data Dynamic Energy Model (RdDEM) for simulating the energy performance of UK houses. The RdDEM eliminates the main drawback associated with dynamic energy modelling, namely: the large amount of required input data compared to steady-state models, by employing a reduced set of data which was originally collected for EPC models. The enhanced zoning and geometry details together with SAP equivalent constructions and boundary conditions were used to develop the RdDEM which is equivalent to the steady-state SAP and the results are comparable.

8.2. Summary of the Main Conclusions

The critical analysis of existing energy models of the UK dwellings revealed the incapability of these models in fully capturing the reality of energy consumption in the dwellings. All the steady-state models developed for UK housing stock were criticised for their low level of transparency. These models were not capable of taking into consideration the complex, interdependencies, and dynamic nature of the energy consumption and carbon emission. Furthermore, the uniformity of the assumptions made by these models resulted in systematic errors that could have negative consequences for energy policy making, and the targeting of energy efficiency measures.

This thesis presented the work undertaken to use a dynamic energy simulation software to overcome the limitations of the steady-state models like SAP. The main concern in using the dynamic simulation to predict energy consumption of the domestic buildings is the larger amount of required input data compared to steady-state models. This issue was overcome by employing the existing EPC datasets as main source of data. Prior to this research, there was no peer reviewed published literature to indicate potential of reduced datasets like EPC, which was originally developed for steady-state models, for dynamic simulation purpose.

The two main area for improvements were identified as zoning and geometry. The methods developed by previous studies to handle reduced zoning and geometry details were improved and employed in the RdDEM.

8.2.1. Methodological Conclusions

The most suitable zoning and geometry modelling strategies using the reduced zoning and geometry details in the modelling dataset were investigated. The reference model, which was developed based on common UK semi-detached dwelling described by Allen and Pinney (1990), was used as a base case and a comparator in investigating suitability of different zoning strategies and geometry modelling techniques. Three different zoning strategies were investigated: a single zone strategy, a two zone strategy assigning individual thermal zones to each floor of the dwellings (floor zoning), and a two zone strategy assigning the living area with one and the rest of the house with another thermal zone (SAP zoning). The summer and winter results were studied and compared to the reference model:

- Under summer conditions, the choice of zoning didn't have significant impact on temperatures but evidence showed that 'Floor' zoning had closer estimates to the reference model. Under winter conditions, choice of zoning showed a more significant impact on indoor temperatures. 'SAP' zoning gave closer estimates to the reference model under winter conditions.
- The 'Floor' zoning strategy showed the closest space heating demand to the reference model. Considering both internal temperatures and space heating demand estimates of the three zoning strategies, choice of 'Floor' zoning was shown to be the most suitable strategy to model the dataset

houses. 'Floor' zoning was also preferred over 'SAP' zoning due to unknown location of the living room in the dataset houses.

 Hence, the 'Floor' zoning was found to be better suited to modelling the UK semi-detached dwellings available in the modelling dataset and this method was chosen for implementation in the RdDEM.

A methodology was developed to preserve all the geometrical details given in the modelling dataset, while creating the full three-dimensional geometry which could be used in dynamic simulation of the dwellings. The proposed methodology introduced an 'Excess Area Block' to the model. In this way the correct floor area, heat-loss perimeter and party wall length, as identified in the modelling dataset, was conserved for all the dwellings. The suitability of this method was tested on an extended version of the reference model.

- The predictions from the model with reduced geometry details were compared to that of the detailed reference model, and a very close alignment within 1% difference was achieved.
- The methodology was also tested when extensions were present and a similar close alignment was observed between the predictions of the model with reduced geometry and detailed reference model. The close alignment of models predictions showed that this methodology was suited to be used in the RdDEM.

8.2.2. Model Verification

The method developed for creating SAP equivalent input data, zoning strategy, and enhanced geometry details were tested and verified through comparison with more detailed models. The verification process was carried out in two steps: once using the detailed reference model and once using detailed models of three actual dwellings from the modelling dataset.

The close alignment of the results from RdDEM with SAP verified the developed methodology for creating SAP equivalent input data. On the other hand, the close alignment between RdDEM and more detailed dynamic model predictions, which was less than 5% different in all the studied houses, verified the data preparation process developed to model geometry, thermal mass and zoning using reduced data.

8.2.3. RdDEM Results and Comparison with SAP

Simulations were run on the 83 modelled houses. The annual space heating demand and monthly internal temperatures predictions from the RdDEM were compared to the SAP results. The estimated reductions in the annual space heating demand from the two models when subjected to same set of external wall improvements were also compared. Following is a summary of main conclusions from the comparisons of the RdDEM and SAP predictions:

- The RdDEM predicted higher annual space heating demand in 28% of the modelling dataset houses and lower annual space heating demand in 72% of the houses compared to SAP estimates. The differences observed between RdDEM and SAP was 1-17%.
- Annual space hating demand in 55% of the houses in the modelling dataset was predicted with less than 5% difference to SAP, in 39% of the houses with 5-10% difference and only in 6% of the houses the predicted annual space heating was more than 10% different to SAP results.
- The RdDEM as a dynamic simulation tool predicted slightly lower internal temperatures and consequently the lower energy demands compared to SAP in majority of the modelled dwellings.
- The RdDEM gave a wider range of mean monthly temperatures in heating season compared to SAP which suggested the assumption made in SAP constrains the temperature predictions.

• The RdDEM predicted lower improved space heating demand compared to SAP for the three studied test house.

8.3. Recommendation for Future Research

The work undertaken within this thesis is an important first step towards understanding the potential of equivalent dynamic energy models over currently used steady-state models for building evaluation and policy making decisions. In this context, several areas of future research have been identified:

- The work presented here has concentrated on the development of equivalent SAP data for a set of semi-detached dwellings. All the decisions were made based on studies performed on the UK common semi-detached dwellings. To get this model to work for entire housing stock, one of the most important areas of future work would be to include other dwellings types (Detached, terraced, and flats) in the data preparation process for the model. This would require identifying further construction materials, new geometries and zoning configurations.
- The model results were verified through comparison of space heating demand and indoor temperatures to more detailed models and to SAP predictions. The other very important area of future work would be to compare the RdDEM predictions with measurements in the modelled dwellings when heating system settings and internal gains are matched.
- The model predictions provide a great opportunity for future studies to investigate impact of future weather conditions on existing dwellings. A detailed overheating analysis based on hourly results when the internal boundary conditions are matched with TM59 would provide insight into the measures that will be required to be taken in the future in order to maintain occupant's comfort in the dwellings.

 Once this model is completed, the algorithms developed for creating SAP equivalent inputs from reduced data will be of significant value as a dynamic alternative to current steady-state policy making tools when transient variations in energy demand and indoor air temperatures are required.

REFERENCES

Allen, E. A. And Pinney, A. A. (1990) Standard dwellings for modelling: details of dimensions, construction and occupancy schedules. Building Research Establishment (BRE), UK.

Allinson, D. (2007) Evaluation of aerial thermography to discriminate loft insulation in residential housing. PhD thesis, The University of Nottingham. Available from: <u>http://etheses.nottingham.ac.uk/284/</u>

Anderson B.R., Chapman P.F., Cutland N.G., Dickson C.M., Henderson G., Henderson J.H., et al. (2002) BREDEM-12 model description: 2001 Update. BRE. Watford, UK.

Anderson B.R., Chapman P.F., Cutland N.G., Dickson C.M., Doran S.M., Henderson G., et al. (2002) BREDEM-8-model description: 2001 Update. BRE. Watford, UK.

ASHRAE (2009) 2009 ASHRAE handbook: fundamentals. SI. Atlanta, GA, American Society of Heating, Refrigerating and Air-conditioning Engineers.

Barrett, M. A. (1981) A Dynamic Physical Model of the United Kingdom. Ph.D. Thesis.

Building Energy Tools Directory (BETD) [online] (2014) Available from: <u>http://www.eren.doe.gov</u> (follow 'EERE Offices' to 'Emerging Technology' and then 'Building Energy Modelling')[Accessed 11 February 2015]

Boardman, B. (2007) Examining the carbon agenda via the 40% House scenario. Building Research and Information, 35 (4): 363-378.

Boardman, B., Darby, S., Killip, G., Hinnells, M., Jardine, C.N., Palmer, J. And Sinden, G. (2005) 40% House project. Environmental Change Institute. [Online]. Available from:

http://www.eci.ox.ac.uk/research/energy/downloads/40house/40house.pdf

Bowman, N.T. & Lomas, K.J. (1985) Empirical validation of dynamic thermal computer models of buildings. Building Services Engineering Research and Technology, 6(4), pp.153–162. Available at:

http://bse.sagepub.com/cgi/doi/10.1177/014362448500600403

BRE (2005) The UK government's standard assessment procedure. Available: http:// projects.bre.co.uk/sap2005 [accessed 19.06.15].

BRE (2011) The UK Government's Standard Assessment Procedure for Energy Rating of Dwellings 2009 edition incorporating RdSAP 2009. Watford, UK.

Burch, D.M. (1980) Infrared audits of roof heat loss. ASHRAE Transactions, Denver, Colorado. ASHRAE Vol.86, Part 2, p.209-225.

Buswell, R., Marini, D., Webb, L. And Thomson, M. (2013) Determining heat use in residential buildings using high resolution gas and domestic hot water monitoring. Proceedings of 13th conference of international building performance simulation association, August 26-28, Chambery, France.

Chen, Z., And O'Neill, P. (2001) Processing demand modulates the effect of spatial attention on the judged duration of a brief stimulus. Perception & Psychophysics, 63: 1229-1238.

Cheng, V. And Steemers, K. (2011) Modelling domestic energy consumption at district scale: A tool to support national and local energy policies. Environmental Modelling and Software, 26: 1186-1198.

Choongwan K. And Taehoon H. (2015) Development of a dynamic operational rating system in energy performance certificates for existing buildings: Geostatistical approach and data-mining technique. Applied Energy 154, 254–270.

CIBSE Guide A (2013) The Chartered Institution of Building Services Engineers, Environmental Design. Clarke, J. A., Johnstone, C. M., Kondratenko, I., Lever, M., McElroy, L. B., Prazeres, L., Strachan, P. A., McKenzie, F. And Peart, G. (2004) Using simulation to formulate domestic sector upgrading strategies for Scotland. Energy and Buildings, 36: 759–770.

Clarke, J. A., Johnstone, C. M., Kim, J. M. And Tuohy, P. G. (2009) Energy, Carbon and Cost Performance of Building Stocks: Upgrade Analysis, Energy Labelling and National Policy Development. Advances in Building Energy Research, 3(1): 1-20. DOI: 10.3763/ aber.2009.0301

Coakley, D., Raftery, P. & Keane, M. (2014) A review of methods to match building energy simulation models to measured data. Renewable and Sustainable Energy Reviews. Vol.37, p.123–141.

Committee on Climate Change (2008) Building a low-carbon economy – the UK 's contribution to tackling climate change. TSO (The Stationary Office). Available from: <u>https://www.theccc.org.uk/publication/building-a-low-carbon-economy-the-uks-contribution-to-tackling-climate-change-2/</u>

DECADE (1997) 2MtC - Two Million Tonnes of Carbon. Energy and Environment Programme, University of Oxford, UK, Environmental Change Unit.

DECADE (1995) DECADE: Domestic Equipment and Carbon Dioxide Emissions, Second Year Report 1995. Energy and Environment Programme, University of Oxford, UK, Environmental Change Unit.

DECADE (1994) DECADE: Domestic Equipment and Carbon Dioxide Emissions, First Year Report 1994. Energy and Environment Programme, University of Oxford, UK, Environmental Change Unit.

DECC (2011a) Energy consumption in the United Kingdom : 2011 Domestic energy consumption in the UK since 1970. Department of Energy and Climate Change. Available online at:

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file /44913 4/ECUK_Chapter_3 - Domestic_factsheet.pdf Note: Version 2011 is no longer available; this link will now allow download of version 2015. DECC (2011b) Pathways to 2050: MARKEL model review and Scenarios for DECC's 4th Carbon Budget Evidence Base. Final Report, AEA.

DECC (2011c) Statistical Release: UK climate change sustainable development indicator: 2010 greenhouse gas emissions, provisional figures and 2009 greenhouse gas emissions, final figures by fuel type and end-user. Department of Energy and Climate Change.

DECC (2012) Statistical Release: 2011 UK Greenhouse gas emissions, provisional figures and 2010 UK greenhouse gas emissions, final figures by fuel type and end user. Department of Energy and Climate Change.

DECC (2014) Statistical Release: 2013 UK Greenhouse gas emissions, provisional figures and 2012 UK greenhouse gas emissions, final figures by fuel type and end user. Department of Energy and Climate Change.

DEn&T (1981) A Simulation Model for UK Energy Demand. Research Report 33, Departments of the Environment and Transport, London, HMSO.

DesignBuilder (2015) DesignBuilder user manual Version 3.4. UK: DesignBuilder software limited, available online at: http://www.DesignBuilder.co.uk/helpv3.4/index.htm

DETR (1998) Energy Efficiency in the Domestic Sector: Technical Support Paper. Department of the Environment, Transport and the Regions. London, HMSO.

De Wilde, P. (2014) The gap between predicted and measured energy performance of buildings: A framework for investigation. Automation in Construction. 41, 40-49.

Dickson C.M., Dunster J.E., Lafferty S.Z., Shorrock L.D. (1996) BREDEM: testing monthly and seasonal versions against measurements and against detailed simulation models. Building Services Engineering Research and Technology, 17(3):135–40.

Dietz, R.N., Goodrich, R.W., Cote, E. A. And Wieser, R.F. (1986) Detailed Description and Performance of a Passive Perfluorocarbon Tracer System for Building Ventilation and Air Exchange Measurements. Measured Air Leakage of Buildings, ASTM STP 904. p.203–264.

Dimitriou, v., Firth, S.K., Hassan. T.M And Fouchal, F. (2016) BIM Enabled Building Energy Modelling: Development and Verification of a gbXML To IDF

Conversion Method. Proceedings of the 3rd IBPSA-England Conference BSO 2016, Newcastle, 12th-14th September 2016, 1126.

DoE (1990) An Evaluation of Energy related Greenhouse Gas Emissions and Measures to Ameliorate Them. Energy Paper 58, Department of the Energy. London, HMSO.

DoE (1979a) Energy Technologies for the United Kingdom: An Appraisal for R, D & D Planning. Energy Paper 39, Volume 1, Department of the Environment. London, HMSO.

DoE (1979b) Energy Technologies for the United Kingdom: An Appraisal for R, D & D Planning. Energy Paper 39, Volume 2 Annexes, Department of the Environment. London, HMSO.

DOE-2 (2016) The home of DOE-2 based building energy use and cost analysis software. Accessed on 03/10/2017 from: <u>http://www.doe2.com/</u>

Druckman, A. And Jackson, T. (2008). Household energy consumption in the UK: A highly geographically and socio-economically disaggregated model. Energy Policy, 36(8): 3177-3192.

DTI (2000) Energy Paper 68: Energy Projections for the UK. Department of Trade and Industry, London, HMSO.

DTI (1995) Energy Projections for the UK: Energy Use and Energy-Related Emissions of Carbon Dioxide in the UK, 1995 - 2020. Energy Paper 65, Department of Trade and Industry, London, HMSO.

DTI (1992) Energy Related Carbon Emissions in Possible Future Scenarios for the United Kingdom. Energy Paper 59, Department of Trade & Industry. London, HMSO.

ECI (2000) Lower Carbon Futures for European Households. Environmental Change Institute, Oxford, University of Oxford.

Emery. A.F. And Kippenhan. C.J. (2006) A long term study of residential home heating consumption and the effect of occupant behavior on homes in the Pacific Northwest constructed according to improved thermal standards. Energy, 31(5): 677 – 693.

EnergyPlus Weather Convertor (2015) Auxiliary EnergyPlus Programs. Accessed on 2nd June 2015 from: <u>http://bigladdersoftware.com/epx/docs/8-</u> <u>3/auxiliary-programs/using-the-weather-converter.html</u>

ETSU (1994) An Appraisal of UK Energy Research Development and Dissemination. Energy Technology Support Unit (ETSU) Volume I-VIII, Strategic Studies Department, R 83, London, HMSO.

European Union. (2002). On the Energy Performance of Buildings. Directive 2002/91/EC of the European Parliament and of the Council, Official Journal of the European Communities, December.

Evans, P. (1997) There Is a Marketable Solution to Sustainable Housing. MSc Thesis.

Evans, R. D. And Herring, H. P. J. (1989) Energy Use and Energy Efficiency in the UK Domestic Sector up to the Year 2010. Energy Efficiency Series 11. Department of Energy, London, HMSO.

Farahbakhsh, H., Ugursal, V.I., Fung, A.S. (1998) A residential end-use energy consumption model for Canada. International Journal of Energy Research, 22(13):1133–43.

Firth, S.K., Lomas, K.J., Wright, A.J. (2010) Targeting household energyefficiency measures using sensitivity analysis. Building Research and Information, 38 (1): 25-41. Hay, R. Walters, G. A. Shorrock, L. D. Lane, K. And Boardman, B. (1999) Assessing Potential Carbon Dioxide Savings Through the Application of Energy-Efficiency Measures in the Housing Stock. In Proceedings of the Chartered Institute of Building Services Engineers (CIBSE) National Conference, 3-5 October 1999, Harrogate. London, Chartered Institute of Building Services Engineers, 411- 420.

Huang Y.J., Brodrick J. (2000) A bottom-up engineering estimate of the aggregate heating and cooling loads of the entire building stock. In: Proceedings of the 2000 ACEEE summer study on energy efficiency in buildings, Pacific Grove, 20th–25th August.

Hughes, M. (2011) A Guide to the Cambridge Housing Model. DECC, London.

Hughes, M. And Palmer, J. (2012) The Cambridge Housing Model. DECC, London.

Hulme, M., Jenkins, G. J., Lu, X., Turnpenny, J. R., Mitchell, T. D., Jones, R.
G., Lowe, J., Murphy, J. M., Hassell, D., Boorman, P., McDonald, R. And Hill,
S. (2002) Climate Change Scenarios for the United Kingdom: The UKCIP02
Scientific Report. Tyndall Centre for Climate Change Research, School of
Environmental Sciences, Norwich, UK, University of East Anglia.

Hulme, M. And Jenkins, G. J. (1998) Climate Change Scenarios for the UK: Scientific Report. UKCIP Technical Report No.1, Norwich, Climatic Research Unit.

l'anson, S.J., Irwin, C. And Howarth, A.T. (1982) Air Flow Measurement Using Three Tracer Gases. Building and Environment. Vol.17 (4), p.245–252.

IEA (1997) Modelling Insights. International Energy Agency (IEA) Energy Environment Update, No. 5, Autumn 1997.

IPCC (2001) Climate Change 2001: The Scientific Basis. Technical Summary. Geneva, Intergovernmental Panel on Climate Change. IPCC (2007) Technical Summary. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

IPCC (2017) Technical Summary. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change.

IWEC Weather Files (2001) International Weather for Energy Calculations: User's Manual and CD-ROM. American Society of Heating, Refrigerating and Air-Conditioning Engineers, ASHRAE, Atlanta, GA.

Johnston D. (2003) A physically based energy and carbon dioxide emission model of the UK housing stock. Ph.D. thesis. Leeds Metropolitan University, UK.

Jones, P., Patterson, J., Lannon, S. (2007) Modelling the built environment at an urban scale e energy and health impacts in relation to housing. Landscape and Urban Planning, 83: 39-49.

Judkoff, R.D. (1988) Validation of Building Energy Analysis Simulation Programs at the Solar Energy Research Institute. 10, pp.221–239.

Judkoff, R. And Neymark, J. (1995) International Energy Agency Building Energy Simulation Test (BESTEST) and Diagnostic Method. NREL/TP-472-6231, National Renewable Energy Laboratory, Golden, CO.

Judkoff, R., Wortman, D., O'Doherty, B. and Burch, J. (2008) A Methodology for Validating Building Energy Analysis Simu- lations, National Renewable Energy Laboratory, Golden, CO.

Kalema, T., Johannesson, G., Pylsy, P. and Hagengran, P. (2008) Accuracy of energy analysis of buildings: a comparison of a monthly energy balance method and simulation methods in calculating the energy consumption and the effect of thermal mass. Journal of Building Physics, 32(2): 101–130.

Kane, T. (2013) Indoor temperatures in UK dwellings: Investigating heating practices using field survey data. Ph.D thesis. Loughborough University, UK.

Kane, T., Firth, S.K. And Lomas, K.J. (2015) How are UK homes heated? A city-wide, socio-technical survey and implications for energy modelling. *Energy and Buildings*, 86: 817–832.

Kannan, R., Strachan, N., Pye, S. And Balta-ozkan, N. (2007), UK MARKAL Model Documentation, London.

Karlsson, F., Rohdin, P. And Persson, M. (2007) Measured and predicted energy demand of a low energy building: impor- tant aspects when using building energy simulation. Building Service Engineering, 28(3): 223–235.

Kavgic, M., Mavrogianni, A., Mumovic, D., Summerfield, A., Stevanovic, Z. And Djurovic-Petrovic, M. (2010) A review of bottom-up building stock models for energy consumption in the residential sector. Building and Environment, 45: 1683-1697.

Khayatian F., Sarto L., Dall'O, G. (2016) Application of neural networks for evaluating energy performance certificates of residential buildings. Energy and Buildings 125, 45 – 54.

Leach, G., Lewis, C., Romig, F., Van Buren, A. And Foley, G. (1979) A Low Energy Strategy for the United Kingdom. London, Science Reviews Ltd.

Letherman, K. M. And Samo, S. R. (2000) Energy Conservation and Carbon Dioxide Emission Reduction In UK Housing – Three Possible Scenarios. In Proceedings of International Conference on Technology Watch and Innovation in the Construction Industry, April 2000, Brussels, Belgium, pp.53-57

Lutzenhiser. L. (1992) A cultural model of household energy consumption. Energy,17(1):47 - 60.

Mallaband, B., Haines, V., Morton, A., Foda, E., Beizaee, A., Beckhelling, J., Allinson, D., Loveday, D. AND Lomas, K. (2014) Saving Energy in the Home through Digital Feedback and Control Systems: An Introduction to the DEFACTO Project. Proceedings of Behave Conference, Oxford, UK. DOI: 10.13140/2.1.2242.4968.
Mccarthy, B., J. J., Canziani, N., A. Dokken, D. J., And White, K. S. (2001) Climate Change 2001: Impacts, Adaptation & Vulnerability. Contribution of Working Group II to the Third Assessment Report of the Intergovernmental Panel on Climate Change (IPCC). Cambridge, Cambridge University Press.

Menezes, C., Cripps, A., Bouchlaghem, D.,Buswell, R. (2012) Predicted vs. actual energy performance of non-domestic buildings: using post-occupancy evaluation data to reduce the performance gap. Energy 97, 355–364.

Mhalas, A., Kassem, M., Crosbie, T. And Dawood, N. (2013) A visual energy performance assessment and decision support tool for dwellings. Visual Engineering.

MIT (1997) Energy Technology Availability: Review of Longer Term Scenarios for Development and Deployment of Climate-Friendly Technologies. Cambridge, Massachusetts, USA, Massachusetts Institute of Technology Energy Laboratory.

Natarajan, S. And Levermore, G.J. (2007) Predicting future UK housing stock and carbon emissions. Energy Policy 35: 5719-5727.

Neymark, J. (2001) HVAC BESTEST: A Procedure for Testing the Ability of Whole- Building Energy Simulation Programs to Model Space Conditioning Equipment Preprint. In *Building Simulation 2001*. Rio de Janiero, Brazil.

Neymark and Judkoff (2002) International Energy Agency Building Energy Simulation Test and Diagnostic Method for Heating, Ventilation, and Air-Conditioning Equipment Models (HVACBESTEST- Volume 1: Cases E100-E200) National Renewable Energy Laboratory, Golden co, USA.

Neymark, J., Roberts, D. (2013) Deep In Data: Empirical Data-Based Software Accuracy Testing Using The Building America Field Data Repository. In *13th Conference of International Building Performance Simulation Association*. Chambery, France. 1202–1209.

Office of Public Sector Information (2008) Climate Change Act 2008. The stationary office limited, UK. Available online at:

http://www.legislation.gov.uk/ukpga/2008/27/pdfs/ukpga_20080027_en.pdf

Office of Public Sector Information (2009) The Climate Change Act 2008 (2020 Traget, Credit Limit and Definitions) Order 2009. The stationary office limited, UK.

Oladokun, M.G. And Odesola, I.A. (2015) Household energy consumption and carbon emissions for sustainable cities – A critical review of modelling approaches. Gulf Organisation for Research and Development. International Journal of Sustainable Built Environment, 4: 231–247

Olivier, D. Miall, H. Nectoux, F. And Opperman, M. (1983) Energy Efficient Futures: Opening the Solar Option. London, Earth Resources Research Ltd.

Oreszczyn, T., Hong, S., Ridley, I. And Wilkinson, P. (2006) Determinants of winter indoor temperatures in low income households in England. Energy and Buildings, 38: 245–252.

Palmer, J. And Cooper, I. (2012) United Kingdom housing energy fact file. DECC, London.

Polly, B., Kruis, N. And Roberts, D. (2012) NREL's Field Data Repository Supports Accurate Home Energy Analysis. US Department of Energy Technical Report. DOE/GO-102012-3513 (February 2012).

Raslan, R. And Davies, M. (2012) Legislating building energy performance: putting EU policy into practice. *Building Research and Information*, 40:305–316.

RdSAP Manual (2012) Energy Performance Certificates for Existing Dwellings. BRE. Accessed on 10/10/2017 from:

http://www.bre.co.uk/filelibrary/accreditation/rdsap9_91/BRE_RdSAP_Manua <u>I_v8_0-March_2012.pdf</u>

Royal Commission on Environmental Pollution (2000) Twenty second Report. Energy - The Changing Climate. Royal Commission on Environmental Pollution. London, HMSO. Saidur, R., Masjuki, H.H., Jamaluddin, M.Y. (2007) An application of energy and exergy analysis in residential sector of Malaysia. Energy Policy 35(2):1050–1063.

SAP (2012) The Government's Standard Assessment Procedure for Energy Rating of Dwellings. Published on behalf of DECC by: BRE. Accessed on 10/05/2015 from: <u>https://www.bre.co.uk/filelibrary/SAP/2012/SAP-2012_9-</u> 92.pdf

Scottish Homes (2003) 2002 Scottish House Condition Survey. Accessed from: <u>www.shcs.gov.uk/</u>

Seryak J. And Kissock. K. (2003) Occupancy and behavioral affects on residential energy use. American Solar Energy Society, Solar conference, Austin, Texas.

Shorrock, L. D. (1995) Potential Carbon Emission Savings from Energy Efficient Housing. BRE Information Paper IP 15/95. Garston, Watford, Building Research Establishment.

Shorrock, L. D. (1994) Future Energy Use and Carbon Dioxide Emissions for UK Housing: A Scenario. BRE Information Paper IP9/94. Garston, Watford, Building Research Establishment.

Shorrock, L.D., Dunster, J.E., Seale, C.F., Eppel, H. And Lafferty, S.Z. (1996) BREDEM: Testing monthly and seasonal versions against measurement and against simulation models. Proc. CIBSE A: Building Serv. Eng. Res. Technol. 17(3): 135-140.

Shorrock, L. D. And Dunster, J. E. (1997a) The Physically-based Model BREHOMES and its Use in Deriving Scenarios for the Energy Use and Carbon Dioxide Emissions of the UK Housing Stock. Energy Policy, Volume 25, No. 12, pp.1027-1037.

Shorrock, L. D. And Dunster, J. E. (1997b) Energy use and carbon dioxide emissions for UK housing: two possible scenarios. Building research establishment (BRE), information paper 7/97. Watford, UK. Shorrock, L. D. And Henderson, G. (1990) Energy Use in Buildings and Carbon Dioxide Emissions. BRE Report BR 170. Garston, Watford, Building Research Establishment.

Shorrock, L. D. Henderson, J. Utley, J. L. And Walters, G. A. (2001) Carbon Emission Reductions from Energy Efficiency Improvements to the UK Housing Stock. BRE Report 435. Garston, Watford, Building Research Establishment.

Shorrock L.D, Henderson J, Utley J.I. (2005) Reducing carbon emissions from the UK housing stock: Building Research Establishment (BRE). p. 80. Watford, UK.

Shorrock L.D., MacMillan S., Clark J., Moore G. (1991) BREDEM 8: A monthly calculation method for energy use in dwellings: testing and development. In: Proceedings of the building environmental performance '91, London, 10th–11th April.

Summerfield, A.J., Raslan, R., Lowe, R.J. And Oreszczyn, T. (2011) How Useful are Building Energy Models for Policy? A UK Perspective. Proceedings of the 12th Conference of International Building Performance Simulation Association, Sydney: 2477-2482.

Swan, L.G. And Ugursal, V.I., Beausoleil-Morrison, I. (2008) A database of house descriptions representative of the Canadian housing stock for coupling to building energy performance simulation. Journal of Building Performance Simulation, 2(2): 75–84.

Swan, L.G. And Ugursal, V.I., Beausoleil-Morrison, I. (2009) Modelling of end-use energy consumption in the residential sector: A review of modelling techniques. Renewable and Sustainable Energy Reviews. Vol.13 (8), p.1819–1835. Available from: doi:10.1016/j.rser.2008.09.033.

Taylor, S., Allinson, D., Firth, S.K. And Lomas, K. (2013) Dynamic energy modelling of UK housing: Evaluation of alternative approaches. 13th Conference of International Building Performance Simulation Association, Chambery, France. 745 – 751.

US Department of Energy (2013) Auxiliary EnergyPlus Programs: extra programs for EnergyPlus. US Department of Energy. Available online at: http://apps1.eere.energy.gov/buildings/energyplus/pdfs/auxiliaryprograms.pdf [Accessed: 2nd June 2015]

Watson, P. (2009) An introduction to UK Energy Performance Certificates (EPCs). Journal of Building Appraisal (5): 241–250. doi: 10.1057/jba.2009.22

Watts, C., Jentsch, M. F., James, P. (2011) Evaluation of domestic Energy Performance Certificates in use. Building Services Engineering Resolution Technol. 32,4 pp. 361–376.

Westphal, F.S. & Lamberts, R. (2005) Building Simulation Calibration Using Sensitivity Analysis. Proceedings of Building Simulation 2005. pp. 1331–1338.

Yilmaz, S., Allinson, D., Taylor, S. And Lomas, K. (2014) A parametric analysis of the energy savings predicted by different thermal models: implications for the UK Green Deal. Building Simulation and Optimization Conference, UCL, London.

APPENDIX A

Example of Energy Performance Certificate (EPC)

Energy Performance Certificate (EPC)



You could

save £2.763

over 3 years

17 Any Street, District, Any Town, B5 5XX

Dwelling type: Detached house Date of assessment: 15 August 2011 Date of certificate: 12 December 2011
 Reference number:
 0000-0000-0000-0000

 Type of assessment:
 RdSAP, existing dwelling

 Total floor area:
 165 m²

Use this document to:

Hot water

· Compare current ratings of properties to see which properties are more energy efficient

£549 over 3 years

£5,367

· Find out how you can save energy and money by installing improvement measures

Estimated energy costs of dv	£5,367										
Över 3 years you could save	£2,763										
Estimated energy costs of	of this home										
	Current costs	Potential costs	Potential future savings								
Lighting	£375 over 3 years	£207 over 3 years									
Heating	Heating £4,443 over 3 years £2,067 or										

These figures show how much the average household would spend in this property for heating, lighting and hot water. This excludes energy use for running appliances like TVs, computers and cookers, and any electricity generated by microgeneration.

£330 over 3 years

£2,604

Energy Efficiency Rating



Totals

The graph shows the current energy efficiency of your home.

The higher the rating the lower your fuel bills are likely to be.

The potential rating shows the effect of undertaking the recommendations on page 3.

The average energy efficiency rating for a dwelling in England and Wales is band D (rating 60).

Top actions you can take to save money and make your home more efficient											
Recommended measures	Indicative cost	Typical savings over 3 years	Available with Green Deal								
1 Increase loft insulation to 270 mm	£100 - £350	£141	Ø								
2 Cavity wall insulation	£500 - £1,500	£537	0								
3 Draughtproofing	£80 - £120	£78	0								

See page 3 for a full list of recommendations for this property.

When the Green Deal launches, it may allow you to make your home warmer and cheaper to run at no up-front cost. To find out more, contact the Green Deal Advice Service on 0800 XXX XXX or visit www.greendealadvice.org.

17 Any Street, District, Any Town, B5 5XX 12 December 2011 RRN: 0000-0000-0000-0000-0000

Energy Performance Certificate

Summary of this home's energy performance related features											
Element	Description	Energy Efficiency									
Walls	Cavity wall, as built, partial insulation (assumed)	★★★ ☆☆									
Roof	Pitched, 75 mm loft insulation	★★★ ☆☆									
Floor	Solid, no insulation (assumed)	-									
Windows	Partial double glazing	★★☆☆☆									
Main heating	Boiler and radiators, mains gas	***									
Main heating controls	Programmer, room thermostat and TRVs	****☆									
Secondary heating	None	-									
Hot water	From main system	★★★☆☆									
Lighting	Low energy lighting in 17% of fixed outlets	★★☆☆☆									

Current primary energy use per square metre of floor area: 298 kWh/m² per year

The assessment does not take into consideration the physical condition of any element. 'Assumed' means that the insulation could not be inspected and an assumption has been made in the methodology based on age and type of construction.

Low and zero carbon energy sources

Low and zero carbon energy sources are sources of energy that release either very little or no carbon dioxide into the atmosphere when they are used. Installing these sources may help reduce energy bills as well as cutting carbon. There are none provided for this home.

Opportunity to benefit from a Green Deal on this property

When the Green Deal launches, it may enable tenants or owners to improve the property they live in to make it more energy efficient, more comfortable and cheaper to run, without having to pay for the work upfront. To see which measures are recommended for this property, please turn to page 3. You can choose which measures you want and ask for a quote from an authorised Green Deal provider. They will organise installation by an authorised installer. You pay for the improvements over time through your electricity bill, at a level no greater than the estimated savings to energy bills. If you move home, the Green Deal charge stays with the property and the repayments pass to the new bill payer.

For householders in receipt of income-related benefits, additional help may be available.

To find out more, contact the Green Deal Advice Service on 0800 XXX XXX or visit www.greendealadvice.org.



17 Any Street, District, Any Town, B5 5XX 12 December 2011 RRN: 0000-0000-0000-0000

Energy Performance Certificate

Recommendations

The measures below will improve the energy performance of your dwelling. The performance ratings after improvements listed below are cumulative; that is, they assume the improvements have been installed in the order that they appear in the table. Further information about the recommended measures and other simple actions you could take today to save money is available at **[link to remote advice centre]**. Before installing measures, you should make sure you have secured the appropriate permissions, where necessary. Such permissions might include permission from your landlord (if you are a tenant) or approval under Building Regulations for certain types of work.

Measures with a green tick 🐼 are likely to be fully financed through the Green Deal, when the scheme launches, since the cost of the measures should be covered by the energy they save. Additional support may be available for homes where solid wall insulation is recommended. If you want to take up measures with an orange tick 📀 be aware you may need to contribute some payment up-front.

Recommended measures	Indicative cost	Typical savings per year	Rating after improvement	Green Deal finance
Increase loft insulation to 270 mm	£100 - £350	£47	E 51	0
Cavity wall insulation	£500 - £1,500	£179	<mark>-D 59</mark>	0
Draughtproofing	£80 - £120	£26	<mark>D 60</mark>	\bigcirc
Low energy lighting for all fixed outlets	£50	£43	<mark>D 61</mark>	
Replace boiler with new condensing boiler	£2,200 - £3,000	£339	C 74	\bigcirc
Replace single glazed windows with low-E double glazing	£3,300 - £6,500	£41	C 75	O

Alternative measures

There are alternative measures below which you could also consider for your home.

- External insulation with cavity wall insulation
- Biomass boiler (Exempted Appliance if in Smoke Control Area)
- Air or ground source heat pump
- Air or ground source heat pump with underfloor heating
- Micro CHP

Choosing the right package

Visit www.epcadviser.direct.gov.uk, our online tool which uses information from this EPC to show you how to save money on your fuel bills. You can use this tool to personalise your Green Deal package.

Green Deal package	Typical annual savings
Loft insulation	
Cavity wall insulation	Total equipme of £597
Draughtproofing	Total savings of £567
Condensing boiler	
Electricity/gas/other fuel savings	£0/£587/£0

Directgov Public services all in one place

You could finance this package of measures under the Green Deal. It could save you £587 a year in energy costs, based on typical energy use. Some or all of this saving would be recouped through the charge on your bill.

17 Any Street, District, Any Town, B5 5XX 12 December 2011 RRN: 0000-0000-0000-0000

Energy Performance Certificate

About this document

The Energy Performance Certificate for this dwelling was produced following an energy assessment undertaken by a qualified assessor, accredited by [scheme name]. You can get contact details of the accreditation scheme at [scheme website address], together with details of their procedures for confirming authenticity of a certificate and for making a complaint. A copy of this EPC has been lodged on a national register. It will be publicly available and some of the underlying data may be shared with others for the purposes of research, compliance and direct mailing of relevant energy efficiency information. The current property owner and/or tenant may opt out of having this information disclosed.

Assessor's accreditation number:	[accredition number]
Assessor's name:	[assessor name]
Phone number:	[phone]
E-mail address:	[e-mail]
Related party disclosure:	No related party

Further information about Energy Performance Certificates can be found under Frequently Asked Questions at www.epcregister.com.

About the impact of buildings on the environment

One of the biggest contributors to global warming is carbon dioxide. The energy we use for heating, lighting and power in homes produces over a quarter of the UK's carbon dioxide emissions.

The average household causes about 6 tonnes of carbon dioxide every year. Based on this assessment, your home currently produces approximately 9.5 tonnes of carbon dioxide every year. Adopting the recommendations in this report can reduce emissions and protect the environment. If you were to install these recommendations you could reduce this amount by 5.3 tonnes per year. You could reduce emissions even more by switching to renewable energy sources.

The environmental impact rating is a measure of a home's impact on the environment in terms of carbon dioxide (CO₂) emissions. The higher the rating the less impact it has on the environment.



Your home's heat demand

For most homes, the vast majority of energy costs derive from heating the home. Where applicable, this table shows the energy that could be saved in this property by insulating the loft and walls, based on typical energy use (shown within brackets as it is a reduction in energy use).

Heat demand	Existing dwelling	Impact of loft insulation	Impact of cavity wall insulation	Impact of solid wall insulation	
Space heating (kWh per year)	22,154	(725)	(4494)	N/A	
Water heating (kWh per year)	2,792				

Addendum

This dwelling may have narrow cavities and so requires further investigation to determine which type of cavity wall insulation is best suited.

APPENDIX B

SAP Worksheet (Version 9.92)

SAP WORKSHEET (Version 9.92)

1. Overall dwelling dimensions				
	Area (m²)		Average storey height (m)	Volume (m ^a)
Basement		(1a) ×	(2a) =	(3a)
Ground floor		(1b) ×	(2b) =	(3b)
First floor		(1c) ×	(2c) =	(3c)
Second floor		(1d) ×	(2d) =	(3d)
Third floor		(1e) ×	(2e) =	(3e)
Other floors (repeat as necessary)		(1n) ×	(2n) =	(3n)
Total floor area TFA = (1a)+(1b)+(1c)+(1d)+(1e)(1n) =	(4)		
Dwelling volume		(3a)+(3	b)+(3c)+(3d)+(3e)(3n) =	(5)
2. Ventilation rate				
	main secondary heating heating	other	total	m ^a per hour
Number of chimneys	+	+ =	× 40 =	(6a)
Number of open flues	+	+ =	× 20 =	(6b)
Number of intermittent fans			× 10 =	(7a)
Number of passive vents			× 10 =	(7b)
Number of flueless gas fires			× 40 =	(7c)
Infiltration due to chimneys, flues, fans, PS	SVs (6a)+(6b)+(7a)+(7b)+(7c) =	+ (5) =	Air changes per hour (8)
If a pressurisation test has been carried o	ut or is intended, proceed to	(17), otherwise	continue from (9) to (16)	
Number of storeys in the dwelling (n _s)				(9)
Additional infiltration			[(9) - 1] × 0.1 =	(10)
Structural infiltration: 0.25 for steel or ti if both types of wall are present, us of openings); if equal use 0.35	mber frame or 0.35 for maso se the value conesponding to	nry construction the greater wa	n Il area (after deducting are	(11) eas
If suspended wooden ground floor, ent	er 0.2 (unsealed) or 0.1 (seal	led), else enter	0	(12)
If no draught lobby, enter 0.05, else en	ter 0			(13)
Percentage of windows and doors drau	ight proofed			(14)
Window infiltration		0.25 - [0.2×	(14) ÷ 100] =	(15)
Infiltration rate	(8) •	(10) + (11) + (1	12) +(13) +(15) =	(16)
Air permeability value, q ₅₀ , expressed in c	ubic metres per hour per squ	are metre of en	velope area	(17)
If based on air permeability value, then (18 Air permeability value applies if a pressuri	3) = [(17) + 20]+(8), otherwis sation test has been done, or	se (18) = (16) r a design or sp	ecified air permeability is b	(18) being used
Number of sides on which dwelling is shell	tered			(19)
Shelter factor	((20) = 1 - [0.07	75 × (19)] =	(20)
Infiltration rate incorporating shelter factor		(21) = (1	8) × (20) =	(21)

Infiltration rate modified for monthly wind speed:

Monthly ave	erage win	d speed f	from Tabl	e U2										
	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec		
(22)m=	(22)1	(22)2	(22)3	(22)4	(22)5	(22)6	(22)7	(22)8	(22)	(22)10	(22)11	(22)12	•	
Wind Facto	r (22a)"	= (22)m	÷ 4											
(22a) _m =	(22a)1	(22a) ₂	(22a) ₃	(22a)4	(22a)6	(22a)6	(22a)7	(22a) ₈	(22a) ₉	(22a)10	(22a)11	(22a) ₁₂		
Adjusted inf	filtration ra	ate (allow	ing for sh	elter an	d wind s	peed) = (21) × (22	a)m						
(22b)m =	(22b)1	(22b) ₂	(22b) ₃	(22b)4	(22b) ₆	(22b)6	(22b)7	(22b) ₈	(22b) ₉	(22b)10	(22b)11	(22b)12	, L	
Cal culate e	ffective ai	ir change	rate for t	he applik	cable cas	9e:								
If mecha	anical ven	filation: a	air change	e rate thi	rough sy	/stem							0.5	(23a)
lf exhau	st air heat	t pump u:	sing Appe	endix N,	(23b) = (23a) × F,	_{wv} (equat	ion (N4))	, otherw	ise (23b)) = (23a)			(23b)
If balanc	ed with h	eat recov	ery: effi	ciency in	n% allow	ing for in	-use fact	or (from i	Table 4h)	=				(23c)
a) If bala	anced me	chanical	ventilation	n with he	at recov	ery (MVH	IR) (2	24a)m = (22b)m +	(23b) ×	[1 – (23c) ÷ 100]	_	
(24a) _m =	(24a)1	(24a) ₂	(24a)3	(24a)i ((24a)s (24a)s (2	24a)7 (2	24a)a (24a)» (2	24a)10	(24a)11	(24a) ₁₂	ĺ	(24a)
b) If bala	anced me	chanical	ventilation	n without	theat red	covery (M	IV) (2	4b) _m = ((22b) _m +	(23b)	_		_	
(24b) _m =	(24b)1	(24b)2	(24b)3	(24b)a ((24b)s (24 b)6 (2	24b)7 (2	24b)a (24b)9 (2	24b)10	(24b)11	(24b)12		(24b)
c) If who	ile house if (22b)m <	extract w < 0.5 × (2	entilation 3b), then	or positi (24c) =	ve input (23b);	ventilatio otherwise	n from ou e (24c) =	tside (22b)m +	0.5 × (23	3b)				
(24c) _m =	(24c)1	(24c)2	(24c)3	(24c)4 ((24c)s (24c)6 (2	24c)7 (2	4c)a (24c)9 (24c)10	(24c)11	(24c)12]	(24c)
d) lf natu	ural ventila	ation or v	hole hou	se positi	ive input	ventilatio	n from lo	ft						
i	if (22b) _m ≥	≥ 1, then	(24d)m =	(22b)m	otherw	rise (24d)	m = 0.5	+ [(22b)	m ² × 0.5	3				
(24d) _m =	(24d)1	(24d)2	(24d)3	(24d)4	(24d)5	(24d) ₆	(24d)7	(24d)a	(24d)9	(24d)10	(24d)11	(24d)12]	(24d)
Effective air	r change i	rate - er	ter (24a) or (24b) or (24c) or (24d)	in (25)							
(25) _m =	(25)	(25)2	(25)3	(25)4	(25)5	(25)6	(25)7	(25)8	(25)9	(25)10	(25)11	(25)12	l	(25)
If Append following i	ix Q appli nstead:	es in rela	tion to air	change	rate, the	effective	e air chan	ge rate is	s calculat	ed via A	ppendix (Q and use	e the	
Effective air	r change i	rate from	Appendix	Q calcu	lation st	neet								
(25) _m =	(25) ₁	(25)2	(25)3	(25)4	(25)	(25)6	(25)7	(25) ₈	(25)9	(25)10	(25)11	(25)12		(25)

3. Heat losses and heat loss parameter

Items in the table below are to be expanded as necessary to allow for all different types of element e.g. 4 wall types. The κ-value is the heat capacity per unit area, see Table 1e

Element	Gross area, m ²	Openings m ²	Net area A.m ²	U-v W/	alue m²K	= A	×U v/k	κ-value k.l/m²·K	Α×κ kl/K		
Solid door	aroa, m			×				NATIO IN	horrs.	(26)	
Semi-glazed door				×						(26a)	
Window				× *b	elow	=				(27)	
Roof window				× *b	elow	=				(27a)	
Basement floor				×		-				(28)	
Ground floor				×		=				(28a)	
Exposed floor				×		=				(28b)	
Basement wall		-	=	×		=				(29)	
External wall		-	=	×		=				(29a)	
Roof		-	=	×		=				(30)	
Total area of exte	mal element	sΣA, m²		(31)							
Party wall				×		=] []	(32)	
(party wall U-value	from Table 3	3.6, к according	to its constru	iction)							
Party floor										(32a)	
Party ceiling										(32b)	
Internal wall **										(32c)	
Internal floor										(32d)	
Internal celling										(32e)	
* for windows and ro ** include the areas of	of windows, u on both sides	se effective wind of internal walls a	ow U-value ca and partitions	liculated (sing form	wia 1/[(1/U-value)	+0.04] as g	iven in paragrap	h 3.2	
Fabric heatloss, W	//K = Σ (A ×	U)		(26)(30) +	(32)		=		(33)	
Heat capacity Cm =	$\Sigma(A \times \kappa)$		(28)(3	0) + (32)	+ (32a)	.(32e)		=		(34)	
Thermal mass para	ameter (TMP	= C _m + TFA) in	kJ/m²K		= (34) + (4)		=		(35)	
$\begin{split} & \text{Semi-glazed door} & \times & & \text{below} & = & (26a) \\ & \text{Window} & \times & \text{below} & = & (27) \\ & \text{Roof Window} & & \times & \text{below} & = & (27) \\ & \text{Basement foor} & & & & \text{semi-glazed foor} & & & & \text{semi-glazed foor} & & & & & \text{semi-glazed foor} & & & & & & \text{semi-glazed foor} & & & & & & & & & & & & & & & & & & &$											
Thermal bridges : 2	C (L × Ψ) cak	culated using Ar	pendix K	u sopara	ory corre	10 0301	(m) (00).			(36)	
if details of ther	mal bridging	are not known	(36) = 0.15	×(31)							
Total fabric heat loss	3				(33)	+ (36)	=			(37)	
Ventilation heat loss	calculated m	onthiy	(38) _m = 0.	33 × (25)	n × (5)	0.1					
(38)m = (38) (3	eb Mar (38) ₂ (38) ₃	Apr May (38) ₄ (38) ₅	Jun Jul (38) ₆ (38)	Aug 7 (38)a	(38) ₉	0α (38)	NOV (38)::	(38)12		(38)	
Heat transfer coeffic	ient, W/K	(39) _m = (37) + (3	8)m		1	1	110	1		(<i>j</i>	
$(39)_m = (39)_k$ (3	(39) ₂	(39)4 (39)4	(39) (39)	7 (39)a	(39)	(39)-	(39)++	(39)12			
(-le (selà	((30)	. 6.010	12010	1	Average =	Σ(39)112/	(12=	(39)	

Heat loss	paramet	er (HLP), W/m²	Ж	(40)	m = (3	39)m ·	÷ (4)								
(40) _m =	(40)	(40)2	(40)3	(40)	4 (40)6 (40) ₆	(40)7	(40)e	(40)9	(40)10	(40)11	(40)12			
											Av	erage = 2	E(40)112	/12=	(4	0)
Number	ofdaysi	n mont	h (l'able	a 1a)								10.1			1	
	Jan	Feb	Mar	/	Apr	May		Jun	Jul	Aug	Sep	Oct	Nov	Dec	-	
(41) _m =	(41):	(41)2	(41)	3 ((41)4	(41)	6	(41)6	(41)7	(41)a	(41)9	(41):0	(41)11	(41):2]	(41)
4. Water	' heating	energy	re qu in	emer	nt										kWh/ye	ar
Assumed	loccupar	icy, N													(42)	
if TF	A > 13.9	N=1+	1.76×	- 11 -	exp(-0.	00034	19 ×	(TFA-1	3.9%)] +	0.0013	(TFA -1	(3.9)			, ,	
if TF	A ≤ 13.9	N = 1										,				
Annual av	verage h	ot water	usage i	in litre	es per (tay V _e	iaver	age = (2	5 × N) +	- 36					(43)	
Reduce t	he annua	al averag	e hot w	vater	usage	by 5%	if th	e dwelli	ng is de	signed to	achieve	a water u	ise target	đ		
not more	that 125	litres pe	r perso	n per	r day (a	llwate	er us	e, hot a	nd cold)					-	1	
	Jan	Feb	Mar		Apr	May		Jun	Jul	Aug	Sep	Oct	Nov	Dec]	
Hot water	r usage ir	n litres p	er day i	orea	ach mor	ntn v _d	l,m =	factor fi	rom I ab	(e 10 × (4	13)	1440	(4.4)	(4.4)	1	
(44)m =	(44)1	(44)2	(44)	3	(44)4	(44)	5	(44)6	(44)7	(44)8	(44)9	(44)10	(44)11	(44)12		1000
Enormy	ontant of	hotunt	www	- 4 1	10V.			AT /3	600 W	Mhimoni		IO ablec 1b	(a) = Σ(44 1o.1d)	+)112 =		(44)
(A5) -	(45),	(45)a	a useu	- 4.1	10 × Vd (45)	/////	lm ∧	(45).	000 M	(45)	(45)	(45) m	(45)	(15)	1	
(40)m -	(45)1	(45)2	(40)	3	(40)4	(45)	9	(40)6	(top)	(40)0	(10)8	(45)10 To	tol = \(\A5	10/12		(45)
If instanta	aneous w	ater hea	ting at	point	of use	(no h	ot wa	ater stor	age), er	ter "0" in	(46) to (6	61)	101-2140	y112-	L	(10)
For comn	nunity he	ating ind	lude di	stribu	tion los	s whe	ether	or nat l	hot wate	r tank is j	present	.,				
Distributio	on loss (46) _m = 0	.15 x (4	15)m												
(46)m =	(46)1	(46)2	(46)	3	(46)4	(46)	5	(46)6	(46)7	(46)a	(46)9	(46)10	(46)11	(46)12	1	(46)
Storage v	volume (li	tres) inc	ludina :	anv s	olaror	WWH	IRS s	torage	within sa	ame vess	el		(4	7)		
lf comm	unity hea	ting and	no tani	k in d	welling	enter	110	litres in	(47)			L		.,		
Otherwis	e if no sto	ored hot	water (this ir	ncludes	; insta	ntan	eous co	mbi boil	lers) ente	r '0' in <mark>(4</mark>	7)				
Water sto	orage loss	3:														
a) If man	ufacturer	's decla	red loss	s fact	or is kn	own	(kWh	v/day):					(4	8)		
Tomo	arah wa f	a star fra	m Taki	- 0h										0)		
remp	erature t	actor tro	milab	8 20									(4	9)		
Energ	y lost fro	m water	storag	e, kW	/h/day		(48)	× (49) :	-				(5	0)		
b) If man	ufacturer	's decla	red loss	fact	or is no	t knov	wn :									
Hot v	watersto	rage loss	s factor	from	Table	2 (KW	h/lite	e/day)					(5	1)		
lf coi	mmun t y	heating :	see sec	tion 4	4.3									-		
Volu	me factor	from Ta	ible 2a										(5	2)		
Tem	perature	factor fro	om Tab	le 2b									(5	3)		
Ener	g y lost fro	om wate	r storag	je, kV	//h/day		(47)	× (51)	× (52) >	< (53) =			(5	4)		
Enter (50) or (54) i	in (55)											(5	5)		
Water sto	prage loss	s calcula	ted for	each	month	(56)m	= (5	5) × (41)m	1000		1000	1000		1	
(56)m =	(56)1	(56)2	(56)3	(5	10) 4	(56)5	(00)6	(56)7	8(00)	(00)9	(56)10	(56)11	(56)12	J	(56)
If the vest	sel conta	ins dedi	cated s	olar s	storage	or de	dicat	ed WW	HRS sto	orage, (57	')m = (56)	m × [(47)	– V _s] ÷ (4	47), else ((57)n = (56)m
where V ₈	is v _{ww} 1	rom App	endix (33 01	r (m11)	from /	нрре	INDIX H	(as appl	icable).		1000	(1979)	(1977)	1	-
(57)n =	(57)1	(57)2	(57)3	(5	97)4 	(57)5	(D7)6	(57)7	(57)8	(57)9	(57)10	(57)11	(57)12	J	(57)
Primary of (modified	arcuit los:	s for eac	n mont	In from	m Table here is	3 Solar:	wołcz	r hoatin	a and a	culindori	hormost	at attown	ah not for	commun	ih DHM ~	etome 1
(59)m =	(59)	(59)	(59)3	(5	59)4	(59)	(59)6	(59)7	(59)a	(59)9	(59):0	(59)11	(59)12	l onwisy	(59)

Combi los	s for ead	h month i	from Tab	le 3a, 3b	or 3c (en	ıter "0" if ı	not a con	nbi boiler)				
(61)m =	(61)1	(61)2	(61)3	(61)3	(61);	(61)3	(61)3	(61)3	(61)3	(61)3	(61)3	(61)12	(61)
Total heat	t required	for wate	r heating	calculate	ed for eac	sh month	$(62)_{m} = 0$).85 × (45)m +(46)	m + (57)m	+ (59)m +	+ (61)m	
(62) _m =	(62)1	(62)2	(62)3	(62)4	(62)5	(62)6	(62)7	(62)8	(62)9	(62)10	(62)11	(62)12	(62)
Solar DHV	olar DHW input calculated using Appendix G or Appendix H (negative quantity) (enter "0" if no sol ar contribution to water heating) add additional lines if FGHRS and/or WWHRS applies, see Appendix G)												
(add addi	itional line	es if FGH	RS and	or WWHF	RS applie	s, see Ap	pendix G	6)					
(63)m =	(63)1	(63)2	(63)3	(63)4	(63)5	(63)6	(63)7	(63)a	(63)9	(63)10	(63)11	(63)12	(63)
Combi loss for each month from Table 3a, 3b or 3c (enter "0" if not a combi bolier) (61) _m = $(61)_1$ (61) ₂ (61) ₃ (61) ₁₂ (61) Total heat required for water heating calculated for each month (62) _m = 0.85 × (45) _m + (46) _m + (57) _m + (59) _m + (61) _m (62) _m = $(62)_1$ (62) ₂ (62) ₃ (62) ₄ (62) ₅ (62) ₆ (62) ₇ (62) ₉ (62) ₉ (62) ₁₀ (62) ₁₁ (62) ₁₂ (62) Solar DHW input calculated using Appendix G or Appendix H (negative quartity) (enter "0" if no solar contribution to water heating) (add additional lines if FGHRS and/or WWHRS applies, see Appendix G (63) _m = $(63)_1$ (63) ₂ (63) ₃ (63) ₄ (63) ₅ (63) ₆ (63) ₇ (63) ₉ (63) ₉ (63) ₉ (63) ₁₀ (63) ₁₁ (63) ₁₂ Output from water heater for each month, kWh/month (64) _m = $(62)_m + (63)_m$ (64) _m = $(64)_1$ (64) ₂ (64) ₃ (64) ₄ (64) ₅ (64) ₆ (64) ₇ (64) ₉ (64) ₉ (64) ₁₀ (64) ₁₁ (64) ₁₂ Total per year (kWh/year) = Σ (64) ₁₁₂ = (64) <i>if</i> (64) _m < 0 then setto 0 Heat gains from water heating, kWh/month 0.25 × [0.85 × (45) _m + (61) _m] + 0.8 × [[46) _m + (57) _m + (59) _m]													
(64) _m =	(64)1	(64)2	(64)3	(64)4	(64)5	(64)6	(64)7	(64)8	(64)9	(64)10	(64)11	(64)12	_
								Total	per year	(kWh/yea	ar) = Σ(64)112 =	(64)
if (64) _m <	0 then se	tto 0											-
Heat gain	s from wa	iter heati	ng, kWh.	month 0	.25 × [0.8	35 × (45)	n + (61)m	$+0.8 \times$	(46)m + (57)m + (5	9)m]		
(65)m =	(65)1	(65)2	(65)3	(65)4	(65)5	(65)6	(65)7	(65)8	(65)9	(65)10	(65)11	(65)12	(65)

(65)_m = (65)₁ (65)₂ (65)₃ (65)₄ (65)₅ (65)₆ (65)₇ (65)₈ (65)₉ (65)₉ (65)₁₁ (65)₁₂ include (57)_m in calculation of (65)_m only if hot water store is in the dwelling or hot water is from community heating

5. Internal gains (see Tables 5 and 5a)

Metabolic	; gains (T	able 5), v	vatts											
	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec		
(66)m =	(66)1	(66)2	(66)3	(66)4	(66)s	(66)6	(66)7	(66)8	(66)9	(66)10	(66)11	(66a)12		(66)
Lighting g	ains (cal	culated in	1 Append	ix L, equ	ation L9 o	or L9a), a	lso see T	able 5						
(67) _m =	(67)1	(67)2	(67)3	(67)4	(67)5	(67)6	(67)7	(67) ₈	(67)9	(67)10	(67)11	(67)12		(67)
Appliance	es gains (calculate	d in Appe	endix L, e	quation L	.13 or L1	3a), also	see Tabl	e 5					
(68)m =	(68)1	(68)2	(68)3	(68)4	(68)5	(68)6	(68)7	(68)8	e(83)	(68)10	(68)11	(68)12		(68)
Cooking	gains (ca	lculated in	n Append	iix L, equ	ation L15	i or L15a)), also se	e Table 5	5					
(69)m =	(69)1	(69)2	(69)3	(69)4	(69)s	(69)6	(69)7	(69)a	(69)9	(69)10	(69)11	(69)12		(69)
Pumps a	nd fans g	ains (Tab	ie 5a)											
(70) _m =	(70)1	(70)2	(70)3	(70)4	(70)5	(70)6	(70)7	(70)a	(70)9	(70)10	(70)11	(70)12		(70)
Losses e.	.g. evapo	ration (ne	egative va	alues) (Ta	able 5)									
(71)m =	(71)1	(71)2	(71)a	(71)4	(71)6	(71)6	(71)7	(71)a	(71)e	(71):0	(71) н	(71)12		(71)
Water he	ating gair	ns (Table	5)											
(72) _m =	(72)1	(72)2	(72)3	(72)4	(72)5	(72)6	(72)7	(72)8	(72)9	(72)10	(72)11	(72)12		(72)
Total internal gains = $(66)_m + (67)_m + (68)_m + (69)_m + (70)_m + (71)_m + (72)_m$														
(73)m =	(73)1	(73)2	(73)a	(73)4	(73)6	(73)6	(73)7	(73)	(73)9	(73)10	(73)н	(73)12	((73)

6. Solar gains

Solar gains are calculated using solar flux from U3 in Appendix U and associated equations to convert to the applicable oitentation. Rows (74) to (82) are used 12 times, one for each month, repeating as needed if there is more than one window type,

	Access factor Table 6d		Area m²	So	lar flux N <i>I</i> m²		g⊥ Specific dat or Table 6b	a	Fi Specifi or Tab	F c data xle 6c		Gains (W)	
North		×		×		× 0.9 ×		×			=		(74)
Northeast		×		×		× 0.9 ×		×			=		(75)
East		×		×		× 0.9 ×		×			=		(76)
Southeast		×		×		× 0.9 ×		×			=		(77)
South		×		×		× 0.9 ×		×			=		(78)
Southwest		×		×		× 0.9 ×		×			=		(79)
West		×		×		× 0.9 ×		×			=		(80)
Northwest		×		×		× 0.9 ×		×			=		(81)
Roofwindows	1.0	×		×		× 0.9 ×		×			=		(82)
Solar gains in w	atts, calcula	ated fo	r each mor	nth (83),	= Σ(74) ₀	n(82)m							
(83)m= (83)1	(83)2	(83)3	(83)4	(83)5	(83)6	(83)7	(83)8 (83)9	(83	3)10 (8	3)11	(83)12]	(83)

Total ga	ins – inte	rnal and	solar (8	34)m = (73	3)m + (83)	, watts								
(84) _m =	(84)1	(84)2	(84)3	(84)4	(84)5	(84)6	(84)7	(84)8	(84)9	(84)10	(84)11	(84)12]	(84)
				-									-	
7. Mean	. Mean internal temperature (heating season)													
Tempera	emperature during heating periods in the living area from Table 9, Th1 (°C)												21	(85)
Utilisatio	n factor fo	or gains fo	or living a	irea, η _{1,π}	(see T	able 9a)								
	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec]	
(86)m=	(86)1	(86)2	(86)s	(86)4	(86)s	(86)6	(86)7	(86)a	(86)e	(86)10	(86)11	(86)12]	(86)
Mean int	em al tem	perature	in living a	area T ₁ (fo	oliow ste	os 3 to 7	in Table	9c)						
(87)m=	(87)1	(87)2	(87)3	(87)4	(87)6	(87)6	(87)7	(87)a	(87) ₉	(87)10	(87)11	(87)12]	(87)
Tempera	ture durin	g heating	g periods	in rest of	f dwelling	from Tal	ble 9, T _{há}	2 (°C)					-	
(88) _m =	(88)	(88)2	(88)3	(88)4	(88)6	(88)6	(88)7	88)a	e(88)	01(88)	(88)11	(88)12]	(88)
Utilisatio	n factor fo	or gains fo	or rest of	dwelling,	η _{2,m} (s	ee Table	9a)						-	
(89)m=	(89)1	(89)2	e(89)	(89)4	(89)s	(89)6	(89)7	(89)a	e(89)	(89)10	(89)11	(89)12]	(89)
Mean int	emaltern	perature	in the res	t of dwell	ling T ₂									
(10)	ow steps	8 to 9 in	l able 9c,	if two me	ain neatr	ng system	is see tu	rtner not	es in Tabi	ie 9C)			7	
(90)m=	(90)1	(90)2	(90)3	(90)4	(90)s	(90)6	(90)7	(90)a	(90) ₉	(90)10	(90)11	(90)12		(90)
Living an	ea fractio	n							f _{LA} =	Living a	rea ÷ (4)	=		(91)
Mean int	emaltern	perature	(for the v	whole dwe	elling) = t	f _{la} × T ₁ +	+(1 – f _{LA}) × T ₂					_	
(92) _m =	(92)1	(92)2	(92)3	(92)4	(92)5	(92)6	(92)7	(92)8	(92)9	(82)10	(92)11	(92)12		(92)
Apply ad	justment t	to the me	an intern	al tempe	rature fro	m Table	4e, wher	e approp	vriate					
(93)m=	(93)1	(93)2	(93)3	(93)4	(93)s	(93)6	(93)7	(93)a	(93)e	(93)10	(93)11	(93)12]	(93)
8. Spac	e heating	require	ment											
Set T _i to	the mean	internal	temperat	ure obtair	ned at st	ep 11 of 1	Fable 9b,	sothat	Г _{і,т} =(93),	and re-	calculate			
the utilis	ation facto	or for gain	ns using T	Table 9 a								-	-	
	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec		
Utilisatio	n factor fo	orgains, 1	ղ _ո ։											

Space heating requirement in kWh/m²/year (98) + (4) = (99) For range cooker boilers where efficiency is obtained from the Product Characteristics Database, multiply the results in (98)_m by $(1 - \Phi_{coso}/\Phi_{watv})$ where Φ_{coso} is the heat emission from the case of the range cooker at full load (in kW); and Φ_{watv} is the heat transferred to water at full load (in kW). Φ_{coso} and Φ_{watv} are obtained from the database record for the range cooker boiler. Where there are two main heating systems, this applies if the range cooker boiler is system 1 or system 2.

(94)m = (94)1 (94)2

Useful gains, n_mG_m, W

(95)m = (95)1 (95)2

(97)m = (97)1 (97)2

(98)m = (98)1

(96)m = (96)1 (96)2 (96)3

(98)2

(94)3

(95)3

(97)3

(98)3

Monthly average external temperature from Table U1

Heat loss rate for mean internal temperature, ${\rm L}_{\rm m}$, W

(94)4

(95)4

(96)4

(97)4

(98)4

 $= (94)_m \times (84)_m$

(94);

(95)5

(96);

(97)5

(98);

Space heating requirement for each month, kWh/month = 0.024 × [(97)m - (95)n] × (41)n

(94)6

(95)6

(96)6

(97)6

(94)7

(95)7

(96)7

(97)7

 $= (39)_m \times [(93)_m - (96)_m]$

(94)a

(95)e

(96)a

(97)8 (97)9

(94)9

(95)9

(96)₉

-

(94)10

(95)10

(96)10

(97)10

Total per year (kWh/year) = Σ(98)1...5,10.12 =

(94)11

(95)11

(96)11

(97)11

(98)10 (98)11 (98)12

(94)12

(95)12

(96)12

(97)12

(94)

(95)

(96)

(97)

(98)

	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec	
leat loss	rate L _m	(calculat	ed using	24°C int	emal tem	perature	and exter	mal temp	erature	for the ap	plicable d	imate (see Appe	andix U)
(100)m =						(100)6	(100)7	(100)a					(100)
Itilisation	n factor f	or loss η	m										
(101)m =						(101)6	(101)7	(101)a					(101)
Jseful los	ss, η_L	, (watts) = (100) _m × (10	1)m								
(102) _m =						(102)6	(102)7	(102)a					(102)
ains (in pplicable	temal ga e climate	iins as fo a, see Ap	r heating pendix U	exceptt)	hat colum	in (A) of T	able 5 is	always u	used; sol	argains	calculated	for the	(103
i voyin –				the sector la		(100)6	(100)/	100/8	1 100				(100)
pace co	oling rec	quiremen	t for mon	in, whole	awelling	(104)	2005 (KWT (104)-	$(104)_{0}$	4 × [(10	5)n - (10	JZ)m] × (4	1)m	
i o i yin						(104)6	(104)	(104)0			Total = Σ	(104)6.8=	(104
Cooled fr	action									f _C =	cooled are	a ÷ (4) =	(105
ntermitte	ncy fact	or (Table	10b)										
106)m						(106)6	(106)7	(106)a					
	olina rov	uiromon	formor	ih - (104) × (105) v (108)					Total = Σ (106)6.8 =	(106)
107)	oning rev			11-(104	/m ^ (100	(107)	m (107) ₇	(107)a					
1017/11						(107)0	(ivi p	(101)0			Total = Σ (107)s.s =	(107)
Space oo	oling re	quiremer	t in KWh/	m²/year							(10)	7) ÷ (4) =	(108)
	-								· · · · · · · · · · · · · · · · · · ·				
r. Fabri	c Energ	y Efficie	ncy (calo	ulated o	niy under	special o	onditions	, see seo	:00n 11)		(00) +	(100) -	(100

9a. Ene	rgy requ	irements	- Indivi	dual hea	iting sys	tems in a	cluding m	nicro-CH	P					
For any	space he	ating, spa	ace cooli i	ng or wat	er heatin	g provide	ed by com	mun t y l	neating us	e the alt	ernatīve v	vorksheet	96.	
Space h	eating:									_		_		
Fraction	of space	heat fror	n second	ary/supp	lementar	y system	n (Table 1	1)	"0" if non	e		(201)		
Fraction	of space	heat fror	n main s	ystem(s)			(202) = 1	- (201) =			(202)		
Fraction	of main I	heating fr	om main	system 2	2 i	f no secc	ond main	system e	enter "0"			(203)		
Fraction	of total s	pace hea	t from m	ain syste	m 1	((204) = (2	02) × [1-	- (203)] =			(204)		
Fraction	of total s	pace hea	t from m	ain syste	m 2		(205)) = (202)	× (203) =			(205)		
Efficient (from da 'space	y of mair tabase o efficiency	n space h r Table 4 / adjustm	eating sy a/4b, adji ent' colun	stem 1 (i usted wh nn of Tab	in %) ere appro ble 4c; for	priate bj gas and	y the amo 1 oil boiler	unt shoi 's see 9.1	wn in the 2.1)	Ľ		(206)		
If there	s a seco	nd main s	vstem co	molete (207)				,					
Efficienc	v of mair	1 space h	eating sv	stem 2 (i	in %)					Г		(207)		
(from da	tabase o	r Table 4	a/4b, adji	usted wh	ere appro	priate b	y the amo	unt sho	wn in the			()		
'space	efficiency	adjustm	enť colun	nn of Tab	ole 4c; for	gas and	f oil boiler	s see 9.	2.1)					
Efficienc	y of seco	ondary/su	pplement	tary heati	ing syster	m, % <i>(fr</i>	om Table	4a or Aj	opendix E) [(208)		
Cooling	System E	Energy Ef	ficiency F	Ratio (see	e Table 1	0c)				Г		(209)		
-	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec	kWh/year	
Space h	eating rec	quirement	t (calcula	ted abov	e)				o op					
	(98)1	(98)2	(98)3	(98)4	(98)5	-	-	-	-	(98)10	(98)11	(98)12		
Space h	eating fue	el (main h	neating sy	stem 1),	kWh/mo	nth								
(211)n =	(98) _m ×	(204) × 1	100 ÷ (20	6)										
(211) _m	(211)1	(211) ₂	(211)3	(211)4	(211)₅	-	-	-	-	(211)10	(211)11	(211)12		
									Total (k)	Wh/year)	=Σ(211) ₁	.5,10.12 =	(211))
Space h	eating fue	el (main h	leating sy	stem 2),	kWh/moi	nth, amit	t if no seco	ond mair	heating:	system				
(213) _m =	(98) _m × ($(205) \times 1$	00 ÷ (20	7)	(0.6.0)					0.400	(0.6.0)	0(0)		
(213)m	(Z13)1	(213)2	(213)3	(213)4	(213)5	-	-	-	- Total /k\	(Z13)10 Mb/uppr)	(Z13)11 -5/2121	(213)12	(213)	
									TOLAI (KY	viv year)	-2(213)	1.5,10.12=	(213)	'
Space h	eating fue	el (secono	dary), kW	h/month										
(215)n = (215)-	(90)m × (215),	(201) × 1 (215)a	(215)a	0) (215),	(215)-	-	_	_	_	(215)	(215)	(215)m		
(2 rojm	(215)1	215/2	1210/3	(215)M	1210/6	-	-	-	Total (k)	Mh/year)	$=\Sigma(215)$	1.5 10 12 =	(215)	١
										,,	2(2.0)	1.0,101.12	(210)	' .
Water h	eating													
Output f	rom wate	r heater ((calculate	d above))									
	(64)1	(64)2	(64)3	(64)4	(64)6	(64);	(64)7	(64)a	(64)9	(64)10	(64)11	(64)12		
Efficience	y of wate	er heater											(216))
(From d for gas	atabase (and oil bo	or Table 4 D i ers use	fa/4b, adj the sumr	iusted wh ner effici	here appr ency, see	opriate L 9.2.1)	by the am	ount sho	wn in the	'DHW ei	fficiency a	djustment'	column of Table 4c	ł,
if water I	neating by	y a hot-w	ater-only	boiler, (2	217) _m = va	alue from	n databasi	e record	for boiler	or Table	4a			
otherwis	e if gas/o	il boiler n	nain syste	ern used f	for water	heating,	(217) _m =	value ca	lculated f	oreach	month usi	ng equatio	n (8) in section 9.2.	1
otherwis	e if separ	ate hot w	ater only	heater (i	including	immersi	on) (217),	n = appli	cable valu	e from 1	fable 4a			
otherwis	e (other r	main syst	em 1 or 2	used for	r water he	eating) (2	217) _n = (2	16)						
(217) _n =	(217)1	(217)2	(217)3	(217)4	(217)s	(217)6	(217)7	(217)8	(217)9	(217)10	(217)11	(217)12	(217))
Euel for	water hea	ating, kW	h/month											

(219)

Space cooling Space cooling fuel, kWh/month $(221)_m = (107)_m \div (209)$

(ee :)m =)	107 Jm - 1	200						1					•	
(221)m						(221)6	(221)	(221a)s		Total	- 51(004)			(224)
										Tota	= 2(221)	6.8=		(221)
Annual to	otals										kV	Vh/year	kWh/year	
Space he	eating fue	el used, n	nain sys	stem 1									(211)	T
Space he	eating fue	el used, n	nain sys	stem 2									(213)	1
Space he	eating fue	el used, s	econda	ry									(215)	1
Water he	ating fue	l used											(219)	1
Space co	oling fue	el used (if	there is	a fixed o	cooling sy	stern, if i	not enter	0)					(221)	1
Electricit	y for pur	nps, fans	and ele	ectric kee	ep-hot (T	able 4f):								-
mechar	nical vent	ilation fan	s - balar	nced, extr	ract or po	sitive inp	out from a	outside]	(230a)
warm a	iir heating) system f	ans										1	(230b)
central	heating p	ump or w	ater pun	np within	warm air	heating	unit						1	(230c)
oil boile	erpump												1	(230d)
boiler f	ue fan]	(230e)
maintai	ning elec	tric keep-l	hot facili	ity for gas	combi b	oiler]	(230f)
pump f	orsolarw	ater heati	ing											(230g)
pump f	or storage	WWHR	6 (see si	ection G3	.3)									(230h)
Total elec	tricity for	the above	a, KWh/y	ear					su	m of (23	0a)(23	0h) =		(231)
Electricit	y for ligh	nting (calc	ulated in	n Append	lix L)									(232)
Energy s	aving/ge	neration	technol	logies (A	ppendice	s M,Na	nd Q)							
Electricity	generate	ed by PVs	(Appen	dix M) (ne	egative q	uantity)								(233)
Electricity	generate	ed by wind	turbine	(Append	ix M) (ne	gative qu	uantity)							(234)
Electricity	used or	net electri	city gen	erated by	micro-C	НР (Арр	endix N)	(negative i	if net ge	neration)				(235)
Electricity	generate	ed by hydr	o-electri	ic genera	tor (Appe	endix M)	(negative	e quantity)						(235a)
Appendib	x Qitems	annual (energy (i	items not	already	ind uded	on a mo	nthly basis)			Fuel	kWh/year	
Appendix	kQ, ⊲iten	1 descrij	ption>											
energ	y saved o	or generat	ed (ente	er as nega	ative qua	ntity)								(236a)
energ	y used (p	ositive qu	antity)											(237a)
Appendix	kQ, ⊲item	1 2 descri	ption>											
energ	y saved o	or generat	ed (ente	r as nega	ative qua	ntity)								(236b)
energ	y used (p	ositive qu	antity)											(237b)
(continue	e this list i	f addition	al items)										-	
Total deli	vered ene	ergy for all	USES					(211)(2	21) +(2	31) + (23	2)(237)	b) =		(238)

10a. Fuel costs – Individual heating systems includi	ng micro-CHP				
	Fuel		Fuel price		Fuel cost
Snace heating - main evelow 1	kWh/year		(Table 12)		£/year
Space rieating - main system i	(211)	×		× 0.01 =	(240)
Space heating - main system 2	(213)	×		× 0.01 =	(241)
Space heating - secondary	(215)	×		× 0.01 =	(242)
Water heating (electric off-peak tariff)					
High-rate fraction (Table 13, or Appendix F for elec	tric CPSU)			(243)	
Low-rate fraction		1.0 – (2	43) =	(244)	
High-rate cost	(219) × (2	(43) ×		× 0.01 =	(245)
Low-rate cost	(219) × (2	44) ×		× 0.01 =	(246)
Water heating cost (other fuel)	(219)	×		× 0.01 =	(247)
(for a DHW-only community scheme use (342a) or (342	b) instead of (24	7)			
Space cooling	(221)	×		× 0.01 =	(248)
Pumps, fans and electric keep-hot	(231)	×		× 0.01 =	(249)
(if off-peak tariff, list each of (230a) to (230g) separately	y as applicable a	and apply	fuel price accord	ding to Table 12a	
Energy for lighting	(232)	×		× 0.01 =	(250)
Additional standing charges (Table 12)					(251)
_					
Energy saving/generation technologies (233) to (2 <description></description>	(35a) as applicat one of (233) to (ble, repea 235a) ×	t line (252) as n	eeded × 0.01 =	(252)
Appendix Q tems: repeat lines (253) and (254) as no	eeded				()
<description>, energy saved or</description>	ne of (236a) etc	×		× 0.01 =	(253)
<description>, energy used or</description>	ne of (237a) etc	×		× 0.01 =	(254)
Total energy cost			(240)(242) +	(245)(254) =	(255)
11a. SAP rating – Individual heating systems includ	ing micro-CHP				
Energy part defeter (Table 12):					0.42 (256)
Energy cost defator (Table 12):					0.42 (200)
Energy cost factor (ECF)		1	(255) × (256)]	÷ [(4) + 45.0] =	(257)
SAP rating (Section 13)					(258)

12a. CO ₂ emissions – Individual heating	systems including micro-	CHP		Feelening forte		Freisriege	
		Energy k\Mb/uppr		Emission facto	N r	Emissions	
Space heating - main system 1		(211)	×	Ng C Og/KWV n	=	Ng CO2/year	(261)
Space heating - main system 2		(213)	×		=		(262)
Space heating - secondary		(215)	×		=		(263)
Energy for water heating		(219)	×		=		(264)
(for a DHW-only community scheme use (361) to (373) instead of (26	4)					
Space and water heating		(261)	+ (262) +	+ (263) + (264)	=		(265)
Space cooling		(221)	×		=		(266)
Electricity for pumps, fans and electric kee	p-hot	(231)	×		=		(267)
Electricity for lighting		(232)	×		=		(268)
Energy saving/generation technologies	(233) to (235a) as applic	able, repea	t line (26	9) as needed			
<description></description>	one of (233)	to (235a)	×		=		(269)
Appendix Q items repeat lines (270) and	d (271) as needed						
<description>, energy saved *</description>	one of (236a) etc	×		=		(270)
<description>, energy used *</description>	one of (237a) etc	×		=		(271)
* where the item is concerned only with (CO2 emissions use the right	-hand colum	n only.				
Total CO ₂ , kg/year			sum of	(265)(271)	=		(272)
Dwelling CO ₂ Emission Rate				(272) ÷ (4)	=		(273)
El rating (section 14)							(274)

13a. Primary energy – Individual heating systems including micro-CHP Same as 12a using primary energy factor instead of CO₂ emission factor to give primary energy in KWh/year

Community heating

9b. Energy requirements – Community heating scheme This part is used for space heating, space cooling or water heating Fraction of space heat from secondary/supplementary heating (Ta	ng provided by a community sche able 11) "0" if none	me. (301)	
Fraction of space heat from community system	1 – (301) =	(302)	
The community scheme may obtain heat from several sources. The includes boilers, heat pumps, geothermal and waste heat from por Fraction of heat from community CHP	e procedure allows for CHP and wer stations. See Appendix C.	up to four other heal (303a)	t sources; the latter
Fraction of community heat from heatsource 2 0 Fraction of community heat from heatsource 3 0 Fraction of community heat from heatsource 4 6	perations obtained from perational records or plant esign specification; omit	(303b) (303c) (303d)	
Fraction of community heat from heat source 5		(303e)	
Fraction of total space heat from community CHP Fraction of total space heat from community heat source 2 <desc Fraction of total space heat from community heat source 3 <desc< td=""><td>(302) × (303a) = iption> (302) × (303b) = iption> (302) × (303c) =</td><td>(304a) (304b)</td><td></td></desc<></desc 	(302) × (303a) = iption> (302) × (303b) = iption> (302) × (303c) =	(304a) (304b)	
Fraction of total space heat from community heat source 5 <desc Fraction of total space heat from community heat source 5 <desc Fraction of total space heat from community heat source 5 <desc< td=""><td>iption> (302) × (303d) = iption> (302) × (303e) =</td><td>(304d) (304e)</td><td></td></desc<></desc </desc 	iption> (302) × (303d) = iption> (302) × (303e) =	(304d) (304e)	
Factor for control and charging method (Table 4c(3)) for community Factor for charging method (Table 4c(3)) for community water here	ty space heating ating	(305) (305a)	
Distribution loss factor (Table 12c) for community heating system		(306)	
Space heating Annual space heating requirement		(98)	kWh/year
Space heat from CHP	(98) × (304a) ×	(305) × (306) =	(307a)
Space heat from heat source 2	(98) × (304b) ×	(305) × (306) =	(307b)
Space heat from heat source 3	(98) × (304c) × ((305) × (306) =	(307c)
Space heat from heat source 4	(98) × (304d) ×	(305) × (306) =	(307d)
Space heat from heat source 5	(98) × (304e) ×	(305) × (306) =	(307e)
Efficiency of secondary/supplementary heating system in % (from Space heating fuel for secondary/supplementary system	Table 4a or Appendix E) (98) × (301) >	< 100 ÷ (308) =	(308) (309)
Water heating			
Annual water heating requirement		(64)	
If DHW from community scheme:			
Water heat from CHP	(64) × (303a) × (305a) × (306) =	(310a)
Water heat from heat source 2	(64) × (303b) × ((305a) × (306) =	(310b)
Water heat from heat source 3	(64) × (303c) × ((305a) × (306) =	(310c)
Water heat from heat source 4	(64) × (303d) × (305a) × (306) =	(310d)
Water heat from heat source 5	(64) × (303e) × (305a) × (306) =	(310e)
If DHW by immension or instantaneous heater within dwelling: Efficiency of water heater			(311)
Water heated by immersion or instantaneous heater	(64) ×	: 100 ÷ (311) =	(312)
Electricity used for heat distribution 0.01 × [(30	7a)(307e) + (310a)(310e)] =		(313)
Cooling System Energy Efficiency Ratio			(314)
Space cooling (if there is a fixed cooling system, if not enter 0)	-	(107) ÷ (314) =	(315)
Electricity for pumps and fans within dwelling (Table 4f): mechanical ventilation - balanced, extract or positive input from	outside		(330a)
warm air heating system fans			(330b)
pump for solar water heating			(330g)

pump for storage WWHRS (see section G3.3)			(330h)
Total electricity for the above, kWh/year (330a) + + (330a)	0h) =		(331)
Electricity for lighting (calculated in Appendix L)	I		(332)
Energy saving/generation technologies (Appendices M and Q)			
Electricity generated by PVs (Appendix M) (negative quantity)	I		(333)
Electricity generated by wind turbine (Appendix M) (negative quantity)	Ī		(334)
Electricity generated by hydro-electric generator (Appendix M) (negative quantity)	1		(335a)
Appendix Q items: annual energy (items not already included on a monthly basis)	Fuel	kWh/year	
Appendix Q, <item 1="" description=""></item>			
energy saved or generated (enter as negative quantity)			(336a)
energy used (positive quantity)			(337a)
Appendix Q, <item 2="" description=""></item>			
energy saved or generated (enter as negative quantity)			(336b)
energy used (positive quantity)			(337b)
(continue this list if additional items)			

10b. Fuel costs - Community heating schen	18				
,	Heat or fuel required	Fuel price		Fuel cost	
Oracle handland from OUID	kWh/year	(Table 12)		£/year	
Space heating from CHP	(307a) ×		× 0.01 =		(340a) (340a)
Space heating from heat source 2	(307b) ×		× 0.01 =		(340b) (240a)
Space heating from heat source 3	(307c) ×		× 0.01 =		(340C) (240-0
Space heating from heat source 4	(30/d) ×		× 0.01 =		(3400)
Space heating from heat source 5	(307e) ×		× 0.01 =		(3408)
Space heating (secondary)	(309) ×		× 0.01 =		(341)
Water heating from CHP	(310a) ×		× 0.01 =		(342a)
Water heating from heat source 2	(310b) ×		× 0.01 =		(342b)
Water heating from heat source 3	(310c) ×		× 0.01 =		(342c)
Water heating from heat source 4	(310d) ×		× 0.01 =		(342d)
Water heating from heat source 5	(310e) ×		× 0.01 =		(342e)
If water heated by immersion heater: High-rate fraction (Table 13) Low-rate fraction	1	I.0 - (343) =	(343) (344)		
High-rate cost, or cost for single immersion	(312) × (343)	×	× 0.01 =		(345)
Low-rate cost	(312) × (344)	×	× 0.01 =		(346)
If water heated by instantaneous water heater	(312) ×		× 0.01 =		(347)
Space cooling (community cooling system)	(315) ×		× 0.01 =		(348)
Pumps and fans	(331) ×		× 0.01 =		(349)
(if off-peak tariff, list each of (330a) to (330g) s	eparately as applicable and	apply fuel price accordii	ng to Table 12a		
Electricity for lighting	(332) ×		× 0.01 =		(350)
Additional standing charges (Table 12)					(351)
Energy saving/generation technologies (3	33) to (335a) as applicable,	repeat line (352) as nee	ded		
<description></description>	one of (333) to (335a) ×		× 0.01 =		(352)
Appendix Q tems: repeat lines (253) and (2	59) as needed				
<description>, energy saved</description>	one of (336a) etc ×		× 0.01 =		(353)
<description>, energy used</description>	one of (337a) etc ×		× 0.01 =		(354)
Total energy cost	= (340a)(342e) +(345)	(354) =		(355)
11b. SAP rating - Community heating sch	eme				
Energy cost deflator (Table 12):				0.42	(356)
Energy cost factor (ECF)		[(355) × (356)] ÷	[(4) + 45.0] =		(357)
SAP rating (Section 13)					(358)
or a read (oppositio)				L	(300)

12b. CO₂ Emissions - Community heating scheme

CO2 from CHP (space and water heating) Omit (361) to (366) if no CHP	
Power efficiency of CHP unit (e.g. 25%) from operational records or design spec.	(361)
Heat efficiency of CHP unit (e.g. 50%) from operational records or design specification	(362)
Energy used Emission factor CO ₂ emissi KWh/year kgCO ₂ /kWh kgCO ₂ /yea	on ar
Space heating from CHP (307a) × 100 ÷ (362) = × Note A =	(363)
less credit emissions for electricity -(307a) × (361) ÷ (362) = × Note B =	(364)
Water heated by CHP (310a) × 100 ÷ (362) = × Note A =	(365)
less credit emissions for electricity –(310a) × (361) ÷ (362) = × Note B = Note A: factor for CHP fuel. Note B: factor for electricity generated by CHP	(366)
CO ₂ from other sources of space and water heating (not CHP)	
Efficiency of heat source 2 (%) If there is CHP using two fuels repeat (361) to (366) for the second fuel	(367b)
Efficiency of heat source 3 (%)	(367c)
Efficiency of heat source 4 (%)	(367d) (367c)
	(3076)
CO ₂ associated with heat source 2 [(307b)+(310b)] × 100 ÷ (367b) × =	(360)
CO ₂ associated with heat source 3 [(307d)+(310d)] × 100 ÷ (367d) × =	(370)
CO2 associated with heat source 5 [(307e)+(310e)] × 100 + (367e) × =	(371)
Electrical energy for heat distribution (313) × =	(372)
Total CO ₂ associated with community systems (363)(366) + (368)(372) = if it is negative set (373) to zero (unless condition in C7 of Appendix C is met)	(373) (373)
Space heating (secondary) (309) × =	(374)
Water heating by immersion heater or instantaneous heater (312) × =	(375)
Total CO ₂ associated with space and water heating (373) + (374) + (375) =	(376)
Space cooling (315) × =	(377)
Electricity for pumps and fans within dwelling (331) × =	(378)
Electricity for lighting (332) × =	(379)
Energy saving/generation technologies (333) to (334) as applicable, repeat line (380) as needed <description> one of (333) to (334) × =</description>	(380)
Appendix Q items repeat lines (381) and (382) as needed	(200)
<description>, energy saved one of (337a) etc × =</description>	(381)
Total CO ₂ , kg/year sum of (376)(382) =	(383)
Dwelling CO ₂ Emission Rate (383) + (4) =	(384)
El rating (section 14)	(385)
PERIOD AND A PERIO	(000)

13b. Primary energy – Community heating scheme Same as 12b using primary energy factor instead of CO₂ emission factor to give primary energy in kWh/year

APPENDIX C

Example of EPC XML File

<?xml version="1.0" encoding="UTF-8"?>

<RdSAP-Report xsi:schemaLocation="http://www.epcregister.com/xsd/rdsap http://www.epcregister.com/xsd/RdSAP/Templates/RdSAP-Report.xsd" xmlns="http://www.epcregister.com/xsd/rdsap" xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance"> <Calculation-Software-Name>NHER EPC Online</Calculation-Software-Name> <Calculation-Software-Version>9.0.0</Calculation-Software-Version> <User-Interface-Name>NHER EPC Online</User-Interface-Name> <User-Interface-Version>9.0</User-Interface-Version> <Schema-Version-Original>LIG-17.0</Schema-Version-Original> <SAP-Version>9.92</SAP-Version> <PCDF-Revision-Number>385</PCDF-Revision-Number> <Energy-Assessment> <Property-Summary> <Wall> <Description language="1">Solid brick, as built, no insulation (assumed)</Description> <Energy-Efficiency-Rating>1</Energy-Efficiency-Rating> <Environmental-Efficiency-Rating>1</Environmental-Efficiency-Rating> </Wall> <Wall> <Description language="1">Cavity wall, as built, insulated (assumed)</Description> <Energy-Efficiency-Rating>4</Energy-Efficiency-Rating> <Environmental-Efficiency-Rating>4</Environmental-Efficiency-Rating> </Wall> <Roof> <Description language="1">Pitched, no insulation (assumed)</Description> <Energy-Efficiency-Rating>1</Energy-Efficiency-Rating> <Environmental-Efficiency-Rating>1</Environmental-Efficiency-Rating> </Roof> <Roof> <Description language="1">Roof room(s), insulated (assumed)</Description> <Energy-Efficiency-Rating>4</Energy-Efficiency-Rating> <Environmental-Efficiency-Rating>4</Environmental-Efficiency-Rating> </Roof> <Roof> <Description language="1">Pitched, insulated (assumed)</Description> <Energy-Efficiency-Rating>4</Energy-Efficiency-Rating> <Environmental-Efficiency-Rating>4</Environmental-Efficiency-Rating> </Roof> <Floor> <Description language="1">Solid, no insulation (assumed)</Description> <Energy-Efficiency-Rating>0</Energy-Efficiency-Rating> <Environmental-Efficiency-Rating>0</Environmental-Efficiency-Rating> </Floor> <Floor> <Description language="1">Solid, limited insulation (assumed)</Description> <Energy-Efficiency-Rating>0</Energy-Efficiency-Rating> <Environmental-Efficiency-Rating>0</Environmental-Efficiency-Rating> </Floor>

<Window>

<Description language="1">Fully double glazed</Description>

<Energy-Efficiency-Rating>3</Energy-Efficiency-Rating>

<Environmental-Efficiency-Rating>3</Environmental-Efficiency-Rating>

</Window>

<Main-Heating>

<Description language="1">Boiler and radiators, mains gas</Description> <Energy-Efficiency-Rating>4</Energy-Efficiency-Rating>

<Environmental-Efficiency-Rating>4</Environmental-Efficiency-Rating>

</Main-Heating>

<Main-Heating-Controls>

<Description language="1">Programmer and room thermostat</Description>

<Energy-Efficiency-Rating>3</Energy-Efficiency-Rating>

<Environmental-Efficiency-Rating>3</Environmental-Efficiency-Rating>

</Main-Heating-Controls>

<Hot-Water>

<Description language="1">From main system</Description>

<Energy-Efficiency-Rating>4</Energy-Efficiency-Rating>

<Environmental-Efficiency-Rating>4</Environmental-Efficiency-Rating>

</Hot-Water>

<Lighting>

<Description language="1">Low energy lighting in 75% of fixed outlets</Description>

<Energy-Efficiency-Rating>5</Energy-Efficiency-Rating>

<Environmental-Efficiency-Rating>5</Environmental-Efficiency-Rating>

</Lighting>

<Secondary-Heating>

<Description language="1">Room heaters, mains gas</Description>

<Energy-Efficiency-Rating>0</Energy-Efficiency-Rating>

<Environmental-Efficiency-Rating>0</Environmental-Efficiency-Rating>

</Secondary-Heating>

<Has-Hot-Water-Cylinder>false</Has-Hot-Water-Cylinder>

<Has-Heated-Separate-Conservatory>false</Has-Heated-Separate-Conservatory>

<Dwelling-Type language="1">Semi-detached house</Dwelling-Type>

<Total-Floor-Area>175</Total-Floor-Area>

<Has-Fixed-Air-Conditioning>false</Has-Fixed-Air-Conditioning>

<Multiple-Glazed-Proportion>100</Multiple-Glazed-Proportion>

</Property-Summary>

<Energy-Use>

<Energy-Rating-Current>58</Energy-Rating-Current>

<Energy-Rating-Potential>76</Energy-Rating-Potential>

<Environmental-Impact-Current>48</Environmental-Impact-Current>

<Environmental-Impact-Potential>68</Environmental-Impact-Potential>

<Energy-Consumption-Current>283</Energy-Consumption-Current>

<Energy-Consumption-Potential>165</Energy-Consumption-Potential>

<CO2-Emissions-Current>8.7</CO2-Emissions-Current>

<CO2-Emissions-Current-Per-Floor-Area>50</CO2-Emissions-Current-Per-Floor-Area>

<CO2-Emissions-Potential>5.1</CO2-Emissions-Potential>

<Lighting-Cost-Current currency="GBP">107</Lighting-Cost-Current>

<Lighting-Cost-Potential currency="GBP">107</Lighting-Cost-Potential>

<Heating-Cost-Current currency="GBP">1638</Heating-Cost-Current>

<Heating-Cost-Potential currency="GBP">1105</Heating-Cost-Potential> <Hot-Water-Cost-Current currency="GBP">114</Hot-Water-Cost-Current> <Hot-Water-Cost-Potential currency="GBP">115</Hot-Water-Cost-Potential> <Energy-Rating-Average>60</Energy-Rating-Average> </Energy-Use> <Suggested-Improvements> <Improvement> <Sequence>1</Sequence> <Improvement-Category>5</Improvement-Category> <Green-Deal-Category>3</Green-Deal-Category> <Improvement-Type>Q</Improvement-Type> <Improvement-Details> <Improvement-Number>7</Improvement-Number> </Improvement-Details> <Typical-Saving currency="GBP">407</Typical-Saving> <Indicative-Cost>£4,000 - £14,000</Indicative-Cost> <Energy-Performance-Rating>67</Energy-Performance-Rating> <Environmental-Impact-Rating>59</Environmental-Impact-Rating> </Improvement> < Improvement> <Sequence>2</Sequence> </mprovement-Category>5<//mprovement-Category> <Green-Deal-Category>2</Green-Deal-Category> <Improvement-Type>W2</Improvement-Type> <Improvement-Details> <Improvement-Number>58</Improvement-Number> </Improvement-Details> <Typical-Saving currency="GBP">63</Typical-Saving> <Indicative-Cost>£4,000 - £6,000</Indicative-Cost> <Energy-Performance-Rating>68</Energy-Performance-Rating> <Environmental-Impact-Rating>61</Environmental-Impact-Rating> </Improvement> <Improvement> <Sequence>3</Sequence> <Improvement-Category>5</Improvement-Category> <Green-Deal-Category>3</Green-Deal-Category> <Improvement-Type>G</Improvement-Type> <Improvement-Details> <Improvement-Number>13</Improvement-Number> </Improvement-Details> <Typical-Saving currency="GBP">64</Typical-Saving> <Indicative-Cost>£350 - £450</Indicative-Cost> <Energy-Performance-Rating>69</Energy-Performance-Rating> <Environmental-Impact-Rating>63</Environmental-Impact-Rating> </Improvement> < Improvement> <Sequence>4</Sequence> <Improvement-Category>5</Improvement-Category> <Green-Deal-Category>2</Green-Deal-Category>

<Improvement-Type>U</Improvement-Type>

<Improvement-Details>

<Improvement-Number>34</Improvement-Number>

</Improvement-Details>

<Typical-Saving currency="GBP">262</Typical-Saving>

<Indicative-Cost>£5,000 - £8,000</Indicative-Cost>

<Energy-Performance-Rating>76</Energy-Performance-Rating>

<Environmental-Impact-Rating>68</Environmental-Impact-Rating>

</Improvement>

</Suggested-Improvements>

<Green-Deal-Package>

<Green-Deal-Improvement>

<Improvement-Type>Q</Improvement-Type>

<Improvement-Number>7</Improvement-Number>

</Green-Deal-Improvement>

<Green-Deal-Improvement>

<Improvement-Type>G</Improvement-Type>

<Improvement-Number>13</Improvement-Number>

</Green-Deal-Improvement>

<Electricity-Saving currency="GBP">0</Electricity-Saving>

<Gas-Saving currency="GBP">474</Gas-Saving>

<Other-Fuel-Saving currency="GBP">0</Other-Fuel-Saving>

</Green-Deal-Package>

<Renewable-Heat-Incentive>

<Space-Heating-Existing-Dwelling>30803</Space-Heating-Existing-Dwelling>

<Impact-Of-Loft-Insulation>-3769</Impact-Of-Loft-Insulation>

<Impact-Of-Solid-Wall-Insulation>-9037</Impact-Of-Solid-Wall-Insulation>

<Water-Heating>2334</Water-Heating>

</Renewable-Heat-Incentive>

</Energy-Assessment>

<SAP-Data>

<SAP-Property-Details>

<Property-Type>0</Property-Type>

<Built-Form>2</Built-Form>

<Multiple-Glazed-Proportion>100</Multiple-Glazed-Proportion>

<Multiple-Glazing-Type>3</Multiple-Glazing-Type>

<PVC-Window-Frames>true</PVC-Window-Frames>

<Glazing-Gap>16+</Glazing-Gap>

<Extensions-Count>1</Extensions-Count>

<Glazed-Area>1</Glazed-Area>

<Door-Count>2</Door-Count>

<Insulated-Door-Count>0</Insulated-Door-Count>

<Percent-Draughtproofed>100</Percent-Draughtproofed>

<Habitable-Room-Count>7</Habitable-Room-Count>

<Fixed-Lighting-Outlets-Count>16</Fixed-Lighting-Outlets-Count>

<Low-Energy-Fixed-Lighting-Outlets-Count>12</Low-Energy-Fixed-Lighting-Outlets-

Count>

<Measurement-Type>1</Measurement-Type>

<Mechanical-Ventilation>0</Mechanical-Ventilation>

<Open-Fireplaces-Count>0</Open-Fireplaces-Count>

<Solar-Water-Heating>N</Solar-Water-Heating>

<Conservatory-Type>1</Conservatory-Type>

<SAP-Building-Parts>

<SAP-Building-Part>

<Building-Part-Number>1</Building-Part-Number>

<Construction-Age-Band>B</Construction-Age-Band>

<Wall-Construction>3</Wall-Construction>

<Wall-Insulation-Type>4</Wall-Insulation-Type>

<Wall-Thickness-Measured>Y</Wall-Thickness-Measured>

<Wall-Thickness>230</Wall-Thickness>

<Wall-Dry-Lined>N</Wall-Dry-Lined>

<Party-Wall-Construction>0</Party-Wall-Construction>

<Roof-Construction>4</Roof-Construction>

<Roof-Insulation-Location>4</Roof-Insulation-Location>

<Floor-Heat-Loss>7</Floor-Heat-Loss>

<SAP-Floor-Dimensions>

<SAP-Floor-Dimension>

<Floor>0</Floor>

<Floor-Construction>1</Floor-Construction>

<Floor-Insulation>1</Floor-Insulation>

<Heat-Loss-Perimeter quantity="metres">22.9</Heat-Loss-Perimeter>

<Total-Floor-Area quantity="square metres">65.4</Total-Floor-Area>

<Room-Height quantity="metres">2.7</Room-Height>

<Party-Wall-Length quantity="metres">12.34</Party-Wall-Length>

</SAP-Floor-Dimension>

<SAP-Floor-Dimension>

<Floor>1</Floor>

<Heat-Loss-Perimeter quantity="metres">25.81</Heat-Loss-Perimeter>

<Total-Floor-Area quantity="square metres">65.13</Total-Floor-Area>

<Room-Height quantity="metres">2.66</Room-Height>

<Party-Wall-Length quantity="metres">12.29</Party-Wall-Length>

</SAP-Floor-Dimension>

</SAP-Floor-Dimensions>

<SAP-Room-In-Roof>

<Floor-Area quantity="square metres">22.89</Floor-Area>

<Construction-Age-Band>I</Construction-Age-Band>

<Insulation>AB</Insulation>

<Roof-Room-Connected>N</Roof-Room-Connected>

</SAP-Room-In-Roof>

<Roof-Insulation-Thickness>ND</Roof-Insulation-Thickness>

<ld><ldentifier>Main Dwelling</ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier><//ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><

<Wall-Insulation-Thickness>NI</Wall-Insulation-Thickness>

</SAP-Building-Part>

<SAP-Building-Part>

<ldentifier>Extension</ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier></ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier></ldentifier></ldentifier></ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//ldentifier><//lde

<Building-Part-Number>2</Building-Part-Number>

<Construction-Age-Band>I</Construction-Age-Band>

<Wall-Construction>4</Wall-Construction>

<Wall-Insulation-Type>4</Wall-Insulation-Type>

<Wall-Thickness-Measured>Y</Wall-Thickness-Measured>

<Wall-Thickness>290</Wall-Thickness>

<Party-Wall-Construction>NA</Party-Wall-Construction>

<Roof-Construction>5</Roof-Construction>

<Roof-Insulation-Location>4</Roof-Insulation-Location>

<Floor-Heat-Loss>7</Floor-Heat-Loss>

<SAP-Floor-Dimensions>

<SAP-Floor-Dimension>

<Floor>0</Floor>

<Floor-Construction>1</Floor-Construction>

<Floor-Insulation>1</Floor-Insulation>

<Heat-Loss-Perimeter quantity="metres">17.16</Heat-Loss-Perimeter>

<Total-Floor-Area quantity="square metres">21.1</Total-Floor-Area>

<Room-Height quantity="metres">2.4</Room-Height>

<Party-Wall-Length>0</Party-Wall-Length>

</SAP-Floor-Dimension>

</SAP-Floor-Dimensions>

<Roof-Insulation-Thickness>ND</Roof-Insulation-Thickness>

<Wall-Dry-Lined>N</Wall-Dry-Lined>

<Wall-Insulation-Thickness>NI</Wall-Insulation-Thickness>

</SAP-Building-Part>

</SAP-Building-Parts>

<SAP-Heating>

<Secondary-Heating-Type>603</Secondary-Heating-Type>

<Secondary-Fuel-Type>26</Secondary-Fuel-Type>

<Water-Heating-Code>901</Water-Heating-Code>

<Water-Heating-Fuel>26</Water-Heating-Fuel>

<Cylinder-Size>1</Cylinder-Size>

<Has-Fixed-Air-Conditioning>false</Has-Fixed-Air-Conditioning>

<Main-Heating-Details>

<Main-Heating>

<Main-Heating-Number>1</Main-Heating-Number>

<Main-Heating-Fraction>1</Main-Heating-Fraction>

<Main-Heating-Category>2</Main-Heating-Category>

<Main-Fuel-Type>26</Main-Fuel-Type>

<Main-Heating-Control>2104</Main-Heating-Control>

<Main-Heating-Data-Source>1</Main-Heating-Data-Source>

<Main-Heating-Index-Number>015166</Main-Heating-Index-Number>

<Boiler-Flue-Type>2</Boiler-Flue-Type>

<Fan-Flue-Present>Y</Fan-Flue-Present>

<Central-Heating-Pump-Age>0</Central-Heating-Pump-Age>

<Heat-Emitter-Type>1</Heat-Emitter-Type>

<Emitter-Temperature>0</Emitter-Temperature>

<Has-FGHRS>N</Has-FGHRS>

</Main-Heating>

</Main-Heating-Details>

<Instantaneous-WWHRS>

<Rooms-With-Bath-And-Or-Shower>2</Rooms-With-Bath-And-Or-Shower>

<Rooms-With-Mixer-Shower-No-Bath>1</Rooms-With-Mixer-Shower-No-Bath>

<Rooms-With-Bath-And-Mixer-Shower>0</Rooms-With-Bath-And-Mixer-Shower>

</Instantaneous-WWHRS>

<Immersion-Heating-Type>NA</Immersion-Heating-Type>

</SAP-Heating>

<SAP-Energy-Source>

<Wind-Turbines-Count>0</Wind-Turbines-Count>

<Wind-Turbines-Terrain-Type>2</Wind-Turbines-Terrain-Type>

<Meter-Type>2</Meter-Type>

<Mains-Gas>Y</Mains-Gas>

<Photovoltaic-Supply>

<None-Or-No-Details>

<Percent-Roof-Area>0</Percent-Roof-Area>

<PV-Connection>0</PV-Connection>

</None-Or-No-Details>

</Photovoltaic-Supply>

</SAP-Energy-Source>

<Heated-Room-Count>7</Heated-Room-Count>

<Low-Energy-Lighting>75</Low-Energy-Lighting>

</SAP-Property-Details>

</SAP-Data>

<Report-Header>

<RRN>9705-2843-7926-9405-0975</RRN>

<Inspection-Date>2015-12-07</Inspection-Date>

<Report-Type>2</Report-Type>

<Completion-Date>2015-12-08</Completion-Date>

<Registration-Date>2015-12-08</Registration-Date>

<Status>entered</Status>

<Language-Code>1</Language-Code>

<Tenure>1</Tenure>

<Transaction-Type>5</Transaction-Type>

<Property-Type>0</Property-Type>

<Energy-Assessor>

<Name>?????????/Name>

<Notify-Lodgement>N</Notify-Lodgement>

<Contact-Address>

<Address-Line-1>??????????/Address-Line-1>

<Address-Line-2>??????</Address-Line-2>

<Address-Line-3>???????</Address-Line-3>

<Post-Town>??????</Post-Town>

<Postcode>??????</Postcode>

</Contact-Address>

<Scheme-Web-Site>www.nesItd.co.uk</Scheme-Web-Site>

<E-Mail>epc@evolvepartnership.co.uk</E-Mail>

<Telephone>02392200598</Telephone>

<Company-Name>Evolve Partnership Limited</Company-Name>

<Scheme-Name>NHER</Scheme-Name>

<Identification-Number>

<Certificate-Number>????????</Certificate-Number>

</ldentification-Number>

</Energy-Assessor>

<Property>

<Address>

<Address-Line-1>?????? </Address-Line-1>

<Post-Town>????????</Post-Town> <Postcode>???????/Postcode> </Address> <UPRN>?????????/UPRN> </Property> <Region-Code>6</Region-Code> <Country-Code>EAW</Country-Code> <Related-Party-Disclosure> <Related-Party-Disclosure-Number>1</Related-Party-Disclosure-Number> </Related-Party-Disclosure> </Report-Header> <Insurance-Details> Insurer>Hiscox</lnsurer> <Policy-No>HU P16 1784443/1785588</Policy-No> <Effective-Date>2011-12-01</Effective-Date> <Expiry-Date>2016-01-31</Expiry-Date> <PI-Limit>1000000</PI-Limit>

- </lnsurance-Details>
- <ExternalDefinitions-Revision-Number>5.0</ExternalDefinitions-Revision-Number>
- </RdSAP-Report>

APPENDIX D

Equivalent Internal Boundary Conditions
Equivalent Internal Boundary Conditions

The modelling dataset does not include any details on the internal boundary conditions of the dwellings. EPCs are calculated based on SAP guidelines for estimating internal heat gains, losses and heating temperatures. Table 5 of SAP 2012 (SAP, 2012) identifies seven sources of internal heat gain: metabolic, lighting, appliances, cooking, losses, water heating and heat gains from pumps and fans in the heating system. For the RdDEM, the heating system, set-point temperatures and heating periods were all taken from Table 9 of SAP 2012 (SAP, 2012).

Metabolic Gains

Following the method described in SAP (SAP, 2012), the number of occupants and the resultant metabolic gains were calculated from the floor area (Equation D1):

$$N = 1 + 1.76 \times [1 - \exp(-0.000349 \times (TFA - 13.9)^2)]$$

+ 0.00013 × (TFA - 13.9)

Where N is the number of occupants, *TFA* is the total floor area of the dwelling. This equation gives a non-integer number for the assumed number of occupants.

Knowing the number of occupants, the metabolic gains were estimated from Equation D2 (Table 5 SAP 2012).

Metabolic Heat Gains =
$$60 \times N$$
 Watts D2

The above equation gives the average metabolic gain in watts (W) for the entire dwelling.

Lighting Gains

The calculation of lighting use was based on the proportion of fixed low energy lighting outlets installed, and on the contribution of daylight as described in Appendix L of SAP 2012. The average annual energy consumption for lighting (E_B) if no low-energy lighting is used can be derived from Equation D3 based on total floor area (*TFA*) and the number of occupants (*N*) (SAP, 2012):

$$E_B = 59.73 \times (TFA \times N)^{0.4714}$$
 kWh/year D3

Appendix L SAP 2012 gives two correction factors: C₁ to take account of fixed lighting outlets with low-energy lamps, as shown in Equation D4:

$$C_1 = 1 - 0.5 \times L_{LE}/L$$
 D4

Where L_{LE} is the number of fixed low energy lighting outlets and C₂ to take account of daylighting, as shown in Equations D5 and D6:

$$C_2 = 52.2 \ G_L^2 - 9.94 G_L + 1.433 \quad \text{if } G_L \le 0.095 \qquad D5$$

$$C_2 = 0.96 \qquad \text{if } G_L > 0.095 \qquad D6$$

Where G_L is calculated from Equation D7:

$$G_L = \frac{\sum 0.9 \times A_W \times g_L \times FF \times Z_L}{TFA}$$
D7

Where *FF* is the frame factor, taken as 0.7 for all dwellings, A_w is the window area, g_L is the light transmittance factor taken as 0.80 for all dwellings and Z_L is the light access factor taken as 0.83 for all dwellings.

The correction factors were then used to calculate the annual energy used for lighting (E_L) from Equation D8:

$$E_L = E_B \times C_1 \times C_2$$
 kWh/year D8

The monthly lighting energy use in kWh then can be derived from Equation D9:

$$E_{L,m} = E_L \times [1 + 0.5 \times \cos\left(\frac{2\pi(m-0.2)}{12}\right)] \times n_m/365$$
 kWh/year D9

Where n_m is number of days in month m. The associated internal heat gain for each month in watts then becomes (Equation D10):

$$G_{L,m} = E_{L,m} \times 0.85 \times 1000/(24 \times n_m)$$
 Watts/month D10

The factor 0.85 is an allowance for 15% of the total lighting usage being external to the dwelling.

Following this method, the average heat gains from lighting were calculated for each month of the year. These values were used in the RdDEM to create equivalent lighting gains in the dynamic simulation. The gains were proportioned between zones based on the ratio of the floor areas. To keep the model simple, no diurnal pattern was included; rather the gains were assumed to be spread evenly over the day.

Appliances Gains

Similar to lighting gains, electrical appliances gains were derived from their annual energy consumption (E_A) in kWh using equations D11 from SAP 2012:

$$E_A = 207.8 \times (TFA \times N)^{0.4714}$$
 kWh/year D11

The annual energy consumption of appliances is a function of total floor area (TFA) and number of occupants (N) in each dwelling. The annual

consumption then was spread over 12 months of year by equation D12 after SAP 2012:

$$E_{A,m} = E_A \times [1 + 0.157 \times \cos\left(\frac{2\pi(m-1.78)}{12}\right)] \times n_m/365$$
 kWh/year D12

Where n_m is the number of days in month *m*. From monthly energy consumption of appliances, the corresponding monthly heat gains were calculated as shown in D13 after SAP 2012:

$$G_{A,m} = E_{A,m} \times 1000/(24 \times n_m)$$
 Watts/month D13

The resulting values were used in the RdDEM to model equivalent appliance gains. The gains were proportioned between zones based on the ratio of the floor areas. To keep the model simple, no daily pattern was included; rather the gains were assumed to be spread evenly over the day.

Cooking Gains

Cooking gains were estimated based on assumed number of occupants, as shown in equation D14 after SAP 2012:

Cooking Heat Gains =
$$35 + 7 \times N$$
 Watts D14

In the RdDEM, the cooking gains were assigned to the ground floor zone only and were assumed to be spread evenly over the day.

Heat Losses

Table 5 SAP 2012 also includes a heat loss factor which comprises heat to incoming cold water and evaporation. This factor was calculated based on number of assumed occupants from Equation D15Error! Reference source not found.:

These losses were considered in dynamic simulation as reductions to the overall internal heat gains in each dwelling.

Infiltration and ventilation

Infiltration was modelled explicitly in the RdDEM. In the absence of any airtightness pressure test data in the reduced dataset, the SAP algorithm was used to calculate the infiltration rate based on the information on chimneys, fans, open flues and passive vents, available in the modelling dataset. The associated infiltration rate was calculated based on quantity of each item present in the dwellings and the associated ventilation rates, as shown in following Table.

A further infiltration rate of 0.1 ACH for two storey dwellings; and 0.35 ACH for masonry construction was added to dwellings' model as specified by SAP 2012.

Ventilation rates of chimneys, open flues, intermittent extract fans, passive vents and flueless gas fires required to calculate infiltration rate of dwellings (recreated from Table 2.1 (SAP, 2012))

Item	Ventilation rate m ³ /hour			
Chimney	40			
Open flue	20			
Intermittent extract fan	10			
Passive vent	10			
Flueless gas fire	40			

Since no information was available on number of sheltered sides for dwellings, two partially and one full sheltered sides was assumed for all dwellings. It was assumed that the dwellings were sheltered from one side due to presence of adjacent buildings and partially sheltered due to the buildings on the front and back of the dwellings. The two partially sheltered sides (front and back) were counted as one sheltered side after SAP 2012 Section 2.5 guidelines (SAP, 2012). The corresponding shelter factor was calculated based on two sheltered sides from Equation D16 after SAP 2012:

Shelter factor =
$$1 - [0.075 \times (sheltered sides)]$$
 D16

The infiltration rate for each dwelling, as modified by the shelter factor, was used as the infiltration rate of the building envelope in EnergyPlus, using the scheduled natural ventilation option. Any ventilation from window opening was not included in RdDEM.

Space Heating

All dwellings in the modelling dataset had a gas powered central heating system with a programmer and room thermostat to control it. For the RdDEM, the heating system was modelled using DesignBuilder's simple HVAC option with a condensing combination boiler and the resulting IDF was used in the template IDF (see Section 6.2.1) for all dwellings.

DesignBuilder provides three options to model heating systems (Designbuilder, 2015):

- i. Simple: where heating system is modelled using ideal loads.
- ii. Compact: where heating system is modelled parametrically.
- iii. Detailed: where heating system is defined in detail with each component placed on a schematic diagram and connected to other components using air and water flow networks.

Considering the limited amount of data available in the modelling dataset, the simple option was the most suitable. EnergyPlus then auto-sizes the system in order that it provides enough heat to meet the requirements of each zone.

Heating set-point temperatures were derived from the guidelines in Table 9 SAP 2012 with 21°C in the living area and a lower temperature for elsewhere in the dwelling which was calculated from the Heat Loss Parameter (HLP) as shown in Equation D17:

$$T_h = 21 - 0.5 \times HLP$$
 °C D17

Where T_h is the heating set-point of elsewhere in the house.

The *HLP* is calculated from the heat loss coefficient of the whole dwelling divided by the floor area as shown in the SAP worksheet (version 9.92) section 3 (SAP, 2012).

Since the dwellings in RdDEM were modelled with different heating zones to SAP (ground floor and first floor as explained in Section 4.3), the set-point temperature for each of the zones was calculated from a floor area weighted average. The living room area was estimated using the same method as SAP by using the number of habitable rooms as shown in following Table. It was assumed that the living room was located on ground floor in all dwellings and floor area averaged heating temperature set-points were derived based on the fraction of living area to ground floor area.

Living area fraction based on number of habitable rooms (re-created from Table S16 (SAP, 2012))

Number of rooms	1	2	3	4	5	6	7	8
Living area fraction	0.75	0.50	0.30	0.25	0.21	0.18	0.16	0.14

Heating periods in the RdDEM were assigned based on Table 9 SAP 2012: 07:00 to 09:00 and 16:00 to 23:00 on weekdays; and 07:00 to 23:00 on weekends. This was the same for all dwellings.